Optimisation of smart- and Vehicle-to-Grid charging strategies in distribution networks

by

Lennart Broere





Integration and optimisation of smart-charging and Vehicle-to-Grid charging strategies in distribution networks based on charging behaviour analysis

Master thesis submitted to Delft University of Technology in partial fulfilment of the requirements for the degree of

MASTER OF SCIENCE

in Management of Technology

Faculty of Technology, Policy and Management

by

Lennart Broere

Student number: 4030982

To be defended in public on January 11th 2018

Graduation committee

| First Supervisor: | Dr.ir. R.A. Hakvoort, Section Energy and Industry |
|----------------------|---|
| Second Supervisor: | Dr.ir. M. Kroesen, Transport and Logistics |
| External Supervisor: | Ir. Jelle Wijnja, Alliander Strategy and Innovation |

Preface

First I would like to express my gratitude towards the supervision I received from both Rudi Hakvoort and Maarten Kroesen from the TU Delft. I also would like to acknowledge and emphasise the excellent supervision I have received from Jelle Wijnja during the execution of this graduation project. Our weekly meetings resulted in valuable new insights and most of all kept me on track. The municipality of Amsterdam and the Hogeschool van Amsterdam receive graditude for providing and maintaining respectively the charging session data set. I also would like to declare this work originated from within myself and was made possible to endorse by the network company Liander, whom recently initiated a Vehicle-to-Grid pilot project in Amsterdam.

Abstract

Currently in the energy system of the Netherlands, lower level consumer demand flexibility is rather obsolete due to sufficient capacity limits of the distribution network and supply of capacity mechanisms by large industrial actors. However, in the upcoming 10-20 years, the power system is in transition to become decentralised with a higher share of renewable energy sources and significant increase in consumption. An operational control structure in the power system, where private consumers provide flexible capacity, is an effective and economical efficient approach to make sure the regulated process of electricity generation to supply at consumers is secure and reliable. Currently as a result of the EU Energy Efficiency Directive of 2012, an institutional base is presented for development initiatives of demand response in Europe [17]. Technical and regulatory standards now enable demand response flexibility to be offered on the wholesale and retail energy market and allow for consumer participation [34]. Demand response schemes are usually distinguished by the various motivation methods offered to the participating consumers. Programs include in general two control methods, centralised direct load control or time-based and incentive-based DR. Because these schemes rely on demand response decision-making by means of a centralised (multi) aggregator perspective, direct load control can be precisely adjusted to technical (local) grid constraints [69]. Practically, the objective of DR in this research is used to reduce congestion in distribution grids by moving part of BEV energy demand from (evening) peaks to the afternoon or night with direct control. By achieving these measures potential benefits arise, including the most profound in the distribution grid [29]:

- Optimising local grid assets by increasing the utilisation factor, and thereby maximise asset efficiency and subsequently decrease costs, which is beneficial for the DSO
- Scheduling of peak charging demand to aid congestion in distribution grids.

The modelling of the demand response charging strategies in Amsterdam fills the knowledge gap towards handling congestion for the DSO. It also provides a new study that addresses the potential to postpone future distribution grid investments by using charging strategies specifically for Amsterdam. The main research question that this study addresses is therefore:

What is the value of demand response management in a Vehicle-to-Grid network and does it provide increased benefits to smart charging for consumers and the distribution system operator in Amsterdam?

In order to grasp the subject of congestion prevention withing the time limits of graduation, the scope of this study is limited to assess the first mentioned item by modelling charging demand, and subsequently simulate optimal demand response charging strategies for a case study in Amsterdam's local power grid. The motivation for this study is threefold. Firstly, providing insight and recommendations in Amsterdam's BEV charging demand to compute the potential to provide demand flexibility by making use of a large-scale charging sessions data (2017-2018). Literature that uses such a vast data set for Amsterdam is scarce. Therefore, typical demand behaviour for daily and seasonal variation or periodicity is assessed by using a method of time series analysis and local regression analysis to derive time- and load-flexibility parameters. These parameters describe the measures for which demand response is generally defined. Secondly, a linear programming model is developed that optimises charging demand of smart charging and V2G charging strategies for demand response purposes to aid congestion at local feeders in the distribution grid of Liander. Lastly, an information-task exchange protocol is described from the perspective of a multi-aggregator to perform coordinated DR.

Further investigation about the usable amount of flexibility is simulated with linear programming for charging strategies (smart charging and V2G charging) to aid in local feeder congestion. In this research a case study is performed on data of a low-voltage feeder under different congestion situations, and a range of different number of BEVS connected to explore future distribution grid implications. Results show that the model's performance works especially well during peak demand periods throughout the day. In addition, the V2G strategy outperforms charged prices per kWh in almost every simulation compared to the uncontrolled scenario, by both charging during periods of low prices and discharging during periods of high prices even though the LP model does not optimise on prices. Thus, the V2G strategy allows the DSO to postpone grid investments in a number of cases while simultaneously the consumer almost always receives remuneration for its delivered services. The V2G strategy therefore provides a significantly added value over a smart charging strategy, by allowing electric vehicles to be charged whenever the feeder exhibits congestion. Future work includes analysis towards the value of DR in a liberalised system regarding the subject of the split-incentives challenge. Who initiates demand response (consumer, retailer, aggregator, DSO) and how should the benefits be divided along the supply chain. Handling an optimisation problem of DR for holds strong requirements for an global system balance in which neither participating actors are discriminated and the whole system benefits. An assessment of relational dependencies between participators in DR should be incorporated in designing charging strategies.

CONTENTS

| Pr | ace | v |
|-----|--|---|
| Ab | tract | vi |
| Lis | of Figures | xi |
| Lis | of Tables | xii |
| Lis | of Abbreviations | xiii |
| Nc | nenclature | xiv |
| 1. | ntroduction .1. Problem statement .2. Research objective .3. Research questions .1.3.1. Sub questions .4. Methodology .5. Research outline | 1 . 5 . 9 . 10 . 10 . 10 . 13 |
| 2. | Distributed energy network developments and electric vehicles 1. Distributed power management 2.1.1. Congestion Management 2.2. Market trends of electric vehicles 2.2.1. Implications for the distribution grid | 14 . 16 . 18 . 18 . 20 |
| 3. | Demand Response management .1. Energy market developments and the decentralisation paradigm .2. Dynamic demand response pricing | 21 . 23 . 25 . 27 . 29 |
| 4. | Demand response with smart charging and V2G charging in local grids .1. Electric vehicle demand management | 30 . 32 . 33 . 34 . 36 . 39 . 43 |
| 5. | Demand response smart charging and V2G charging economics .1. Business case for the Distribution System Operator | 44 . 46 |

| | 5.2. | Vehicle-to-Grid business model | 47 |
|----|--------------------|--|-------------------|
| | | 5.2.1. Implications for the distribution system operator | 50 |
| 6. | Мос | del design smart charging & Vehicle-to-grid | 51 |
| | 6.1. | Design criteria | 53 |
| | 6.2. | Modelling approach for optimal scheduling of charging sessions | 54 |
| | 6.3. | Modelling components - Input and Objective functions | 55 |
| | | 6.3.1. Model input | 55 |
| | | 6.3.2. Optimisation formulation | 59 |
| | 6.4. | Charging strategies quantification | 62 |
| 7. | Pred | dicting trends and periodicity in charging behaviour & Simulation | 63 |
| | 7.1. | Data collection and preparation | 64 |
| | 7.2. | Analysis of trends and seasonality in Electric vehicle session characteristics | |
| | | in Amsterdam | 66 |
| | | 7.2.1. Frequency of charging sessions | 70 |
| | | 7.2.2. Analysis of arrival-departure times | 71 |
| | | 7.2.3. Analysis of interconnection session characteristics | 76 |
| | | 7.2.4. Analysis of charging demand | 80 |
| | | 7.2.5. Analysis of time- & load- flexibility | 83 |
| | 7.3. | Charging behaviour implications for the distribution grid | 92 |
| | 7.4. | Case study - Simulation of electric vehicle demand response strategies | 93 |
| | | 7.4.1. Analysis of effects on grid congestion | 97 |
| 8. | Con | clusions | 107 |
| 9. | Futi | ure work and recommendations | 111 |
| Α. | Ene | rgy market transition | 113 |
| | | | |
| В. | Who | olesale and retail electricity market | 115 |
| C. | Ana C.1. | lysis of distributions of electric vehicle session characteristics Time- & Load flexibility | 116 120 |
| D. | feed | ler power load profiles | 122 |
| E. | Dut | ch electric vehicle market share & battery characteristics | 124 |

LIST OF FIGURES

| 1. | Global amount of electric vehicles on the road [29] | 8 |
|-----|--|-----|
| 2. | Conceptual framework of the methodology used in this study | 11 |
| 3. | Energy system control structure after Electricity Act of 1998 [3, 36, 42, 50] | 16 |
| 4. | Ratio public charging points per BEV [29] | 19 |
| 5. | USEF flexibility market structure [74] | 27 |
| 6. | Direct load control | 40 |
| 7. | Business model framework | 47 |
| 8. | Averaged Real-Time-Price tariff derived from EPEX-day-ahead market | |
| | prices for 24 hour [18] | 58 |
| 9. | Model input | 61 |
| 10. | Diagram for the optimisation simulation | 61 |
| 11. | Contour plot of charging sessions' frequency | 70 |
| 12. | Count of charging sessions' arrival (a) and departure (b) over time of day | |
| | with seasonal differences | 71 |
| 13. | Count of charging sessions' arrival time during week days (a) and weekend | |
| | days (b) | 72 |
| 14. | Decomposition of daily observations in weekly patterns for arrival time | 73 |
| 15. | Count of charging sessions' departure time during week days (a) and week- | |
| | end days (b) \ldots | 74 |
| 16. | Decomposition of daily observations in weekly patterns for departure time | 75 |
| 17. | Decomposition of daily observations in weekly patterns for connection time | 77 |
| 18. | Auto-correlation function of (a) arrival time, (b) departure time, (c) con- | |
| | nection time, (d) idle time | 79 |
| 19. | (a) Boxplot of daily charging demand for monthly effects | 81 |
| 20. | (a) Mean charging demand throughout the day for seasonal variation, (b) | |
| | mean charging demand throughout the day for daily variation | 82 |
| 21. | Decomposition of daily observations in weekly patterns for mean charging | |
| ~ ~ | demand | 83 |
| 22. | Charging profiles for the baseline and either smart charging and V2G | 84 |
| 23. | Boxplot of time-flexibility fraction of charging sessions for during every | |
| | season. [Note: bold black line = median; box = $25-75\%$; black whiskers | ~ ~ |
| 0.4 | inside 99% of the data = minimum and maximum, excluding outliers () Maximum data = $\frac{1}{1000}$ | 86 |
| 24. | (a) Mean time flexibility per hour of charging sessions throughout the | |
| | day for seasonal variation, (b) mean time flexibility per hour of charging | 07 |
| 05 | sessions throughout the day for weekday variation | 87 |
| 25. | (a) Aggregated total battery capacity per nour of the day for weekday | |
| | variation, (b) aggregated initial (arrival) battery capacity per nour of the | 00 |
| າເ | (a) A generated discharging note for results therein a sister of the former of the for | 88 |
| 20. | (a) Aggregated discharging rate for regular charging points per hour of | |
| | the day for weekday variation, (b) aggregated discharging rate for V2G | 00 |
| | charging points per nour of the day for weekday variation | 89 |

| 27. | Boxplot of load-flexibility for each individual charging session during every month. [Note: bold black line = median; box = $25-75\%$; black whiskers | |
|-----|---|-----|
| | = minimum and maximum, excluding outliers;outlier dots $=$ outside 99% | |
| | of the data] | 90 |
| 28. | Auto-correlation function of (a) charging demand, (b) time-flexibility, (c) | |
| | load-flexibility | 91 |
| 29. | Charging demand of sampled sessions | 94 |
| 30. | Feeder limit constraints according to base load of 144kW | 95 |
| 31. | Simulation LP program of the V2G charging strategy for 34 BEVs and a | |
| | feeder limit of 70% its maximum base load $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$ | 105 |
| 32. | APX spot Day-ahead electricity prices | 116 |
| 33. | Frequency distribution of the ratio charging time/connection time | 117 |
| 34. | Probability-density plot of connection time duration by sub-clusters of weekend days and weekdays for respectively $24h$ (a) & (c), $2nd$ $24h$ cluster | |
| | (b) & (d), 3rd 24h cluster (c) & (f) | 117 |
| 35. | Boxplot and regression smoothing line of connection time duration by sub- clusters of weekend days and weekdays for 24h (a) 2nd 24h cluster (b) | |
| | 3rd 24h cluster (a), $3rd 24h$ cluster (b), | 118 |
| 36 | Probability-density plot of idle time duration by sub-clusters of 24h con- | 110 |
| 50. | noction duration | 118 |
| 37 | Boxplet and regression smoothing line of idle time duration by sub-clusters | 110 |
| 57. | of weekend days and weekdays for 24h (a), 2nd 24h cluster (b), 3rd 24h | |
| | cluster (c) | 119 |
| 38. | (a) Probability-density distribution of charging demand | 120 |
| 39. | Frequency polygon time-flexibility fraction of the charging session distri- | |
| | bution | 121 |
| 40. | Frequency polygon load-flexibility of the charging session distribution | 122 |
| 41. | Total power load on feeder k2195-k2193 at Amsterdam Nieuw-West with | |
| | two V2G charging point | 123 |
| 42. | Total power load on feeder k2192-k2193 at Amsterdam Nieuw-West with | |
| | two V2G charging point | 123 |

LIST OF TABLES

| 1. | Example of keys and attributes of the recorded sessions | 64 |
|-----|--|-----|
| 2. | Charging points and session frequency per district in Amsterdam | 71 |
| 3. | Measure of each components influence of relative variance on total time | |
| | series variance of arrival time | 73 |
| 4. | Measure of each components influence of relative variance on total time | |
| | series variance of departure time | 75 |
| 5. | Summary of average and standard deviation of connection-, idle-, and | |
| | charging time clusters | 76 |
| 6. | Measure of each components influence of relative variance on total time | |
| | series variance of connection time | 77 |
| 7. | Measure of each components influence of relative variance on total time | |
| | series variance of charging demand | 82 |
| 8. | Summary central tendency of time- & load-flexibility for sub-cluster dis- | |
| | tributions and 50% IQR values \ldots | 84 |
| 9. | Parameter values of the sampled sessions, which are the input of the LP | |
| | model | 94 |
| 10. | Optimisation results for 50% (72kW) feeder limit of the maximum base load | 98 |
| 11. | Optimisation results for 60% (87kW) feeder limit of the maximum base load | 99 |
| 12. | Optimisation results for 70% (101kW) feeder limit of the maximum base | |
| | load | 99 |
| 13. | Optimisation results for 80% (116kW) feeder limit of the maximum base | |
| | load | 99 |
| 14. | Optimisation results for 90% (130kW) feeder limit of the maximum base | |
| | load | 100 |
| 15. | Optimisation results for 100% (144kW) feeder limit of the maximum base | |
| | load | 100 |
| 16. | Energy charged per BEV for both uncontrolled charging the V2G charging 1 | 101 |
| 17. | Additional BEV connection potential in the V2G optimisation cases 1 | 102 |
| 18. | Optimisation cost results for varying feeder limits and amount of electric | |
| | vehicles | 104 |
| 19. | Top five most sold PHEVs in the Netherlands and market share $[61, 43,$ | |
| | 78, 77] | 124 |
| 20. | Top ten most sold BEVs in the Netherlands and market share $[61, 49, 57,$ | |
| | 7, 28, 48, 58, 53, 45, 46 | 124 |

LIST OF ABBREVIATIONS

| BEV | Battery Electric Vehicles |
|------|------------------------------------|
| PHEV | Plug-in Hybrid Electric Vehicle |
| ICE | Internal Combustion Engine vehicle |
| DEG | Distributed Energy Generation |
| REG | Renewable Energy Generation |
| DR | Demand Response |
| V2G | Vehicle-to-Grid |
| SOC | State-Of-Charge |
| TSO | Transmission System Operator |
| DSO | Distribution System Operator |
| BRP | Balance Responsible Party |
| CPO | Charge Point Operator (Aggregator) |
| PDF | Probability Density Function |
| CDF | Cumulative Density Function |
| LP | Linear Program |
| | |

Nomenclature

INDICES

| n | BEV user [n] |
|-----------------|---|
| h | hourly time step indicator [h] |
| | PARAMETERS AND DECISION VARIABLES |
| Н | Time window for optimisation in hours [h] |
| $P^c_{n,h}$ | Load charging demand [kW] |
| $P^d_{n,h}$ | Load discharging [kW] |
| E_{max} | Maximum available battery capacity [kWh] |
| SOC_{min} | Minimum available battery capacity [kWh] |
| Ν | Total number of BEVs in the optimisation time window |
| Н | Total number of time slots in the optimisation window |
| t_{start} | Start date & time of the optimisation window [yyyy-mm-dd hh:mm:ss] |
| t_{end} | End date & time of the optimisation window [yyyy-mm-dd hh:mm:ss] |
| $c_{t,h}$ | total costs (dis)charging during optimisation window $[{\ensuremath{\in}}]$ |
| $p_{c,h}$ | electricity purchasing price $[{\textcircled{\sc end}}/k Wh]$ |
| $p_{d,h}$ | electricity selling price $[{\ensuremath{\in}}/{\rm kWh}]$ |
| $T_{Flex,t}$ | Time flexibility [h] |
| $P_{Flex,up}$ | Load flexibility up [kW] |
| $P_{Flex,down}$ | Load flexibility down [kW] |
| | |

1. INTRODUCTION

During the last decade, electrical mobility received an increase in appeal in response to technological developments, governmental subsidies, and a strict environmental policy for the near future that advocates an environmental and sustainable energy outlook for the near future. Reasons for this rise in popularity is denoted to consumer operational cost efficiency as alternative to conventional vehicles, while exhibiting promising characteristics in decreasing GHG emissions and increasing utilisation of renewable energy resources. Evolving policies resulting from the Dutch Electricity Act changed the organisation of electricity production and distribution due to disintegration of the energy market. Increasing distributed generation facilities of renewable energy appear to be economic competitive and supported by (local) governments. However, due to the stochastic characteristics of wind and solar, energy generation is difficult to predict and hence more complex to control. In addition, growing energy demand resulting from an increase in electric vehicles, electrical housing estate heating, and other modes of electrification might impose operational difficulties in the power system due to increasing peak power and intermittent generation. Therefore, the near future will impose implications to maintain reliable, efficient and secure energy transmission and distribution operations in the power system. The ongoing electrification also opts for cost effective methods to postpone large capital grid investments. This research therefore, aims to aid in local grid congestion by coordination of electric vehicle charging operations through smart-grid and V2G charging strategies. In addition an assessment of the DR capabilities for smart charging and Vehicle-to-Grid from a distribution system operator perspective is elicited to investigate the benefits for smart charging and Vehicle-to-Grid charging.

Due to increasing decentralised generation an increase in electricity consumption, and changing market structure, energy market-clearing mechanisms may no longer fit the developed technology standards and consumer roles [23], [30]. Subsequently, the need for energy market-clearing mechanisms that match decentralised supply with growing demand is in the near future a necessity. There is a requirement for consumer-side demand response - back-up for flexible power generation- to mitigate local grid overload (congestion) and match energy flows in the power system. These requirements are a cause of controlling an increasing amount of distributed generation sources while governing a central control strategy. In contrast, use of Renewable Energy Generation (REG) in distributed local grids has the potential to increase with use of local storage options or controllable consumer consumption - one of the applications of a demand response strategy. However, if development of distributed generation hampers due to slow transition of the top-down control approach this storage potential to stabilise the (local) grid is dissipated. If demand response potential benefits are to be exploited completely, a bottom-up control strategy is more effective.

Future expectations about the growth in the market share of BEVs in the Netherlands are positive and significantly large such that contingent energy demand can impose local distribution grid congestion problems. On the contrary, charging electric vehicles can also be controlled to provide flexibility for DR, which is one of the approaches to alleviate grid congestion. Electric vehicles are potentially an effective local grid DR approach, because of two fundamental characteristics, i.e. the spatial mobility to disperse an energy source over a region any point in time, and the fact that it can store electricity. Furthermore, research pointed out the flexibility BEVs have to defer energy demand due to longer connection time spans than theoretically necessary, and their inherent energy storage capabilities which at later times can be fed back in the grid when necessary. Due to new charging technologies, i.e. smart charging and Vehicle-to-Grid charging, which are presently slowly integrating in the Dutch power system infrastructure, real-time grid information and communication between energy system actors and consumers can be shared to jointly arrive at an effective and efficient management of energy operations and curtailment on a local distribution level.

While these challenges have been put forward rigorously in international literature, only disperse or limited research is available that elicits impact and DR potential of electric vehicles in the local distribution grid of Amsterdam. In addition, system-level comparison between smart charging and V2G is generally not addressed in literature. Therefore, in collaboration with (Al)Liander, the DSO of Amsterdam, this study addresses the notion to solve congestion problems at consumer demand level in Amsterdam for both charging strategies. This work provides the reader with an exploratory assessment on the methods to integrate electric vehicles in a direct demand response approach and statistically analyses patterns in historical charging session data. Furthermore, a model is developed that optimises planning of BEV charging sessions, by adoption of smart charging and Vehicleto-Grid charging strategies to reduce network peak electricity flows. Furthermore, this study develops a dynamical energy pricing tariff that expresses a charge for energy demand and flexibility as a competitive market price, and assesses the benefits in charging costs of the charging strategies compared to the uncontrolled charging. The motivation of this research is to ameliorate implications of uncontrolled BEV charging effects by aiming to relief grid loads and decrease additional capital expenditures. Therefore, the main research question driven by the need for sustainable and economical efficient energy management for electric vehicles is:

What is the value of demand response management in a Vehicle-to-Grid network and does it provide increased benefits to smart charging for consumers and the distribution system operator in Amsterdam?

Under the principle of consumer self-participation in demand response programs and the motivation of this research, an active participation in the electricity system by consumers contribute to market clearing and could receive remuneration for their provided service, while the power grid simultaneously benefits. Forthcoming sustainable objectives set by the Dutch coalition advocate an economical and operational efficient energy system emphasised by increasing renewable energy generation initiates this research.

1.1. PROBLEM STATEMENT

PRACTICAL RELEVANCE

From power generation, transmission and distribution to electricity consumption at consumers, electricity flows in the system become more difficult to predict with growing renewable energy generation and increasing consumption. Due to the uncertainties in predicting the stochastic behaviour of solar and wind, while fossil fuel power generation facilities In addition, increased consumption puts local distributional grids under stress, which consequently can overload local distribution assets, such as low-voltage cables and feeders. A potential result is unreliable, inefficient and costly distribution of electricity to consumers [75, 39]. However, while this problem is currently not apparent for Amsterdam's distribution grid, it is nonetheless a future challenge.

Optimisation of energy consumption and (local) distributed generation is one of the objectives of enabling smart grid technologies. It is an approach to reduce energy losses during production, transmission and distribution, and most importantly in this research, demand during peak hours. A key mechanism described in the literature is an active approach to plan, control, and monitor local energy consumption by performing demand response [19, 64, 66, 75, 84] with electric vehicles. Demand response is considered one of the solutions in a smart grid system to alter or reschedule consumers demand (profiles) incentivised by for example real-time electricity tariffs, to make it match technical grid constraints and generation. For example, in times of high demand, higher electricity use.

The main goals of demand response aim to aid blackouts, reduction of operational costs, increasing efficiency of the whole energy system [17, 75]. It is considered a cost-effective and reliable solution for alteration of demand profiles for both consumers and the power grid.

THEORETICAL RELEVANCE

Existing literature shows insufficient research on prediction of electric vehicle charging demand and the potential for smart charging and V2G charging in Amsterdam by using direct control schemes to coordinate charging of electric vehicles and complying to multiple energy actor's interests. An important note is to attain satisfactory BEV consumer charging service requirements. If the BEV consumers are not ensured with sufficient charging energy, it is likely that either they will not allow their BEV to be used for DR purposes and they may switch from charging point operator (aggregator) [69]. The grid services that can be offered through using V2G enable attractive energy products for the consumer in the Dutch liberalised energy system, but have to comply to other energy system actor's interests as well. It is therefore important to address what actors need to cooperate in the whole power system to enable demand response services to alter consumer loads precisely to distribution grid constraints.

This study therefore aims to fill the knowledge gap by proposing a coordinated information exchange framework that addresses the interactions between power system actors and prosumers necessary to assess the potential of demand response with V2G charging optimisation for the DSO and consumer. Furthermore, BEV charging demand predictions are necessary that are used to to model non-intrusive load levelling further downstream in the distribution grid to aid in congestion problems. The modelling of the demand response charging strategies in Amsterdam fills the knowledge gap towards handling congestion for the DSO. It also provides a new study that addresses the potential to postpone future distribution grid investments by using charging strategies specifically for Amsterdam.

CENTRALISED CONGESTION MANAGEMENT

Congestion management is achieved by controlling the charging-discharging rate of electric vehicles. In order to reduce peak power demand, load scheduling decisions have to account for DSO imposed grid constraints, consumer preferences, applied pricing structure and the availability of scheduling flexible and inflexible loads.

Specifically applied for managing energy demand of BEVs in the existing distributional grid, the main goals of demand response are directed to adjust load shapes by shifting demand to off-peaks hours to reduce peak demand. A method to enhance grid reliability and delay investments by reducing (peak) demand in a period of time to maintain electricity flows within technical boundaries. The motivation of this research is to investigate local grid congestion problems in Amsterdam due to increasing BEV electricity consumption, and applies demand response with two different charging modes. It is the first study carried out for Amsterdam that quantifies flexibility of BEVs under their current charging behaviour. It furthermore incorporates a Real-Time pricing scheme that reflects the electricity curtailment prices on an hourly basis of the wholesale day-ahead market, which entail a fair cost of electricity on nearly real-time.

POTENTIAL OF ELECTRIC VEHICLES

While the power system deals with increasing (BEV) consumer demand, local grids in Amsterdam can subsequently endure peak loads jeopardising grid operation [14]. Demand response with BEVs in the light of recent charge point innovations, is considered one of the solutions to improve local grid conditions under increasing demand. DSOs and Liander in specific elicit research in the area of congestion prevention for their local grids in order to suspend additional grid investments by performing demand response with BEVs. Liander's Vehicle-to-Grid pilot project in Amsterdam shows potential to mitigate local congestion by use of electric vehicle demand response, but requires a rigour assessment of its feasibility with real world BEV charging data. The benefits of BEV demand response through Vehicle-to-Grid charging from the DSO perspective are not yet captured for the urban area of Amsterdam in the current BEV penetration scenario and near future scenarios. The next section, therefore, makes assumptions for short-term and long-term scenarios which will be used in the calculation of the charging strategies.

Scenarios

An increasing trend of sold BEVs in the Netherlands is observed while consumers are increasingly supported by a growing accessibility to a charging infrastructure and electric vehicle procurement subsidies amongst others. Moreover, the governments most recent coalition report advocates that every new car sold in the Netherlands should be zeroemission by 2030 at the latest [79]. This entails the necessity for a responsible charging infrastructure development to meet consumer's energy demand. Therefore, in order to foresee any power system related implications of this growing need for BEV energy, short-term and long-term realistic scenarios are investigated that denote future BEV market share penetration. Furthermore, the market share of electric vehicles on the road indirectly relates, next to the total demand, also to the potential BEVs exhibit to participate in charging strategies for demand response. Thus, to derive near future power system demand and potential influence of DR charging strategies on peak power moments through time, the near future outlook for electric vehicle market share is derived. However, the short-term and long-term scenarios are the most extreme expected scenarios for the near future, though in the case study in paragraph 7.4, a varying penetration rate is adopted to provide more insight in the V2G charging strategy's capabilities.

As more countries, such as the Netherlands follow similar environmental objectives as is constituted in the Paris Agreement, the global outlook for penetration growth of BEVs, as is analysed by [29], up to 2030 and their expected number is depicted in figure 1. These numbers can be roughly translated to the expected market share globally of 10% in 2020 and 75% in 2030 of all vehicles on the road.



Notes: PLDVs = passenger light duty vehicles; LCVs = light commercial vehicles; BEVs = battery electric vehicles; PHEV = plug-in hybrid electric vehicles.

Figure 1: Global amount of electric vehicles on the road [29]

The following realistic scenarios are developed according to the figure above and the most recent policy measures advocated in [79]:

- Short-term scenario 2020: The Dutch Ministry of infrastructure and Water management advocates goals in which 50% of all new passenger cars sold will be driven by electric energy. At least 30% of that percentage will amount to fully electric vehicles [79]. This means contingent changes in electric vehicle penetration over all vehicles in the Netherlands is estimated to be equal to 10% market share in 2025, otherwise similar to approximately 175.000-200.000 BEVs according to [35, 65, 29, 79].
- Long-term scenario 2030: The Government exerts a mandate to completely stop the sale of cars with internal combustion engines (ICE). All new passenger cars sold will produce zero CO2 emission. The total BEV penetration is estimated to grow to approximately 75% depending on realistic scenarios [29, 16].

1.2. Research objective

The goal of this research is to investigate how BEV charging can aid congestion problems in the local grid of the DSO by considering BEV demand response possibilities utilised in smart charging and Vehicle-to-Grid charging strategies. The aim is to show that within current smart grid technologies electric vehicles can complement instead of burden Amsterdam's local distribution grid by directly altering consumer's consumption habits. Therefore, the strategies' objective is to deliver the maximum amount of required energy to the electric vehicles while ensuring the local grid to operate within technical limits. Within a V2G charging infrastructure, electric vehicles can facilitate on local power system-level demand and supply energy flows by charging during off-peak hours and discharging during peak hours, through which consumers can sell back energy to the grid [17, 75]. The direct load control method is analysed in this research and subsequently modelled due to the recent developments in Dutch institutional frameworks (USEF) that make integration of BEV DR relatively easily [1].

By performing an assessment on the potential benefits of BEV demand response with a market-driven approach, the feasibility of both smart charging and Vehicle-to-Grid charging is investigated. An algorithm is developed that computes the effect of timeand load- shifting of individual charging sessions within their session window, to ascertain mutual benefits to both the DSO and the consumers. Therefore the objectives of this study are:

- 1. Providing insight in the current power system and the implementation of demand response explicitly for BEVs to derive their present and future role in the local distribution grid of Amsterdam. Subsequently to develop a framework that focuses on a centralised control approach where multiple power system actors participate simultaneously and influence charging schedules for grid operations. In addition, the most important demand response value flows for the DSO are elicited.
- 2. Time-based assessment on prediction of BEV charging behaviour in Amsterdam for public charging points to quantify temporal trends, potential flexibility to utilise for demand response, and impact on specific feeders throughout Amsterdam for which data is available.
- 3. To develop a proof-of-concept linear programming model that optimises BEV demand profiles in terms of congestion minimisation and associated charging/discharging costs, for aim to relieve local grid congestion problems.

1.3. Research questions

The following research questions follow logically from the objectives set in this research: What is the value of demand response management in a Vehicle-to-Grid network and does it provide increased benefits to smart charging for consumers and the distribution system operator in Amsterdam?

1.3.1. SUB QUESTIONS

- 1. How could the distribution power network (in Amsterdam) benefit from application of demand response from BEVs?
- 2. What are the trends and periodicity that can be distinguished in BEV charging behaviour and how does this relate to impact on the grid and potential time- and load- flexibility to be utilised in demand response?
- 3. How to optimally schedule BEV charging demand in Amsterdam for smart charging and Vehicle-to-Grid charging strategies to mitigate congestion and minimise grid operational costs for both the distribution system operator and consumer?
- 4. How much peak load can be deferred by BEV vehicles at the existing distribution grid now and in future scenarios for smart charging and Vehicle-to-Grid charging strategies to tailor grid congestion requirements?

1.4. Methodology

The general approach for this research is carried out in three parts, and also visually depicted in figure 2:

- 1. A statistical assessment is performed to derive mostly temporal characteristics of current BEV charging behaviour from a real-world data set of recorded charging sessions in Amsterdam.
- 2. Demand response is modelled by the making use of the scaled BEV charging sessions from the former assessment. The model schedules BEV sessions to lower peak demand periods within their predetermined charging window to minimise local grid congestion and is solved by making use of linear programming.
- 3. A business value proposition scheme from a DSO perspective is proposed that aims to highlight the logic of an electric vehicle DR business system and the interrelations necessary to create value flows. Accordingly, the widely used business model literature by [54] is adopted. By assessing different near future scenarios with varying penetration rates of BEVs, the value of demand response is computed in terms of the total peak load shed relative to the uncontrolled case, and associated charging costs.



Figure 2: Conceptual framework of the methodology used in this study

ENERGY DEMAND DATA MINING

Characterising behaviour of BEV charging by analysing recorded charging sessions is necessary to understand their potential in smart grids. To characterise power demand over time, real-world charging sessions are statistically analysed using R-language to transform the data into comprehensible information structures. Its assessment is to derive exploratory information about BEV charging behaviour of individual charging sessions in terms of spatial and temporal measures. The distributions of the most focal session characteristics, i.e. arrival-departure, charging demand, charging time, reveals information regarding the available time- and load-flexibility that is qualitatively described. Furthermore, time series measures are obtained by making use of the statistical models local regression (LOESS) and auto-correlation. These models perform statistical operations on the (univariate) time series data to capture the underlying characteristics such as trend, seasonality, periodicity, and serial correlation.

Modelling approach

Demand response operational control mechanisms can generally be divided into direct (controllable) and indirect (price-based) approaches. Either control method forward similar objectives; lower systems costs and relief the local grid of peak loads. Two of those capabilities are defined in the literature as congestion management and costs minimisation. The applications for BEV demand response in the energy system are diverging, whereas the focus is to investigate direct congestion control for local grids. In paragraph 4.3 a thorough discussion about the control mechanisms is provided, including a proposed information exchange protocol to determine optimal charging schedules. The model expresses prior mentioned objectives in a linear mathematical formulation and uses linear programming to optimise the utilisation of direct load DR for different charging strategies. This approach schedules the demand, i.e. charging-discharging rate, for aggregated charging sessions to avoid peak demand while complying to consumer needs and grid constraints.

Demand response Benefits

A growing importance of business model evaluation for the DSO in the context of intelligent charging, such as smart charging and V2G charging, is deemed essential for large scale adoption of electric vehicles. Therefore, the business model in the perspective of the DSO is conceptualised according to [54] into four essential building blocks (and nine interrelated blocks) that serve to identify the essential model issues of V2G charging. The model is subsequently only partly evaluated on value flows by using an extensive real-world data set and simulated charging strategies.

Benefits for the DSO of the simulated strategies are expressed in the percentage of peak demand that is minimised and the increased number of vehicles that can be connected simultaneously as opposed to uncontrolled charging scenario. The benefits for consumer are computed by making use of a competitive pricing scheme, in which dynamic changing prices throughout the day are derived from the Dutch EPEX Day-ahead energy market. The tariff structure used in this research is called real-time pricing (RTP) that lists the electricity price per hour of the day.

1.5. Research outline

This thesis is organised as follows. Chapter 2 provides this thesis with an explanation of power system developments and electric vehicles to provide background information, where chapter 3 explores the notion of demand response in the current Dutch system. Chapter 4 discusses characteristics of the demand response application for BEVs in the energy system and explores implementation of demand response in a smart grid context by designing a DR protocol for centralised control. Chapter 5 subsequently introduces a general business model for a DSO that is developed in the context of centralised DR for electric vehicles. Next, the focal driver in the business model for DR with BEVs in this research is peak demand minimisation that can be approached by a linear mathematical framework, which is presented in chapter 6. Chapter 7 analyses the effects and impact on the grid of BEV charging and household demand in temporal- and spatial means. Subsequently, this chapter builds forth on the former results and introduces a discussion concerning the results of the simulation. The conclusions will be presented in chapter 8. Finally, future remarks and recommendations for this study can be found in chapter 9.

2. DISTRIBUTED ENERGY NETWORK DEVELOPMENTS AND ELECTRIC VEHICLES

Current environmental and political contingencies in the power system of the Netherlands, urge associated actors to adapt and redefine operational structures of the current power system and market. A large-scale adoption of sustainable energy generation is pushed by global environmental awareness about salient effects of greenhouse gasses and incentivised by governmental organisations. The Dutch policy's objective for its energy outlook in the near future aims for an energy generation portfolio that contains at approximately twothirds renewable generated energy based on a 37% reduction target of greenhouse gas emissions by 2030 in the electricity-transportation sector [15]. As a result an increase is noticed in decentralised renewable production sources by both public and private actors (households). Furthermore, the energy transition is also denoted by a rapid increase in the adoption of electric vehicles. On that account, the institutional arrangements of the Electricity Act of 1998 (see appendix A) is outdated in our decentralising power system that now also allows for private consumer participation in the energy market under the recent EU Energy Efficiency Directive of 2012. Although, emergence of demand response policies is still limited due to amongst others high costs of associated technology and infrastructure.

Therefore, this chapter provides the reader with knowledge about the current operational state of affairs of the Dutch power system. The power system actors and their roles within the electricity sector are briefly described in 2.1 which is essential information leading up to the necessity for demand response. In addition, the current state of affairs for the electric vehicle outlook is provided in 2.2 for a prospective view on the near future.

2.1. DISTRIBUTED POWER MANAGEMENT

In the scheme below (3) the governance energy market structure is depicted that originated after the Electricity Act of 1998 (background about the development of regulatory power system arrangements can be read in A). This scheme allows non-discriminatory access to government regulated transmission and distribution by third parties through separation of transmission, distribution and production. Particularly, local distributed energy generation (DEG) is gained access to the market favoured by the Dutch government in an act to offer electricity security and a diversifying portfolio of generation.

Currently, both the transmission system operator (TSO) and the distribution system operator (DSO) are operated under governmental regulation preventing it from becoming a natural monopoly. The distribution System Operator (DSO) came under state regulation after the Electricity Act of 1998, to ensure reliable and secure power delivery from the Transmission System Operator to consumers. These operations must meet requirements for power quality, such as: specific voltages, frequency balance, real and reactive power flows amongst others [34, 76]. However, both generation and retail currently operate under privatised sectors to allow for free competition. Operating in a competitive market, and the fact that from 2004 on households could choose their energy provider, enabled competitive pricing for energy.



Figure 3: Energy system control structure after Electricity Act of 1998 [3, 36, 42, 50]

In these privatised sectors, a few large commercial and industrial consumers with constant energy demand make contractual agreements with generators, that can extent for years. Smaller-scale consumers in the residential sector for example, operate on the energy market through their utility provider. Generators therefore offer their estimated production of energy on the market for a price per volume. In turn, the balance responsible party (BRP) is tasked to maintain balance in demand and supply of its portfolio by making a balanced energy planning for the TSO on a national level. This planning contains the dispatching amounts, information of production and consumption, and international trade contracts.

In response to rising DEG and REG the vulnerability of a balanced power system increases if the market does not ensure sufficient supply [6]. Multiple benefits are attached to distributed generation such as an increase in zero-emission renewable energy generation. Also, if distributed generation is connected closer to consumption sources, transmission and distribution require significantly reduced investments due to local consumptiongeneration control [1, 42]. On the contrary, distributed systems also endure larger frequency fluctuations compared to centralised power systems. Therefore, to allow these sources to develop and increase their penetration, capacity mechanisms are frequently considered to cover energy demand when distributed sources are temporarily not sufficiently available [6]. Capacity mechanisms are referred to as policy measures that ensure sufficient investment in generation capacity, which will lead to an efficient energy market operation [6]. These capacity mechanisms, e.g. ancillary services are out of the scope of this research and will therefore be left out. Further downstream at the distribution level, congestion control is one of the methods to alleviate peak demand that may be introduced by unanticipated high demand or low supply. The section below presents a brief description about congestion control.

2.1.1. Congestion Management

If more energy is produced and offered at the energy market than the grid could transport at a certain period, there is a phenomenon called congestion, overload on the distribution/transmission grid. Vice versa, when demand exceeds technical grid constraints congestion also exists. The imposed tasks of the DSO are directed to deliver electricity through a reliable, secure and cost efficient distribution grid to its end-consumers. The above described capacity mechanisms can also be applied further downstream in the distribution grid, by contracting aggregators for example [6]. This mediating actor couples the consumers (prosumers) that provide flexibility for the DSO or BRP.

Deployment of demand response flexibility for congestion management allows the DSO to operate between the capacity constraints of the grid [32]. DR offers significant benefits to the DSO, mainly to reduce both capital- and operating expenditures such as delayed investments in grid reinforcements [34, 50]. As such, these measures also decrease the possibility for a power failure and reduces peak demand. The next section introduces the relevance of electric vehicles both as a means to transportation and the growing impact they may have on the power system by having the possibility to actively cooperate with the DSO to prevent congestion in local grids.

2.2. Market trends of electric vehicles

Technology development of the transport sector to achieve climate goals has become imperative. The automotive transport sector has grown steadily in size the past decades and will keep on growing in the future. Increasing harmful gas emissions impose negative implications related to this increase in mobility demand. While oil resources still fuel the majority of the automotive sector, related emissions make up for 23% of the total global energy related emissions. In addition, the transport sector has not developed the same decline in emissions as other sectors have [1]. As agreed in the Paris agreement global concerns address the importance to decarbonise the transport sector and propose solutions by integrating electric vehicles [72]. Associated GHG emissions dependent on the BEV market share and emissions of power generation. Based on economic and environmental improvement for a low-emission automotive sector, consumers could benefit from innovated infrastructures for alternative energy sources, decreased usage costs, and low emission electric vehicles [26, 63].

The development and uptake of electric vehicles on the road in the Netherlands has risen to a total of 26.000 BEVs and 98.000 Plug-in Hybrid Electric Vehicles (PHEVs) on the road by April 2018. In contrast with the total amount of passenger cars in the Netherlands, 8.373.200, this is a mere 1.5% BEV penetration approximately [9]. Electric vehicles statistics for Amsterdam in [52], show a sharp increase in environmental friendly vehicles since they started collecting data collection in 2013. By the first measurements only 3505 PHEVs and 573 EVs were present, while in 2017 the total compiles to 5242 PHEVs and 5014 EVs, that is 4.4% of the total vehicles in Amsterdam [52]. The increase of the penetration rate of BEVs in general can be accounted to multiple factors. Mostly used measures that explain penetration are environmental objectives and subsidies of governmental organisations, and technological standards. These latter consist of infrastructural developments in charging points, or BEV typical innovations that decrease battery costs, increase range, and enable to charge through different charging technologies [29]

However, this increase in BEV penetration is not without implications for the energy system that endures increasing electricity demand. Estimated global electricity demand from electric vehicles has increased 21% from 2016 to 2017 and is forecasted to increase in the near future under the current (global) policies. In order to accommodate the increasing requirement for energy the charging infrastructure ought to grow contingently to the BEV market share. In the figure below typical values are predicted for the amount of BEVs per electric vehicle supply equipment (charging point) now, and with the EU 2020 target:



Figure 4: Ratio public charging points per BEV [29]

The EU 2020 target assumes that under the EU AFI Directive, member states have ensured that every charging point at most covers ten BEVs within sufficient proximity. This ratio may be lower than expected because of the faster uptake of BEVs than deployment of charging points, as is now apparent in Norway. If the ratio decreases, the average occupancy may increase that causes an increase in charging demand. This however is probable for the Netherlands as well [23]. According to [23], however, it is reasonable to assume this ratio can be achieved by 2020. Although, if average charging point occupancy increases the potential of electric vehicles to be used as flexibility source decreases. This effect can be explained by the fact that there exists merely flexibility if an electric vehicle exhibits a connection time greater than its theoretical required charging time, which dissipates if average occupancy per charging point increases. More about the current ratio of charging time to connection time can be read in appendix C.

2.2.1. Implications for the distribution grid

Concluding, in response to rising DEG and REG the vulnerability of a balanced power system in-creases if the market does not ensure sufficient supply, which remains a challenge due to the uncertainty in prediction of REG. Furthermore, due to the electrification of households and mobility, congestion overloads can be expected in the future in local distribution grids. Therefore, a possible candidate to control demand from the bottom-up (from the prosumer side) might be a cost efficient solution to aid in congestion problems. Deployment of demand response flexibility for congestion management allows the DSO to operate between the capacity constraints of the grid [34]. DR offers significant benefits to the DSO, mainly to reduce both capital- and operating expenditures such as delayed depreciation of distributional grid assets [36, 52].

3. Demand Response management

Implementation of bottom-up controlled electricity demand and delivery in real time is currently being adopted in the Dutch power system and is usually denoted as: smart grids. The use of smart-grid technologies in the power system introduces opportunities for effective operation of electricity distribution that are pushed by economical and technological, and political pillars. While an increase in DEG, REG, BEVs penetration rate is observed, operation of electricity in distributed grids is significantly affected. Providing reliable electricity supply to consumers will be a challenge in the near future and calls for a requirements of flexible storage while dealing with limiting distribution capacities. One of the promising and cost effective methods enabled by smart grids, is to control electricity demand from the bottom-up, i.e. demand load alteration induced by consumers in respond to prices or exerting direct load control by an aggregator. In very short time spans, this can result in meeting network capacity constraints [75, 17, 22, 64, 76]. A recent application of demand response in the Netherlands incorporating home energy management systems in households that react to energy system operating states, was temporarily implemented as a pilot project to assess the technical and economical viability. The general market framework that couples participating actors reliant on and participating in demand response is the USEF, which is a collaboration between multiple energy market actors. This framework potentially provides a promising starting point to include electric vehicles in DR and is shortly addressed in the last paragraph of this chapter.

In the first two paragraphs of this chapter, paragraph 3.1 reviews the possibilities concerning implementation and the efficacy of general demand response in our current power system through a so-called smart-grid interface. Dynamic pricing tariffs in DR are seen as the engaging market mechanisms to allow distributed electricity control, which are presented in 3.2. Paragraph 3.3 describes the USEF framework, that is potentially a base for demand response with BEVs.
3.1. Energy market developments and the decentralisation paradigm

Private investments in distributed energy resources, renewables specifically, has mainly been driven the past years by governmental support and tax incentives [79]. Benefits of distributed generation is clearly present in a higher implementation of renewables, decreased transmission losses, and lower investments in transmission and distribution networks [64, 19, 33]. While small-scale distributed generation is made accessible for the consumer's public, an increase in active participation of the energy supply side with conventional energy actors will result in improved cost competitiveness for both the consumers and other related actors in the energy market. A change in a consumer participation movement that continues to accelerate by decreasing investment costs for distributed resources [34]. Now that consumers have the possibility to participate in the electricity market as producers and adjust their demand according to prices and generation, new markets and new products emerge such as 'smart-grids' and 'demand side management' [66, 14, 64, 33, 75]. However, it is still believed demand response potential in the EU energy market remains underutilised as of today, while it presents promising applications.

Smart grids

Today's energy decentralisation paradigm calls for the introduction of so-called 'smart grids', an interoperable way to integrate control over electricity consumption and delivery in real-time to interactive and responsive consumers. Smart grids are realised with smart metering installations and ICT communication technologies to be worthwhile for distributed electricity management. In smart grids consumers (prosumers) can modify their demand behaviour according to (price) incentives and thereby realise improvement of electricity reliability and energy-cost efficiency [64]. In practice home energy management systems (HEMS) receive real-time information regarding pricing tariffs (that can be day-ahead, peak power loads) that incentivise efficient and effective use of electricity demand by an aggregator or consumer. The communication standards relevant for DR implementation are not further discussed in this research, but can be found in [26, 64].

Demand response management

As an intro, prior to the review for demand response opportunities specifically tailed for BEVs (in 4.3) the general definition of DR is given below that confers the benefits in the power system. In the context of increasing distributed electricity resources and growing electricity demand, reliable supply requires a back-up for flexible power generation given limited transmission and distribution capacity. A mechanism denoted as demand response ensures changes in consumer's demand patterns by delivering timeand load- flexibility though intermediaries on a local scale to enhance system electricity coordination, and is usually quantified with four parameters [63, 17, 75]:

- 1. Direction of flexible capacity (upward or downward)
- 2. Size of capacity
- 3. Time at which the demand response service is required, and the duration of the provided demand response
- 4. Location

Usually, demand response is deployed through use of different resource types, such as home energy management systems, storage (BEVs), and distributed generation (renewable sources) to mainly lower peak loads (both demand and supply) in local grids or to provide DR in ancillary services to balance deviation. Consumers can interact through intermediaries known as aggregators that enable offers of flexibility on one of the wholesale electricity markets. On the contrary, consumers can also be approached indirectly through price-mechanisms that induce incentives to lower electricity costs. Participation at the demand side has various benefits in the energy system, the most noteworthy can be summed up to but are not limited to [66, 39, 40]:

- Consumers have the ability to shift loads in periods of peak demands where high prices are charged, to periods of low demand and hence low prices.
- Shifting loads result of aggregated consumption to relieve the local grid from overloads, and reduces overall electricity transportation/distribution costs.
- Minimisation of electricity consumption costs
- Production companies are limited in their market power due to higher system competition.
- Higher exploitation of renewable energy generation by matching demand with generation

Demand response schemes are distinguished by the various motivation methods offered to the participating consumers. Programs include in general two control methods, centralised direct load control (possibly with multi-aggregators) or time-based and incentivebased DR. Depending on the type of customer, the former is often used for private, residential consumers, while the latter suits large industrial consumers but is not limited to that particular type. Methods for centralised approaches comprise of direct load control by directly modifying consumption of consumers [75, 17, 44], to sustain power system reliability.

On the other hand, incentive-based DR programs are driven by power system stress and consequently produce a price signal to consumers in order to restore the system requirements. Therefore, the system stress is correlated to time-varying prices accordingly, and introduces incentives to consumers to lower electricity use. These programs are marketbased, as a DSO, aggregator, retailer or other actor can place offers or bids of flexibility on either the wholesale or retail market. In turn, consumers receive remunerations according to their performance of reducing an amount of electricity demand by their retailer beforehand. Consumer participation is promoted by inducing a voluntarily choice to react to price signals for which they can lower their bills. [44]. However, a vision where the consumer is its own manager of energy consumption, or potentially generation, does not necessarily solve grid technical implications that correspond with the interests of energy actors as of today [21]. Therefore, to enable substantial modification in consumption behaviour, compensation schemes are required that integrate dynamic-based electricity tariffs for either consumption or provision of flexibility back into the grid. The following section describes the most frequently used tariff structures in the literature.

3.2. Dynamic demand response pricing

Currently, consumer electricity prices are are set by the utility provider into uniform consumer tariffs and charged the amount of demand that is acquired (\in /kWh) and can be distinguished into simple tariff, flat rate tariff, two part tariff, maximum demand tariff, etc. These tariff structures do not reflect a fair pricing mechanism, because they consist of preset prices and do not take account for real-time market related electricity charges [66, 39]. In demand response programs, consumers are considered price-takers who exhibit the means to adjust their demand by being allowed to place bids for energy on the retail market. Mediated by an central aggregator or utility provider. As of today, the UTP charges the consumer through one of the simple tariff structures earlier mentioned, while within a smart grid network prices may need to be coherent to the nature of the costs (time dependent) [33, 34].

On the retail energy market offers and bids for capacity regulation are usually priced in relation to the volume of capacity. This does not ensure fair electricity prices in a system that is denoted by a growing share of renewable energy generation. The results of [66, 34] are in accordance with this statement, where the authors assessed the effectiveness of conventional bids for capacity on the day-ahead market with different characteristics. The results suggest conventional price-volume bids are more often declined and leave a fraction of the total required energy unsatisfied, compared to demand-shifting bids. The latter is more effective in both cost of consumption and management of unsatisfied demand. Demand shifting is becoming increasingly used to balance low-level demand response by use of households and BEVs, which makes a similar price system particularly suitable.

Several electricity tariffs can be implemented for either control mechanisms, that are responsive to to changes in wholesale electricity prices over time. The most adopted schemes for residential and BEV consumers include:

- Real-time-pricing (RTP) the utility provider announces the electricity pricing on a quarter-hourly/hourly basis
- Time-of-use (TOU) preset prices for time periods during the day
- Critical peak pricing (CPP) utility provider can set higher prices in times of high demand to incentivise load decrease.
- Peak-time rebate PTR consumers are incetivised to react to on top of their standard tariff, receive a rebate for load reduction they provide.

Two of the most frequent used pricing schemes, time-of-use and real-time-pricing, are elaborated on below. Time-of-use pricing is frequently used to induce a change in consumption patterns for distributional grid purposes, for two apparent reasons. 1) Due to its relative low requirements to implement in the energy market, as usually only at two peak times during the day prices are increased, and 2) because these prices explicitly reflect distributional grid states. The authors of [32] & [44] implement actual TOU schemes that are combined with separate demand and energy charges , although this scheme requires less communication due to its preset prices, but on the contrary does not reflect real-time fair prices, and also possibly discriminates prospective consumers based on varying prices for peak stress in the distribution grid on a specific location and time [75].

On the other hand in a Real-Time-Pricing scheme, electricity prices are based on dayahead energy prices, usually in hourly periods, multiplied by a specific factor to resemble the variation in price, whereas the average price equals the traditional charged tariffs. These prices best reflect the costs incurred by the UTP/retailer to charge their consumers [33]. This scheme relies on a near real-time communication flow between the consumers and the UP. While some large industrial consumers use this scheme, household consumers are careful in adaptation due to higher risk in price volatility due to a negative tendency to make system related decisions [75]. As opposed to these associated drawbacks, centrally activated DR could diminish these risks and potentially offer fair retail prices in respond to short periodic changes of the wholesale price. That is, the higher the participation, the higher the reduction in costs. For RTP on the day-ahead market, as used in this research, aggregators (consumers) make schedule decisions to procure expected electricity consumption from the retailer according to prices every hour for the next day (max 24h) prior to the energy market closing time, i.e. 12:00. During the auction day, the retailers' bids are matched and the day-ahead price for the predicted consumption is charged. One of the possible drawbacks that occurs due to uncertainty in demand prediction, may lead to deviation in the day-ahead schedules, for which energy is charged based on real time electricity rates. Case studies from multiple literature sources as reviewed in [75] show DR benefits in decreased demand and power losses and a lower peak-to-peak distance by incorporation of this tariff-scheme.

3.3. Demand response facilitating USEF framework

USEF is a collaborative foundation between Liander, Stedin, IBM, ICT Group, ABB, DNV-GL. USEF describes a standard for unlocking the value of energy flexibility by making it a tradeable commodity, delivering a new market structure with associated mechanisms and rules to make flexibility trading effective. These demand response mechanisms allow prosumers to alter their consumption patterns and contribute to actively decrease system demand and help restore grid imbalance during power deviations or congestion [31]. Drivers for participation are dynamical electricity prices and remunerations for load reduction. Control of DR in the USEF role model requires a new actor, called an aggregator, who acts as third party to connect consumers/prosumers into the energy system with either the DSO or BRP to deliver flexibility [23]



Figure 5: USEF flexibility market structure [74]

The USEF is a mere framework to facilitate the aggregator in selling flexibility from prosumers to DSO/BRP parties for maintaining grid constraints on wholesale and balancing markets. Research by [1] showed how the BRP can use flexibility to trade on the unbalance Day-ahead market to benefit from both a decrease in cost of electricity and restore balance in the power system. For participation in the energy markets and to place capacity bids, rules restrict capacity bids below a threshold, an aggregation of individual consumers is necessary to be in compliance with the market bidding rules. Due to smart grid technologies, active participation from the prosumer side could now be facilitated.

For example, if a demand peak occurs either the DSO or the BRP call to order for flexibility from the aggregator. In this process the aggregator is required to make forecasts for the amount of flexibility that can be offered for bids on the Day-Ahead/Intraday market. Subsequently the aggregator operates under two strategies, 1) by controlling smart appliances in Home Energy Management (HEM) systems according to meet the required power arrangements with DSO/BRP, 2) or by forecasting the expected residential consumption, that inherently has a load forecast error and hence is to be balanced with required flexibility. The forecast error either contributes or opposes flexibility dependent on the sign of the error [31].

Results show that Liander's electricity distribution grid in the municipality of Amsterdam is expected to experience increased congestion at local grid connections, due to increased PV solar generation, BEV penetration and household heat pumps [1]. Liander expects additional grid reinforcement costs, on national level, to be accounted for 2-5% every year up to 2030, and 7% every year from 2030-2050. Furthermore, while demand response curtails grid overloads, gross capital investments can save up to \notin 700M up to 2050. However, a thorough investigation where electric vehicles are introduced to participate in this framework by utilising new charging technology standards is not performed, hence the following chapter complements with approaches for BEV integration.

3.4. Implications for demand response management

Application of demand response in a distribution network is currently one of the methods to increase economic efficiency and effectivenes of grid operational controls. The trend towards using hourly pricing tariffs can incentivise the BEV owner's demand request to be shifted towards hours with low electricity prices and hence also less energy demand. Therefore, to enable substantial modification in consumption behaviour, compensation schemes are required that integrate dynamic-based electricity tariffs for either consumption or provision of flexibility back into the grid. The most focal benefits o fDR that potentially arise in the distributional power system include:

- Shifting loads result of aggregated consumption to relieve the local grid from overloads, and reduces overall electricity transportation/distribution costs for the DSO
- Minimisation of electricity consumption costs for the consumer
- Higher exploitation of renewable energy generation by matching demand with generation

The recent developed USEF energy market framework allows for application of demand response by making flexibility a tradeable commodity on energy markets. Furthermore, while demand response curtails grid overloads, gross capital investments in local distribution grids can save up to \notin 700M up to 2050 for Liander's network. Thus, this research focusses on a (multi-aggregator) centralised approach of directly controlling and optimising the consumers' energy demand by introducing a planning protocol that may be integrated in the USEF framework.

4. Demand response with smart charging and V2G charging in local grids

Demand response resource types are, broadly speaking, control schemes that either manage energy consumption by delay, or use energy storage devices to temporarily capture capacity. The battery of a BEV is essentially a flexible storage unit, and although its use is transportation, the battery can be utilised for ancillary grid functions. The inherent potential of zero-emission vehicles as an alternative to Internal Combustion Engine (ICE) vehicles can only be fully ensured if BEVs are charged with REG resources, having a well-to-wheel emission of zero. However, BEV energy demand may be out of synchronisation with REG generation unable to use this generated energy decreasing its economical benefits. In addition, when BEV market share grows an increase in energy demand is expected that may potentially enlarge peak loads during existing peak demand hours. [13] predicted in their real-time modelling approach, under an increase of BEV penetration from 16% to 63%, significant voltage deviations, high power losses and expensive costs in ancillary generation. If as a consequence capacity mechanisms have to be used to cover the demand that is usually generated by fossil-fuel based units, this potential is waisted [6]. Additionally downstream in the distribution grid, capital asset expenditures require an expansion in this situations as well.

The structure of this chapter therefore serves as a constructive review that describes the approach to incorporate a management control strategy in which electric vehicles play a role in providing grid operational aid. After the introduction in paragraph 4.1, a review is provided that highlights the distinction in demand forecasting methods for different classes of data, for which each class may serve a specific application in the energy system. Paragraph 4.3 addresses DR control methods and gives the information exchange protocol between the prosumer, aggregator, and DSO for a centralised DR approach to unlock flexibility from electric vehicles.

4.1. Electric vehicle demand management

Managing growing BEV energy demand can significantly reduce grid investments and additional generation capacity, as well as an increase of REG. Currently as a result of the EU Energy Efficiency Directive of 2012, an institutional base is presented for development initiatives of demand response in Europe [17]. Technical and regulatory standards now enable demand response mechanisms to curtail flexibility on the wholesale or retail energy market and allow for prosumer participation for which its purpose is to serve economically efficient and environmental friendly operational grid measures. Demand response also serves as a coupling method between wholesale and retail markets (background about energy markets is described in B) that allows fair electricity pricing in a competitive system. As described in the previous paragraph (3.1) energy market actors developed a framework to facilitate DR between the prosumer and DSO/BRP that will serve as a basis for the control and business model of electric vehicles demand response. Practically, DR in this research is used to reduce congestion in distribution grids by moving part of BEV energy demand from (evening) peaks to the afternoon or night. By achieving these measures potential benefits arise, including [29]:

- Optimising local grid assets by increasing the utilisation factor, and thereby maximise asset efficiency and subsequently decrease costs, which is beneficial for the DSO
- Reduce the need for expensive additional generation capacity and to increase use of renewable energy, by spreading the charging loads throughout the day while price arbitrage opportunities can take place. This is both beneficial for the BRP, prosumer and aggregator.

The charging technology standards, briefly introduced below, make application of DR possible. However, in order to operationalise DR with BEVs it is imperative to know the charging demand characteristics to assess impact on the power system. The next paragraph reviews different models used in the literature that assess and predict charging demand.

4.1.1. Charging technology characteristics

The Dutch Ministry of Economy, Climate, and Infrastructure of the Netherlands has addressed budgetary outlay of 7.2 million Euro the past five years to stimulate development and uptake of charging infrastructure in the Netherlands [61]. The amount of charging points is increasing at a steady rate up to 17.500 public and 17.400 semi-public charging points in 2018. The amount of private charging points is estimated to be 84.000. This research applies a case study to analyse BEV charging behaviour for the municipality of Amsterdam for which a historic data set of charging sessions from 1832 charging points is recorded. In contrast Amsterdam currently resides 3761 public and semi-public charging points. In the upcoming near future, technical standards develop to potentially become a mature and economical part of the infrastructure, increasing the importance to direct research towards implementation of smart-charging or V2G charging points.

In Europe the charging standard IEC 62196 is adopted, that proposes three different modes of charging, to be distinguished into slow-fast charging and with-without communication availability to control the battery management and charging rate. The charging strategies considered in this research are:

• Regular uncontrolled charging

Uncontrollable charging is the strategy generally considered as the standard, where charging occurs immediately when BEVs connect to the charging point until the battery reached its requested capacity while the charging rate is constant and depends on the charging point class.

• Smart-charging

Smart charging is a unilateral charging strategy in which there is an indirect- or direct control with the BEV prosumer possible to schedule consumption to current local grid and electricity cost parameters. This is considered an unidirectional charging strategy, where regulation up- & down can be enabled by either increasing or decreasing the charging rate [47]. Regulation up- & down refers to the under- or over-estimation of energy supply respectively, inverted reciprocal to the demand.

• Vehicle-to-Grid

Vehicle-to-Grid charging strategy is similar to smart charging but additionally involves bilateral power flow between the BEVs and charging points [26]. The efficacy of V2G is expected to increase the provision of flexibility to the grid in charging-discharging as compared to unidirectional charging. With V2G more percentage of load shed can be realised than with smart charging, because peak energy flows can not only be re-scheduled, but also be decreased by discharging [47].

4.2. Classification of models for BEV charging behaviour and consumption pattern forecasting

Understanding BEV charging behaviour in terms of user's energy demand in an urban charging infrastructure is imperative to assess impact on a network level. Combined electricity consumption behaviour by residents at certain periods throughout the day that coincides with BEV consumption can temporarily lead to a shortage of distribution capacity due to local peak demands [41, 2, 82, 64]. It is found in [22] that depending on the charge point output, a BEV usually has a demand more than twice as large as a residential demand. The near future scenario about the electrification of the society imposes realistic challenges to manage a significant increase in energy demand. The importance of forecasting future electricity consumption of BEVs cannot be stressed enough although the implications on the local distribution grid are not explicitly apparent currently. Therefore, the following section of this paragraph shortly explores forecasting techniques for BEV demand with different classes of data and different purposes.

CHARGING SESSION FORECASTING TECHNIQUES

Literature about works of demand response for electric vehicles exists in abundance in which many of those works address mathematical modelling approaches to predict electric vehicles use patterns based on influential variables such as connection time, charging point occupancy, charging demand economic factors etc. Literature is roughly divided into deterministic and stochastic load forecasting [5, 26] to either design long-term policy measures or use short-term forecasts for grid operational measures. Because the variables exhibit stochastic randomness, to be able to fully capture and forecast the charging behaviour and generate energy consumption profiles, most commonly applied techniques are classified in the categories statistical and artificial intelligence based modelling, dependent on the applications in the energy system and the time window for which the forecast application serves[33]:

- Statistical based modelling: Multiple regression analysis, Auto Regressive, Moving average, State space, Kahlman filter, exponential smoothing
- Stochastic time series: Monte Carlo, Markov chain
- Time series decomposition, auto-correlation
- Artificial intelligence: Artificial Neural Network, fuzzy logic
- Heuristic optimisation based on knowledge

The distinction in performance of modelling of the two categories lies between the formulation of the mathematical equations. For statistical modelling, mathematical equations are generally linear and tend to neglect the natural non-linearity of the data, whereas artificial intelligence based models have the ability to deal with this non-linearity and predict with more accuracy energy demand requirements [82, 30, 12, 30].

ANNUAL LONG-TERM FORECASTING MODELS FOR POLICY MEASURES

Early mathematical models that assess charging behaviour impact on the grid for different deployment scenarios, can be summarised by annual vehicle use and car ownership of conventional ICE vehicles. Usually its application lies in development of long-term policy measures, such as GHG emmission outlook and energy security analyses. Unfortunately, metropolitan travel survey data do not capture the difference in effect of charging demand on infrastructural factors and travel patterns specifically for BEVs [11]. Assuming travel patterns remain unchanged raises concerns about the applicability in describing BEV charging behaviour due to the fact that BEVs have range limitations and are confined in limited charging possibilities [11, 82]. However, an increasing amount of real-world electric mobility data enables a classification of forecasting methods along a finer grained time dimension resulting in accurate predictions.

SHORT-PERIOD DEMAND FORECASTING MODELS

The purpose of short-period models is to forecast peak loads or congestion phenomena. This is key in the determination for operation and loads planning in a power system for utility providers or grid operators. Short period modelling uses smaller time resolution data of for example daily activity travel schedules and daily mileage driven (of conventional vehicles) which are better usable constructs to assess demand behaviour. With these assessments, demand behaviour prediction can be used for operational tasks and planning of energy in a power system [33]. However, this kind of statistics data about conventional travel demand for arrival times and distance travelled, strongly limits the accuracy to model BEV use for operational grid planning tools due to inaccurate data that don't resemble usage behaviour of electric vehicles. In the work of [44] for example, the authors use arrival time statistics of conventional vehicles and translate those parameters to charging tendencies of individual vehicles. Although they fit a regression model to the distributions of correlated variables, their accuracy in representative prediction of BEV use lacks replicability and rigor because averaged values for charging demand are used. This is a natural consequence in using fixed or preset factors for charging demand, charging output and battery capacity.

Currently, due to the availability of a growing amount of short-period BEV charging session data in urban areas, attributed with temporal-spatial measures, BEV patterns can be explicitly derived from individual charging sessions that truly capture the characteristics [62, 56]. These finer grained period models exhibit information to forecast BEV's associated load demand on local grid level and allow expansion to network level.

Other methods model electric vehicle demand describing the sequence of states a BEV resides in, for which the probability of the current state depends solely on the previous state. A state can be defined by driving, idling, or charging. Assumed that probability of every state is based on SOC, average driving distance. For example, the studies of [55, 82, 31, 5] model a probabilistic decision making process (e.g. Markov chain) by

prediting when a BEV should be charged. In some situations, the arrival SOC may be high enough and the charging session periodicity frequent such that planning of charging demand is facilitated by making the decision not to charge. According the parameters of SOC and future trip predictions, the current level of SOC ought to suffice for these trips. Usually, the data used is based on non-gaussian multivariate distributions, describing the correlation between the charging session variables. Despite the ability to generate consistent BEV patterns over long time periods, these models neglect the state responsiveness to dynamic electricity tariffs and other non-predetermined behaviour. Thus limiting in implementation for grid operational energy planning.

By using real-world charging session data most research converges to solve the following general planning challenges for operators that manage energy demand[12, 67, 64]:

- Assess if current electricity generation is sufficient to provide for the additional BEV demand
- Evaluate marginal electricity costs for smart charging/V2G charging BEV demand for various ancillary services
- Assess if growing demand due to increasing BEV penetration can create congestion and voltage deviations at local grids
- Evaluate the benefits of DR by prediction of BEV charging profiles

4.3. Electric vehicle flexibility control strategies

Different grid operational aiding tasks can be enabled through either central or decentralised energy scheduling approaches, each with its inherent pros and cons. Approaches are generally distinguished in two classes of optimisation goals, i.e. 1) maximising (operational) benefits of the grid-side, and 2) maximising charging cost benefits at the BEV-side [47]. DR flexibility can only be enabled by models that assess decisions tools concerning management of energy in a local distribution grid. Therefore, the scheduling problem of charging sessions is modelled as an optimisation problem where the formulation of objective functions depend on a threefold of demand response model characteristics. That is, 1) prosumer interaction, 2) optimisation method, and 3) time-scale [19].

The first characteristic describes the prosumer interaction and is either decentralised, meaning individual prosumer response without coordination, or centralised where cooperative prosumers are controlled by a multi-aggregator. The second characteristic differentiates between deterministic approaches that assume the current prosumer state depends only on the previous state, and stochastic approaches that consider real-world phenomena to be stochastic of nature when the samples are large enough [4]. The third and last characteristic relates to the time-scale required for the optimisation and is either day-ahead or real time. As both grid-side and BEV-side objectives are modelled in this research, the next section in the paragraph below is a brief review on the different control approaches to manage charging of BEVs for grid operational tasks and the abstraction into mathematical modelling required to describe and manage aforementioned goals.

CHARGING CONTROL MODELS

The most frequently adopted optimisation models or algorithms in current literature that are capable of coordinated BEV charging in a distribution network are summarised below and are ranked according to increasing computation time, problem complexity, and problem convexity (optimal/sub-optimal solutions).

- (Integer) linear programming
- Markov chain
- Mixed integer programming (fast computation time, global optimum)
- Non/-/convex optimisation problem (able to handle complex systems)
- Quadratic programming
- Non-linear programming (large computation time)
- Heuristic neural network models
- Game theory (interaction between different sources of DR in coordinated applications)
- Multi-agent systems

The most focal challenges surrounding planning of energy requirements of the prosumers can be distinguished into the following terms: *interoperability, scalability, behaviour, security and privacy,* [33]. While the scope of this research is limited to the first three challenges, a study towards security and privacy is therefore not included. Dependent on the models mentioned above and the control structure of DR, each approach scores differently on these terms. The section below reviews the pros and cons of the control method to enable flexibility at BEVs.

CENTRALISED- AND DECENTRALISED CONTROL

More research effort is currently dedicated to tackle distribution grid related problems where DR approaches for electric vehicles can roughly be distinguished in centralised and decentralised optimisation strategies [86, 44, 19, 22], as been shortly introduced in the previous chapter. Main objectives in DR at the grid-side comprise of minimising costs of supply, enhance economical- and technical distribution of grid operations and maximising integration of thermal-, wind-, & solar plants to accommodate for stochastic generation uncertainty and thereby enhance utilised renewable generated energy. On the prosumer-side of the market, participation in DR can be motivated by objectives that aim for a minimisation of charging costs, increasing user convenience and minimising GHG emissions [38, 37, 59, 8, 25, 47].

In decentralised planning, [86] proposed a decentralised random access framework in which smart agents reschedule their own charging independently, according to information about grid and costs status received from the DSO (to aid congestion). Although they claim their model does not need accurate predictions on BEV load or arrival time, it is complex to extend its scale for high BEV penetration, due to the lack of a priori prosumer and distribution grid parameters. It may be that decentralised control decreases the privacy concerns prosumers have by letting their BEV charging be controlled by an aggregator, but prosumers are currently not actively engaged in the smart grid [21, 47].

A centralised control as opposed to decentralised control requires high level control of decisions for which low level (prosumers) demand is explicitly in the hands of the central actor. Subsequently, this leaves the central actor with the choice to provide the flexibility for a specific purpose and actor in the energy system. While those purposes can roughly be distinguished into up-stream (high-voltage network, TSO) or down-stream (low-voltage network, DSO) services, either incurred provision of DR can be in conflict with operational tasks of the different actors in power system. Hence, a coordination strategy that incorporates multi-objectives is required if demand response is executed, because some V2G services can have conflicting interests [26, 70]. A conflict arises when two actors need the flexible load capabilities from the BEV for conflicting services, such as frequency control and peak shaving/shifting which can cause implications in a part of the power system [70]. A prioritised service list will be required when dealing with multiple actors, but is not used in this research.

Not only conflicting services must meet a trade-off prior to DR, a fair compensation scheme ideally needs to describe a maximisation of benefits from both the system operators and prosumers perspective [44]. While prosumers adopt solely cost saving strategies, as potential consequence the shifted demand may define new energy demand peaks during times with low prices. However, only a few literature works developed models that consider the optimisation of both system operators load deviation and prosumers charging costs simultaneously. The works of [44, 75, 81] stress the importance of modelling the relations between the actors' strategies to optimise a balanced system that provides a Pareto optimal solution. The model developed in [44] opts for such an approach in which the authors found a global system optimum that satisfies both the prosumer, aggregator and DSO. They formulate a Mixed Integer Linear (MILP) problem where weights are assigned to conceptualise ranks and create mutual benefits. The authors simulate the relations between strategies for each participating actors. Although this algorithm suggests a distribution grid optimisation simultaneously with driver needs (costs effective) by scheduling charging loads, this model omits uncertain conditions in the power system such as congestion network constraints.

In short, the authors of the work [60] explicitly compared a local decentralised and centralised charging strategy in distribution networks and found that local control can accommodate a larger penetration of BEVs on a feeder while requiring a smaller communication infrastructure, but suffers to keep demand within distribution grid constraints. Whereas with a centralised strategy, network constraints can be retained on an acceptable level as a more accurate schedule can be produced due to the information of both BEVs and the grid. Due to the direction of this research, in which optimising energy demand within distribution grid constraints as the most important objective, a centralised control charging is modelled.

4.4. INFORMATION EXCHANGE SCHEME FOR CENTRALISED LOAD CONTROL

The optimisation problem of V2G charging in an existing institutional framework as of today, requires aside from energy exchange also vast amounts of information exchange between the BEV, aggregator (CPOc), supplier, and the DSO. An already established communication infrastructure allows two way information exchange to offer a service related DR V2G mechanism, the actual charging demand and allocated billing according to the electricity tariff-scheme. However, research has shown signs of integration complexity in using a decentralised control for DR that can mainly be addressed by the dependency on prosumer participation and inherent uncertainty in predicting the stochastic nature of driving and charging [60, 26, 75]. Hence, in compliance with current system innovation in the energy system (USEF), as a responsible, easy to integrate, and scalable approach, this research adopts a direct control approach for multi-aggregators to trade energy an the USEF flexibility market, but differs slightly in the task protocol. This section, therefore, describes an information exchange protocol to enable DR from an aggregator perspective.

One of the direct control (centralised) methods of demand response is through aggregation of BEVs by aggregators, to perform adequate operational management and the ability to integrate various grid-providing services. The process to manage energy demand between these actors is expressed by the joint service and information relationship between the prosumer, aggregator, and DSO. The successive steps and relationships are exemplified in the scheme below. The scheme also shows the tasks that need to be executed in the process of demand response within the time window where flexibility from BEVs is required to aid in congestion control by influencing charging schedules:



Figure 6: Direct load control

• Data collection

In practice, the BEV user sends information to the aggregator regarding the charging demand, initial SOC, required SOC and departure time when it is connected to the grid at a certain location at the beginning of each hour. To design the appropriate charging schedule, the aggregator sends a request for information about grid base loads and feeder asset load boundaries to the DSO and requires day-ahead electricity pricing information from the supplier.

• Electric vehicle demand prediction

The scheduling optimisation problem requires additionally a prediction of future BEV energy demand such that flexibility can be exploited and planned a priori before arrival of new BEVs. According to historical data and current connected BEVs, charging behaviour is predicted that includes the variables mentioned above. This increases the operational planning capabilities of demand such that the aggregator can optimise associated charging costs a priori according to the hourly day-ahead prices and takes prospective BEV demand into account [32]. After day-ahead demand predictions are made, this schedule is sent to the DSO, which subsequently assesses the expected periods for congestion. In addition for every successive, newly connected BEV, the charging point extracts the unknown BEV energy demand characteristics, initial SOC and updates the DSO with the energy usage to compute the potential flexibility capacity.

• Charging schedule optimisation

The aggregator is responsible for charging all individual BEVs while it is the responsibility of the DSO for retaining efficient and reliable grid operation within the system constraints. Every beginning of each hour, the aggregator optimises BEV charging schedules to minimise peak loads in the local grid - based on the base load input of the DSO - and respects BEV prosumers' energy needs while simultaneously minimising the charging associated energy costs for both the prosumer and CPO until an equilibrium is achieved. The details of each optimisation step are as follows:

1. The first step in this process requires an optimisation problem that uses the supplier's day-ahead pricing information to minimise charging costs by scheduling charging rates of all connected BEVs while taking into account future predicted BEV demand. Hence, chapter 7 assesses a prediction of BEV demand behaviour. Although, actual price minimisation is not considered in this research, the resulting electricity costs in the charging schedule optimisation are analysed for the simulated case studies by using the RTP tariff scheme.

- 2. Moreover, the second step includes sending these BEV demand profiles to the DSO that performs a check whether the determined BEV profiles and it's own baseline demand prediction fall within the grid constraints. If by newly connected BEVs or an unanticipated high base load the grid constraints are not in compliance with the limits, the DSO can offer a request for flexibility. If accepted by the aggregator, this task is the second optimisation that iterates until the scheduled profiles are in accordance with the grid constraints and simultaneously uses 'cheap' hours for charging/discharging as the second constraint. Meaning that the aggregator determines the charging/discharging rates to fulfil the flexibility offer for maintaining grid operations, for which it receives monetary compensation from the DSO and uses day-ahead pricing to lower associated charging costs.
- 3. For the last step, the aggregator submits hourly electricity offers to in accordance with their BEV demand predictions and sends this information to the supplier. Next, for each receding hour the supplier receives the demand request from the aggregator and charges energy costs equal to the marginal electricity price on the energy market. Finally, the resulting charging schedules are communicated back to the BEV. In addition this reduces the congestion that is a result of misalignment in the energy curtailed by the aggregator from the supplier and the actual consumed energy. If the aggregator remains with flexibility after complying to the DSO's requirements, it can be offered to the BRP or TSO for balancing services. [47, 1].

In conclusion, this direct load control approach is for the most part complementary to the operations expected from an aggregator in the context of the USEF framework. The difference in this control protocol is the aggregator's electricity pricing information request from the supplier. Although, the aggregator is decoupled from actual energy supply that is governed by the UTP/supplier, a day-ahead pricing request allows for cost minimisation for the aggregator where it balances the difference in variation of its predicted- and actual demand schedule. According to this "shared" actor objective optimisation, the value of demand response shall be shared between the prosumer, aggregator and DSO which mainly serves the purpose to delay expansion of the distribution grid and minimise electricity costs. The next chapter introduces the business model from a DSO perspective that conveys the possible value of V2G charging.

4.5. Implications for model design

Potential economic and power system impacts of demand response integration for BEVs is studied for various applications in the energy system. Whether controlled charging and discharging strategies are applied to provide ancillary balance power further upstream in the energy system (to improve REG integration), or applied to aid congestion on a local level, all strategies are beneficiary to the grid's operation and the prosumer.

- Reliability, stability, security of electricity in decentralised grids are all grid operational objectives in which demand response is an effective method to realise these objectives.
- Thus, to accurately analyse urban BEV charging demand behaviour, reliability and consistency of prediction increases if large-scale data sets are used for analysis of typical trends in energy demand.
- Application of BEV flexibility in demand response schemes is highly dependent on the specific task it will be utilised for, and hence its value proposition is diverging for different actors in the energy system, which can be dealt with by value-stacking.
- In addition, if active prosumer participation in the energy system remains a reverse salient, indirect or incentive price-based schemes could hamper the full utilisation of the flexibility BEVs potentially have. Therefore, a centralised direct load control method is currently in the Dutch smart grid environment easy to integrate together with a RTP pricing scheme to reflect fair electricity tariffs.
- The various options in strategies to provide demand response in different parts of the power system induce potential conflicting interests between energy actors (i.e. TSO, DSO, CPO, prosumer). Within a given conflicting situation demand response should be activated for operations with higher priority. However, this research limits the scope to applications of BEV demand response for separate system operator congestion prevention and charging cost minimisation, and does not further elaborate on conflicting situations.
- Various studies developed operational control models of DR with BEVs for different actors in the energy system with balanced objectives. Focussing on the gap between multiple interests of system operators and BEV prosumers simultaneously, ensures a mutually beneficial equilibrium. Different actor states are compared and their strategies related by weighting functions. Because this falls out of the scope of this study, further research on concerning this subject is necessary.

5. Demand response smart charging and V2G charging economics

Management of charging processes of electric vehicles is generally considered to be a forthcoming and promising method in support of grid operations for a distributed power system. The economical value for provision of demand response for such operations - which are diverging for different required grid services of the power system - is for consumers and aggregators market driven while system operators are confined to regulated non-commercial purposes, and hence twofold in its benefits. Remuneration for grid-stabilising flexibility is paid by the DSO and ought to be shared among the consumers and the aggregators that control the dispatched flexibility. Simultaneously, the aggregator communicates with the supplier to receive hourly day-ahead electricity prices to further reduce charging costs by designing an economically efficient charging schedule. In the Dutch privatised energy sector, costs for electricity charges, network charges and taxes are separately defined and, therefore, induce multiple objectives.

A business value proposition scheme from a DSO perspective is proposed that aims to highlight the logic of an electric vehicle DR business system and the interrelations necessary to create value behind this former mentioned process. The value of DR ought to be divided among the participation of each actor in the supply chain. The following section in this chapter is dedicated to explain the business case surrounding integration of a V2G infrastructure, and the structure how value is added by incorporating a bilateral charging service to provide aid in grid congestion.

5.1. Business case for the Distribution System Operator

Recent investments in charging infrastructure, currently initiated by municipality of Amsterdam, is largely focused on design of public charging programmes, by realising investment in new charging technology standards to maximise prosumer contribution to demand response. The various DR strategies mentioned in 4.3 comment the promising benefits in monetary compensations for prospective prosumers that allow electric vehicles to be used as a buffer and thereby decrease peak load and supply in local grid through new charging standards, i.e. smart charging, V2G, and tariff-schemes. After all, vehicles are merely used for a small proportion of the day, and if connected to a charging point poses the excellent capability to return energy to the grid when it is ought to be necessary.

This paragraph elicits a business model analysis that delivers a pictorial scheme of all building model elements that constitute costs and benefits for a V2G charging infrastructure from a DSO perspective. The value flows in the schematic overview presented in paragraph 5.1 are assessed on a preliminary level in 7.4 due to uncertainties concerning the underdeveloped market conditions for DR and management of compensation schemes that come in a large variety with different cost and revenue structures [17, 8]. Remuneration to prosumers in return for delivered flexibility can be approached by adopting a variety of value flows, i.e. different contractual agreements or tariff mechanisms.

While it is assumed that prosumers are price-takers, participation in DR is significantly dependent on charging costs, and therefore imposes uncertainties on the outcome of the added value. These assumptions result in a large variance of the expected benefits. In addition, it can be argued that the aggregator as a DR enabling actor deals with so-called value-stacking challenge, where it may need to propose a trade-off between selling flexibility for differing grid operational tasks. The aggregator is a profit maximising entity as it operates in a competitive market and as a result chooses to deliver flexibility to the actor that compensates the most. Subsequently, the DSO develops a potential competitive advantage to the supplier by forming strategic alliances with the aggregator, because it has the ability to adapt prices in temporal and spatial means [68]. Whenever congestion is expected, the utility provider charges high prices just because energy demand is high. However, if the DSO during these peak moments requests an offer for flexibility, the aggregator shifts and schedules energy demand to unlock flexibility by coordination and information exchange of the DSO and supplier. The incurred costs of energy demand that is subsequently not purchased from the supplier means less revenue, while the aggregator becomes a supplier due to its offerings of electricity to the DSO. Interactions between these power system actors becomes unavoidable, and remains a subject for future research. The value of DR should be divided along the complete value chain to all participating actors in order to avoid unwanted value asymmetry. Despite a growing literature on business models that investigate V2G opportunities, the main focus for the business cases consider V2G charging for applications in the regulation market and not so much on the subject of congestion flexibility from the perspective of the DSO [8].

5.2. Vehicle-to-Grid business model

The widely applied business model ontology of [54]; Osterwalder, serves the purpose to characterise future performance indicators of a business' success, by dissecting all relevant elements of a business case, i.e. V2G system relations, components and participating actors in electric vehicle demand response. These elements are called building blocks in [54] where the author uses as an all encompassing business characterisation that describes a business case according to four essential and nine smaller interrelated building blocks (depicted in figure 7). These building blocks serve the purpose to measure and monitor key performance indicators beyond merely financial indicators. In the section below, each of these nine elements are described. Furthermore, the business model is financially evaluated for prosumers in paragraph 7.4, and the opportunity costs of Captial expenditure (CAPEX) for investment on distribution grid assets are considered for the DSO. Modelling of multiple future scenarios with different local grid peak demands allow to inspect whether the scheduled charging strategies allow more vehicles to be connected simultaneously.



Figure 7: Business model framework

PRODUCT/INNOVATION PERSPECTIVE

Demand response is characterised by a complex network dependent mechanism, unlocked by the growing ICT infrastructure and increased prosumer connectivity. Co-management of service dependent DR is the hart of unlocking a value-added business model. The *value proposition* element, as conferred in the business model ontology of [54] can be explained by the benefits that the DSO delivers to the target customer, i.e. the aggregator, for enabling an offer for flexibility on the USEF based flexibility market. Value transfer is defined in terms of price/kWh delivered flexibility which is paid for by the DSO. The prosumer (participator), and hence the aggregator are therefore considered as pricetakers. To put it simply, flexibility offers on the market as item of value for the DSO can be described as a regulated product offering whereas the government prohibits the DSO from altering market processes for suppliers [17]. Regulation of consumer prices prevents the DSO to develop as a natural monopoly (see appendix A), which is necessary because flexibility settlement may discriminate between location and time, inducing beneficial asymmetries for different customers [17].

The intermediary actor, i.e. the aggregator has the ability to pursue other demand response services for different actors in the power system and therefore potentially establishes competitive advantage in the V2G scenario. It operates in a competitive privatised market and, therefore, competes with other aggregators. However, when offers for flexibility are made, the aggregator is bound to fulfil its accepted offers on the day-ahead market and delivers the contracted capacity. While in the current system, energy is delivered by the supplier, that if due to acting of the aggregator fails to meet its energy portfolio can get penalised. Penalties may be induced that prioritise DR flexibility for specific grid operational tasks that require priority to prevent endangering security and reliability of electricity supply, which constituted in governmental regulations. As already has been discussed, a prioritised DR list needs to be constructed to deal with multi-objectives from different actors to design appropriate shared compensations schemes.

CUSTOMER INTERFACE

The building block of the customer interface contains three elements, i.e. *target customer*, *relationship*, and *distribution channel*, that define how and to whom the DSO delivers its value proposition under a specific type of relationship with its customers, either contractbased or one-time. The aggregator and the DSO must beforehand agree on the amount of flexibility that is required to prevent congestion, either long-term contractual agreement or on single basis. BEV prosumers are the actual provides of the flexibility and may enter into recurring long-term contracts or enter on a single basis with the aggregator. The profits of delivered flexibility ought to be distributed along the supply line. The *distribution channel* through which the prosumers are billed can be integrated in the USEF framework to utilise the existing communication infrastructure [1].

INFRASTRUCTURE MANAGEMENT

[54] describes three building blocks for the delivery and retaining prosumers in the value system configuration, i.e. *capability, partnership*, and *value configuration. Capability* refers to the repeated long-term offering of flexibility for prospective customers. This flexibility, defined as the *value configuration*, can be enabled by scheduling BEV charging demand, which is facilitated by instalment of charging infrastructure and a market framework to realise energy curtailment from aggregators and provide charging for the prosumer. The crucial actor that creates the value flows is essentially the aggregator who mediates between the prosumers and the DSO. Therefore, a strategic long-term *partnership* with the aggregator to access the target customers enhances potential value creation and realises certainty. Value configuration is subsequently dependent on the amount of prosumers that participate in V2G DR services and need to be made aware on their capabilities to provide aid in congestion while being compensated.

FINANCIAL ASPECTS

In order to enable the core capability of V2G charging electric vehicles, the government and CPO make the necessary investment in the smart- and V2G charging infrastructure. For the DSO this means that costs incurred by additional investment in distribution grid assets should surpass the cost incurred by purchase of flexibility. The DSO's opportunity costs depend on the *cost* and *revenue* elements [54]. The *revenue* element is in this situation not actual revenue, but denotes deferred costs in the procurement of flexibility against additional grid investment. The value configuration has a large dependency between the costs of the 1) local grid operational efficiency, 2) procurement of delivered flexibility (amount and when is flexibility ordered) and 3) additional grid investment. Whereas an equilibrium point should be calculated according to sensitivity analyses of the former three mentioned cost- and value flows, to investigate whether investment in V2G flexibility outweighs additional grid investment. Therefore the *revenue*, or rather opportunity cost stream is directly correlated with the amount of BEVs that deliver flexibility, and thus determine the delay in grid asset investments. On the other hand, the price of flexibility and the over- or underestimated amount of flexibility only slightly influence total value configuration [1].

5.2.1. Implications for the distribution system operator

Concluding, the business case as conveyed proceeds on an already existing but unproven communication infrastructure, market framework, and an intermediary actor that has real-time access to all BEVs. This model provides insight in prospective value configuration for the DSO, for which procurement of delivered flexibility should weigh up to additional grid reinforcement costs. As described in the model above, the value flow for the DSO is significantly dependent on the size of BEV enabled flexibility, and hence on the size of the connected BEV fleet. Therefore, to investigate this value flow, a comparative simulation between smart- and V2G charging is elicited for the amount of flexibility (capacity) that can be utilised for congestion prevention. These result are subsequently expressed in percentage of feeder load shed and maximal amount of BEVs that can be connected while retaining grid limits for different future scenarios. Other financial elements influencing value configuration, i.e. operational efficiency, CAPEX investment, can be integrated in the linear model of chapter 6 in future research.

6. MODEL DESIGN SMART CHARGING & VEHICLE-TO-GRID

The insights from the literature reviewed in the previous chapters are bundled together and used to make a proposition for a smart-charging and V2G charging model that reduces local peak loads and aids in congestion. In general, for all BEV charging sessions the charging starts immediately after connection, and usually ends long before vehicles depart from the charging point. Within the connected time window there often appears to be a period where no charging activity occurs. The charging profile scheduling problem proposed in this research simulates how, using a historic data set, charging profiles can either be moved in time or reversed (discharged) within this 'idling' period to aid grid congestion and lower associated charging costs. In order to minimise consumption, the central aggregator that is responsible for scheduling the BEVs load profiles, must receive information beforehand regarding the energy consumption and connection times in the ideal situation. However, in real life scenarios this assumption is concerned with privacy and security anxiety of consumers as is described in paragraph 4.3. To propose charging schedules used to predict behaviour of connected BEVs, this research therefore first forecasts the most probable aggregated distributions of charging demand scheduling that BEVsessions from the data set exhibit. Secondly, this research delivers a proof-of-concept linear programming model that optimises demand response by using two charging strategies aimed to minimise peak demand.

The first paragraph of this chapter introduces the design criteria for the model. Paragraph 6.2 briefly introduces the scheduling problem as a linear program approach to model smartand V2G charging strategies. Next, paragraph 6.3 describes all the components in the model and its input, before defining the model's objective functions and describing it in mathematical form. Figures 9 & 10 show a schematic overview of the models input and simulation steps. In paragraph 6.4 a formulation is provided for quantification of the charging strategies in percentage of total peak demand shed within the optimisation window and charging costs.

6.1. DESIGN CRITERIA

This subsection shortly addresses the criteria attached to provision of BEV demand response within the constraints addressed by the distribution system operator. Application of demand response in a smart grid environment has a clear distinction in the task it is used for and in what part of the power grid. The main criteria in this research for assessing the potential of BEV's in grid overload alleviation, is the *control in demand rate and storage* to minimise local grid loads and prevent local grid congestion while charging the BEVs as fast as possible.

ENERGY STORED IN BEV FLEET

To study the effects of smart charging and Vehicle-to-Grid charging strategy it is required to compute aggregated BEV fleet battery characteristics in which the following terms are considered: characterised by a vector ($SOC^{n,initial}$, $SOC^{n,departure}$, P_n^h , SOC^n), where n is the vehicle ID, $SOC^{n,initial}$ & $SOC^{n,departure}$ the arrival and departure State-Of-Charge, P_n^h required consumption, and SOC^n denotes the BEV battery capacity.

There is no data available in the set that contains information about the BEV type and its battery capacity, this parameter will henceforward be computed by making a distribution of the battery capacities from the top five and top ten most sold BEVs, the fraction of charging time related to the connection time, and capacity charged for each BEV. Furthermore, due to battery limitations, the charging-discharging power is set equal and is computed with the algorithm of [41] whom integrate, among others battery specific characteristics and charging power rate.

CRITERIA OF THE LOW-VOLTAGE DISTRIBUTION GRID IN AMSTERDAM

The low-voltage feeder and its components considered in this research have technical component boundaries, i.e. maximum allowable power load, which are significantly higher compared to other low-voltage grid assets in Amsterdam. Subsequently, other local feeders throughout Amsterdam endure usually higher loads and lower technical limits. In order to prevent peak loads in both cables and feeders and provide congestion control, the maximum load values specify the constraints for which power flows should fall within this particular range. In accordance with professionals at the DSO Liander, the case study models the load on a specific feeder in Amsterdam Nieuw-West that is capable to endure maximum loads of 115kW, in either positive or negative direction. However, the assumption is made not to use previously mentioned limit, but to vary the limit of the feeder according to the maximum feeder base load to realise a practical and more realistic case for other feeders located in Amsterdam.

BATTERY DEGRADATION

An increasing amount of studies apply research on methods to quantify cost revenue structures for BEV charging to meet travel needs and optimise V2G grid balancing. One of the implications concerning V2G balancing is increased charging cycles imposed on the battery. Battery wear is an irreversible process and a known cause of driving patterns, amount of charging cycles, and depth of discharge amongst others. [80, 71].

Results from the study [80] point to the contrary. The authors performed a quantitative analysis on real-world data to assess the battery degradation impacts for peak load shaving and frequency regulation and decouples these impacts to a scenario of uncontrolled charging and driving. They found a low correlation between accelerated degradation from provision of grid services and regular calendar ageing plus driving and charging. This only amounts to an additional average capacity loss below 1% for a simulation of 10 years. [71] developed a comprehensive battery degradation model which they subsequently validate with real-world usage cycles. Their model has been integrated with algorithms to simulate V2G charging strategies in order to minimise battery degradation. Results show battery operation, both for driving mode SOC and V2G SOC, is best kept within a fixed range to ameliorate the life cycle of the battery. An improvement over uncontrolled charging. In addition, large depth of charge-discharges within the maximum physically possible ranges of the battery are correlated with an increase in battery degradation. According to [69] a general battery SOC operation range should be within: $SOC \in [0.2, 0.9]$ to avoid early degradation and capacity loss.

6.2. MODELLING APPROACH FOR OPTIMAL SCHEDULING OF CHARGING SESSIONS

Charging decisions are made through predefined objective functions to optimally utilise the effectiveness of the flexibility for the scenarios. By building forth on the results of the charging behaviour as described in the data mining chapter, an optimisation through linear programming is carried out that schedules BEV load profiles according to the strategies mentioned earlier[1], [2]. Optimisation is based on the amount of capacity that can be delayed or provided within the constraints of the BEV owner (minimal SOC at departure), the time available to provide these services (offering flexibility), the system operator's reference load curve (max. load) and lastly, the duration or period in which flexibility can be offered [63, 17, 75]. The algorithm performs the simulation under the objective function to minimise the subtraction of the absolute differences between BEV charging demand and the feeder load while capacity limits are respected. Furthermore, this model aims to ensure to not minimise a change in the original energy consumption and charge BEVs as fast as possible. However, whenever grid constraints are exceeded, the linear model will find a solution that violates the consumer requirements as little as possible. This means if grid constraints are in conflict with BEV demand, the total consumption of BEVs might be lower than with uncontrolled charging.

6.3. MODELLING COMPONENTS - INPUT AND OBJECTIVE FUNCTIONS

The problem to solve is a discrete-time system that takes on the form of objective functions, which are expressed into linear aggregated demand functions to minimise energy peak demand and compute an optimal solution for scheduling sessions. The base of the model described in [83, 68] is adopted and rewritten to be applied for the BEV scheduling problem. The goals of scheduling the charging sessions are twofold, aimed to minimise local distribution loads and assess the financial costs of power supply for the consumers. In the optimisation window charging sessions are iteratively solved each time slot in a horizontal time receding approach. This is done by controlling each BEV's battery power and keeping the battery at a required reference SOC, i.e. $E_{t,ref}$. The reference SOC equals the maximum allowable energy for each BEV to intend the fastest charging upon arrival for consumer satisfaction. Therefore, to pose this problem as an objective function, it can be defined as a non-linear expression by minimising the absolute term of the BEV battery energy and given reference BEV energy at time step h. This problem will in a later section be rewritten into a linear programming model.

$$\underset{P_n^h}{\text{Minimise}} \quad \sum_{n=1}^{N} \sum_{h=1}^{H} |E_{n,h} - E_{n,h,ref}| \tag{1}$$

6.3.1. Model input

This section describes the expressions and constraints used to model the smart charging and V2G strategies.

TIME WINDOW

To capture the electric vehicle load sessions, the demand profiles are discretised in hourly slots $h \in H = [1, 2, ..., H]$ for an optimisation window with a length of the earliest BEV arrival to the latest BEV departure, which is defined by a start time slot t_{start} and an end time slot t_{end} . For each charging session or user (n) a column vector is created $n \in N = [1, 2, ..., N]$ that represents the charging demand vector from start-to-end of each individual session $\delta t = [t_n^d, t_n^a]$ within the optimisation window $[h_{start}, h_{end}]$:

$$P_n^h = \begin{bmatrix} P_1^h, & P_2^h, & \dots, & P_N^h \end{bmatrix}^T$$

$$\tag{2}$$

BEV ENERGY DEMAND

The model is described in such a way that the control task is assigned to the aggregator, who models aggregated BEV fleet loads according to the maximum and minimum energy and battery power boundaries of all BEVs, defined as $E_{max/min}$, $P_{max/min}$ respectively. Both the uncontrollable and controllable electric vehicle charging demands are modelled separately, for which the charging- and discharging power rates as well as the battery energy boundaries are specified by the set of charging profiles according to the following expression:

$$E_{min} < E_{n,h} < E_{max} \tag{3}$$

Constraint 3 is the energy level of the BEV, which is bounded for the minimum and maximum value. Minimum state-of-charge energy level required before it can discharge represents whether the BEV can immediately start charging or discharging throughout its interconnection. Furthermore, note that the charging and discharging efficiency is assumed to be similar for both trajectories and equal to 100%. The motivation behind this assumption is that efficiency depends on the charging rate and charging profile, which is out of the scope of this research. However, the maximum and minimum charging output for each charging point is computed by the algorithm of [41] that includes exogenous and BEV characteristic battery values and influence the theoretical charging rate.

1. Uncontrolled charging demand Whenever a BEV connects with an expected charging duration no longer than its total connection time to ensure the battery is sufficiently charged (assumed a SOC of 100%), its total connection period is used for recharging. The charging rate and energy boundaries have no flexibility. For uncontrollable charging demand, only the charging power rates are used. The uncontrolled battery charging boundary is mathematically expressed as:

$$P_{n}^{c}(h) = \begin{cases} 0 & h \le t_{n}^{a}, h > t_{n}^{d} \\ P_{n,h,max} & t_{n}^{a} < h \le t_{n}^{d} \end{cases}$$
(4)

2. Controlled charging demand An electric vehicle with an expected connection duration longer than the necessary charging time, exhibits flexibility in charging and discharging rates. Therefore as considered for smart charging points, the battery output power can be adapted dynamically, such that the battery charging boundaries can be defined for smart charging described in expressions 5 & 7, and for vehicle to grid charging 5, 6 & 8. The indices c & d denote either charging or discharging. The boundaries are given by:

$$P_{n,h}^{c} = \begin{cases} 0 & h \le t_{n}^{a}, h > t_{n}^{d} \\ P_{n}^{c} & t_{n}^{a} < h \le t_{n}^{d} \end{cases}$$
(5)

$$P_{n,h}^{d} = \begin{cases} 0 & h \le t_{n}^{a}, h > t_{n}^{d} \\ P_{n}^{d} & t_{n}^{a} < h \le t_{n}^{d} \end{cases}$$
(6)

$$0 \le P_{n,h} \le P_{n,h,max}^c \tag{7}$$

$$P_{n,h,min}^d \le P_{n,h} \le P_{n,h,max}^c \tag{8}$$

DISTRIBUTION NETWORK POWER BOUNDARIES

1. Power constraints on distribution feeder

In accordance with professionals at Liander, the considered feeder in this case study has a varying capacity limit of 100-90-80-60-50% of the maximum base load in both positive and negative direction, which can be expressed as a constraint:

$$|P_{f,h,min}| = P_{f,h,max} \le limit \tag{9}$$

Next, the network minimum and maximum power output constraints of the grid can be expressed in a relationship with the minimum and maximum battery power constraints for every electric vehicle. The grid power constraints determine whether either charging or discharging at a specific time step is possible for each BEV separately. The grid power constraints for smart charging and Vehicle-to-Grid can be computed from 10, and are expressed below in 11 & 12, where $P_{b,h}$ denotes the base load on the feeder:

$$P_{h} = \begin{cases} P_{h,min} = -P_{f,h,max} - P_{b,h} \\ P_{h,max} = P_{f,h,max} - P_{b,h} \end{cases}$$
(10)

2. Smart charging minimum and maximum power demand

The minimum power output for smart charging is limited to zero as no discharging is possible in this unilateral charging strategy. The sum of the charging rates of all BEVs should remain within the boundaries, and expressed as:

$$0 \leqslant \sum_{n=1}^{N} P_{n,h} \leqslant P_{h,max}, \quad \forall h \in H, \forall n \in N$$
(11)

3. Vehicle-to-Grid minimum and maximum power demand

$$P_{n,h,min} \leqslant \sum_{n=1}^{N} P_{n,h} \leqslant P_{h,max}, \quad \forall h \in H, \forall n \in N$$
(12)

ELECTRICITY COSTS

The costs associated with (dis)charging are calculated by integrating a RTP day-ahead pricing scheme, and henceforward uses hourly rates of the electricity price on the day-ahead market. Charging costs are, therefore modelled through use of a dynamic retail price tariff incorporating day-ahead price variation. To propose a feasible variable tariff, the hourly variation in the day-ahead price is multiplied by a coefficient such that the averaged price over the time interval shadows the conventional used charging electricity cost (during this research the price equals $0.32 \in /kWh$). This price is charged to the consumer by the UTP. The total electricity cost for the following day per hour for all BEVs is calculated by taking the product of total load consumption in a particular time slot and the RTP pricing tariff (in the same time slot):

$$c_h(P_n) = (p_{c,h} \cdot P_h^c \cdot \Delta t) - (p_{d,h} \cdot P_h^d \cdot \Delta t)$$
(13)

With $p_{c,h} \& p_{d,h}$ are the respectively charging- and discharging electricity prices for each BEV derived from the EPEX day-ahead market prices and can be seen in the appendix B. The constructed the RTP-tariff is depicted in figure 8. $P_h^c \& P_h^d$ are the charging and discharging power for each BEV specific, and Δt the length of the time interval.



Figure 8: Averaged Real-Time-Price tariff derived from EPEX-day-ahead market prices for 24 hour [18]
6.3.2. Optimisation formulation

Each charging session n is combined with the system load vector of households as described in the previous section, and subsequently included in the charging demand vector to determine the total allowable load on the system feeders. The optimisation problem is written below in standard form. However, while this problem imposes a nonlinear formulation that expresses the minimisation of the absolute energy differences, it is necessary to decompose this problem into linear constraints prior to solving [24]. After this operation it is possible to formulate the problem into linear programming. This process involves the introduction of dummy or auxiliary variables for each BEV session: $\overline{E_n}$ in the constraints. This means that the absolute value of equation 1 can be rewritten into the single variable $\overline{E_n}$ by substitution of two variables into the original constraints, i.e. $\overline{E_n} = E_{n,h,ref} + E_{n,h}$, and $\overline{E_n} = E_{n,h,ref} - E_{n,h}$.

Furthermore, the linear program used in this research attempts to approximate the constrained optimisation problem with an unconstrained problem, after which it computes the solutions. By adding a term to the objective function that attaches a large penalty value to the violation of the constraints, approximation can be accomplished. Basically, this converts the optimisation problem with inequality constraints into an unconstrained problem. This is formally defined as the barrier method that favours solutions, which are in the feasible region over those near the constraint [20]. Thus, may requirements unable to be satisfied, this method will find a solution that minimises the violation of the constraints as little as possible. In addition, whenever no feasible solution can be computed for the charging strategies, a desirable behaviour of the optimisation is to still find a solution to meet the required BEV demand and grid constraints. According to the barrier method, the variable P_{over} is introduced and given a large penalty weight: $c \gg \overline{E_h}$.

SMART CHARGING STRATEGY

Taking into account the linearisation and the barrier function, the linear demand optimisation problem can finally be written as:

$$\underset{E_{n,h},P_{n,h}}{\text{Minimise}} \quad \sum_{n=1}^{N} \sum_{h=1}^{H} \overline{E_h} + c \cdot P_{over} \tag{14}$$

Subject to:

$$\frac{\overline{E_n} + E_{n,h} \ge E_{ref}}{\overline{E_n} - E_{n,h} \le E_{ref}}$$

$$E_{min} < E_{n,h} < E_{max}$$

$$E_{n,h+1} - E_{n,h} + P_{n,h,over} \ge 0$$

$$E_{n,h+1} - E_{n,h} - P_{n,h,over} \le P_{n,h,max}$$

$$E_{n,1} = E_{n,start}$$
(15)

59

$\rm V2G$ charging strategy

The objective equation 14 is similar for the V2G strategy as conveyed above, but depends on different constraints:

$$\underset{E_{n,h},P_{n,h}}{\text{Minimise}} \quad \sum_{n=1}^{N} \sum_{h=1}^{H} \overline{E_h} + c \cdot P_{over}$$
(16)

Subject to:

$$\frac{E_n + E_{n,h} \ge E_{ref}}{\overline{E_n} - E_{n,h} \le E_{ref}}
E_{min} < E_{n,h} < E_{max}
E_{n,h+1} - E_{n,h} + P_{n,h,over} \ge P_{n,h,min}
E_{n,h+1} - E_{n,h} - P_{n,h,over} \le P_{n,h,max}
E_{n,1} = E_{n,start}$$
(17)

Model input and simulation diagram

The optimisation can be schematically depicted as in the figures below.



Figure 9: Model input

Figure 10: Diagram for the optimisation simulation

6.4. Charging strategies quantification

This study provides exploratory insight into the potential of different charging strategies towards impact on grid load impacts for different actors in the system. The linear program is simulated for different near future use cases for which the BEV penetration and feeder load parameter values will vary. To demonstrate the potential of the conveyed smart charging and V2G charging strategies, the percentage of total peak shed is computed from the results of the optimisation given by the ratio of the scheduled demand against the unscheduled demand for every hour in the optimisation window:

$$PeakShed = \left(1 - \frac{\sum_{h=1}^{H} P_{V2G,h}}{\sum_{h=1}^{H} P_{uncontrolled,h}}\right) \cdot 100\%$$
(18)

Furthermore, associated potential charging costs are computed for the strategies in price per kWh according to the electricity tariff proposed in 6.3 and subsequently compared to the electricity price in the uncontrolled scenario.

7. PREDICTING TRENDS AND PERIODICITY IN CHARGING BEHAVIOUR & SIMULATION

In this research a large-scale data set is collected that is comprised of BEV charging sessions, acquired from 1832 public charging points located throughout the area of Amsterdam. The municipality of Amsterdam recorded 682,476 usable charging sessions that span a full year, 2017-03-01 - 2018-03-01.

The focus of this research is to investigate typical trends and impacts of BEV charging behaviour demand on Amsterdam's distribution grid. In this case study, the potential of BEV flexibility is derived that can have the purpose to serve for demand response strategies to relieve the local grid from congestion and to minimise charging costs. Both smart charging and V2G charging strategies are modelled for DR and subsequently compared on their performance and benefits to the DSO and the consumer.

This chapter is structured by starting with a brief introduction in paragraph 7.1, which presents the data sets used in this study and refers to the normalisation and scaling of charging sessions into usable load profiles. Subsequently, paragraph 7.2 delivers a statistical assessment of these charging profiles to explore distinct stochastic behaviour and trends in temporal-spatial measures. Demand response related factors are computed that describe the availability and amount of flexibility at different periods throughout the year. Furthermore in paragraph 7.4, the charging strategies are simulated under the objectives to minimise both congestion (over consumption or supply) and associated charging costs by using the results from the previous assessment on predicting behavioural BEV demand trends. To realise a solvable optimisation window the proposed charging schedule optimisation is performed for a maximum period of 36h with an hourly time step resolution. Both the period can be shortened and the resolution can be adapted to finer time steps to gain solution accuracy.

7.1. DATA COLLECTION AND PREPARATION

DATASET ELECTRIC VEHICLE CHARGING SESSIONS

The BEV charging session dataset, already shortly addressed in the introduction of this chapter, is provided by the municipality of Amsterdam, in which they recorded individual charging sessions. Each uniquely recorded charging session is attributed with transaction ID, Charging point ID, and Location ID keys, and contains measured parameters as shown in table 1.

| Parameter | Explanation |
|--------------------|--|
| Charge session key | Individual key for each session |
| Start connection | Date and time a session |
| End connection | Date and time a session ended |
| Connection time | Connection time of a session in hours |
| Idle time | Time connected and not charging |
| Charging time | Charging time of a session in hours |
| kWh | Capacity charged/Power consumption |
| Charging point ID | ID code of a charging point |
| Location ID | Postal code of charging point |
| Latitude | Latitude coordinate of charging point |
| Longitude | Longitude coordinate of charging point |
| P_h^c) | Charging point demand |

Table 1: Example of keys and attributes of the recorded sessions

To provide insight in the temporal and spatial effects of charging sessions in the data set, various plots are depicted that show distributions over time for different variables on time- or interval scales. On forehand, assumptions are made which sessions to omit in the analysis denoted as outliers that negatively influence validity of the assessments:

- Sessions that have charged less than 1 kWh, because these small charge capacities usually mean a faulty recording (deleted 69,917 sessions)
- Sessions that have a longer connection time than 72h are discarded, because these excessive sessions indicate a heavy-tailed distribution and influence the distributions of the data (deleted 41,140 sessions) Furthermore, these data sessions are not representative for commuter charging behaviour as compared to the complete distribution.

To analyse the characteristics of sessions, the following variables are calculated that further describe charging behaviour, and are formally expressed as:

$$Connection time \equiv t_{departure} - t_{arrival} \tag{19}$$

$$Idle time \equiv t_{connection} - t_{charging} \tag{20}$$

DATA SET FEEDER LOAD OF HOUSEHOLD-, BEV, REG LOAD

To analyse the effects of BEV charging on the local grid, Liander provides this research with a low-voltage feeder data set containing load profiles of 40 connected households with- or without REG, and two BEV charging points. This feeder is located in district Nieuw West in Amsterdam. For further visual specification and graphs see appendix C A. The feeder load profiles, as well as the other load profiles, are scaled to hourly time intervals to match with BEV load profiles. The model simulates total load impact on this specific feeder by integration of different BEV charging strategies, and can be conceived as a proof-of-concept for these mentioned strategies.

DATA SET VEHICLE-TO-GRID

During this research two V2G charging points originating from the City-zen pilot project V2G are taken into operation from 2018-05-07 on, and are connected to the same feeder as described above. Data is collected of the total power load, both consumption and generation. Power load graphs can be found in appendix D and serve the purpose to provide insight in bilateral charging effects on feeder loads. Note however, the discharging profiles are activated when the battery SOC is above a specified value and furthermore are not optimised in any way. In contrast to the results of this data, the charging profiles which are simulated with the model of chapter 6 are optimised to the feeder load.

7.2. Analysis of trends and seasonality in Electric vehicle session characteristics in Amsterdam

By performing a statistical assessment on spatial and temporal effects of charging in Amsterdam from the sessions that are recorded, insight is gained to understand the probability distributions of factors that comprise charging demand and flexibility behaviour. Distributions of arrival-departure times and charging demand, which influence available time- and load-flexibility are analysed for typical recurring behaviour. Global measures are obtained by making use of the statistical models local regression (LOESS) and auto-correlation. These models allow statistical operations on the (univariate) time series data to capture the underlying characteristics such as trend, seasonality, periodicity, and serial correlation [10]. The results of this assessment on behaviour can be used for aggregators to make preliminary predictions about the expected BEV demand characteristics throughout the day and year. The next section describes the method that is adopted to analyse the time series variables.

TIMES SERIES METHOD - TREND AND SEASONALITY

Abstraction of the stochastic nature of charging session characteristics shows insight in the components of these time series that have recurrence in short cyclical nature, and makes it possible to characterise patterns. Usually univariate time series are abstracted or decomposed to forecast their parameters into trends, seasonality periodicity, and residuals (remainder). Trends are defined as long-term changes in the mean of the time series, whereas seasonality is the periodicity of the series over fixed intervals (e.g. month of the year). [10] proposed a model to work with finer grained time resolution steps of, for example one day to show oscillatory effects that can be analysed by the seasonal-trend decomposition method based on (non-parametric) local regression. This method assumes the time series data is an additive series that is composed of: trend (T), seasonality (S), and remainder (Y'_t) , which can be mathematically expressed as:

• Original data

$$Y_t = T_t + S_t + Y'_t \tag{21}$$

• Decomposed trend

Let X_t be the de-trended data with the following computation:

$$X_t = Y_t - T_t \tag{22}$$

• Decomposed seasonality

Let Z_t be the de-seasonalised data:

$$Z_t = Y_t - S_t \tag{23}$$

66

• Decomposed remainder

Let Y'_t be the remainder data:

$$Y_t' = Y_t - T_t - S_t \tag{24}$$

To specify the terms in the equations above, a model is fitted to the data measurements. For non-parametric data as used in this research, local regression or LOESS builds forth on classical models such as linear regression, but instead of defining a global function to fit all data points it formulates polynomial functions for 'smoothed' localised subsets of the data points. Most distributions of the variables used in this research (see 7.2) are not normally distributed, show high heteroscedasticity. Hence in cases with significant skewness and kurtosis, general predefined distributions do not fit the data well. With a LOESS approximation data points are fitted using the standard linear regression least squares method, however only applied to the nearest neighbour algorithm. Hereby allowing a straightforward way to model the seasonal component and trend.

Suppose x_i from i = n for all n are the measurements of the variable. Let g(x) be the LOESS smoothing curve, than an integer q is selected of x_i , which is the nearest to x based on the distance from x. Subsequently, the neighbourhood weight for any x_i is computed. Now the tricube weight function can be expressed by the following equation from [10]

$$W(u) = \begin{cases} (1-u^3)^3 \& & \text{for } 0 \le u \le 1\\ 0 & u \ge 1 \end{cases}$$

where u is the distance of a measurement from the point on the curve that is fitted. To decompose the time series, the number of observations per seasonal cycle has to be determined. The model is specified by daily seasonal observations for every week of the year, due to the expected daily periodicity in driving and charging behaviour during daytime. 95% of all charging sessions have mean connection times of 8.05 hour, and are therefore assumed to periodically recur in charging frequency each successive day. In addition the larger Figures 14, 16, and 21 show the result of fitting LOESS to decompose the time series. The plots in one window show the original time series at the top, followed by the seasonal component, the estimated trend (third), and the irregular remainder component at the bottom. The length of the bars at the right side express the relative magnitude of the decomposed effects on the same span in the y-axis. A larger bar means a variation attributed to one of the decomposed terms has a small influence on the data, and vice versa for small bars.

Although the LOESS model returned smoothed patterns of seasonal periodicity, it can merely be used for visual inspection, whereas to explain the characteristics of the time series, a measure of the strength of the trend, seasonal, and periodicity components is required. Furthermore, as a formal measure of strength, all component's influence in the original series can be reviewed on their contribution to the overall series variance (squared standard deviation). Equation 25 below computes seasonal influence by calculating the relative variance of the remainder with the decomposed seasonal component. Equation 26 computes trend influence by taking the relative variance of the remainder with the trend component, and lastly, the influence of the relative variance of the remainder with the original series in equation 27:

$$1 - \frac{Var(Y_t')}{Var(Z_t)} \tag{25}$$

$$1 - \frac{Var(Y_t')}{Var(X_t)} \tag{26}$$

$$1 - \frac{Var(Y_t')}{Var(Y_t)} \tag{27}$$

AUTO-CORRELATION - PERIODICITY

The statistical auto-correlation is a tool to find repeating patterns or periodicity in a time series for which successive values in time, at certain lags, correlate with each other. In other words, it computes randomness in the data for which the auto-correlation factor is (near) zero if there is no correlation and thus randomness, and (near) 1 for a perfect auto-correlation [27]. It finds, by using equation 28, the similarity, i.e. Pearson correlation, between values as a function of itself but lagged in time.

$$ACF(s,t) = \frac{E[(X_t - \mu t)(X_s - \mu s)]}{\sigma_t \sigma_s}$$
(28)

Where time 't' is discretised in daily intervals for the value X. Then the same holds for the lagged version for time step 's'. E is the expected value computed by subtracting the mean from the value at 't' or 's'. Positive correlations are strong indications of long runs of several consecutive observations above or below the 5% significance level . Negative correlations indicate low incidence of such runs. In the correlograms of the charging session variables below, the values that exceed the significance level have a dependence with successive values in time but are lagged for a certain period. Thus, the ACF values indicate how much correlation lagged values have with each other where the lag is the period for which the successive values correlate. Expected is that between adjacent days the lag the correlation is the largest.

The next sections of this paragraph provide the reader with an analysis of the stochastic distributions of the charging session variables to provide insight in the behaviour. For the most important variables, which are used in the simulation of the proposed linear program, a time series decomposition using LOESS is performed.

7.2.1. FREQUENCY OF CHARGING SESSIONS

A first overview of charging characteristics in Amsterdam is given by analysing the frequency of sessions that took place throughout the total time span of the data set. A distinction is made between districts in Amsterdam to see how non-residential aggregated demand is spatially dispersed. The districts can be seen in the left mapping of figure 11, in which the small black dots represent the locations of all charging points in the data set. First observations of this graph show relative minor deviations between frequencies for different weekdays in comparison to the location of those session. In weekend days however public charging points are less occupied and show overall decreasing session frequencies. Locations such as the city's centre and southern part (agglomeration of large business ventures) show the highest amount of sessions. The charging point occupancy, measured in number of sessions per charging point, and charging demand is highest for the city's centre.



Figure 11: Contour plot of charging sessions' frequency

| District | # of charging points | # of sessions | Charging Demand [MWh] |
|-----------|----------------------|---------------|--------------------------|
| Centre | 187 | 126,886 | 1110,39 |
| New West | 151 | $73,\!045$ | 928,060 |
| North | 117 | $34,\!403$ | 394,078 |
| East | 203 | 107,421 | 1022,418 |
| West | 187 | 96,843 | 887,502 |
| Westpoort | 2 | 728 | 7,05309 |
| Zuid | 370 | 222,033 | 1908,542 |
| Zuidoost | 58 | $21,\!117$ | 198,417 |

Table 2: Charging points and session frequency per district in Amsterdam

7.2.2. Analysis of arrival-departure times

Temporal effects of charging sessions that take place during the day are depicted for a complete year, distinguished by different seasons to show overall trends of sessions. A quick glance at figure 12 (a) and (b) in appendix C shows minor seasonal shifts in arrivaland departure time respectively, while there is a larger deviation observed between the frequency of arrival sessions for the summer. Because this dataset solely comprises recorded sessions from public charging points and hence mostly used by commuters, an explanation can be accounted to summer holidays where people don't commute to work.



Figure 12: Count of charging sessions' arrival (a) and departure (b) over time of day with seasonal differences

Arrival time

A more refined plot in figure 13 shows labelled peak and valley characteristics for the frequency distribution of arrival time during week days and weekend days. The week days and weekend days comprise 76% and 24% of the sessions respectively. In the literature charging behaviour in terms of arrival-departure times is usually categorised into residential charging, park-to-charge (or shopping), and commuter charging [63, 82]

Distinct arrival times during the day show peaks at 8:45 and 18:15 during weekdays. Making the consideration that most sessions coincide with typical workday hours in the Netherlands, these peaks can be denoted to mainly commuter charging behaviour. Charging sessions in the weekend indicate a deviant behaviour from the weekdays, where the lower plot of figure 13 shows negative kurtosis due to a larger spread in in the arrival times indicating the mass of the BEV distribution arrival times connect their BEV preferably at later periods of the day.



Figure 13: Count of charging sessions' arrival time during week days (a) and weekend days (b)

Described below is the analysis of the time series by fitting a local regression model and calculation of the auto-correlation values to assess seasonality, trends and periodicity in arrivals.

Table 3: Measure of each components influence of relative variance on total time series variance of arrival time

| | seasonal | trend | $\operatorname{remainder}$ |
|---------|-----------|-----------|----------------------------|
| Arrival | 0.3340263 | 0.4978265 | 0.1272259 |

The decomposed plot of 14 shows a varying trend denoted by two large valleys at weeks 24 and 45, which have a larger influence on the variation in the original data (0.498) compared to the daily seasonal influence (0.334). Only a small variance in the data is explained by daily seasonal effects, and the magnitude of relative variation in trend imposes the largest influence on the data. What is clear from examining the remainder component are the irregular negative outliers at week 9, 13, 42, 45 and positive outliers at week 52. Furthermore, as apparent from the figure 13 in paragraph 7.2 weekdays and weekend exhibit dispersion (with two distinct peaks) in arrival times, which can have an influence on the magnitude of the seasonal and trend component.



Figure 14: Decomposition of daily observations in weekly patterns for arrival time

DEPARTURE TIME

Similarly to the previous case, only minor variations between seasons can be inspected that influence the departure time. In figure 15 plots are depicted of departures during a whole week, in which one can notice a majority of the departures during weekdays occur at 8:15, with an additional smaller peak occurring at 17:15. A possible explanation for this distribution is the cross-correlation between the arrival-departure times of commuter drivers and work hours in the Netherlands as been said before. Figure 15 (b) indicates similar behaviour to be recognised for weekend days being positively skewed and having a large variance in time.



Figure 15: Count of charging sessions' departure time during week days (a) and weekend days (b)

Table 4: Measure of each components influence of relative variance on total time series variance of departure time

| | seasonal | trend | $\operatorname{remainder}$ |
|-----------|-----------|-----------|----------------------------|
| Departure | 0.4239381 | 0.4304912 | 0.1100280 |

Decomposed time series curves and auto-correlation results are have similar shapes as the arrival decomposed time series, as can bee noticed from figure 16. A logical explanation resides in the fact that 95% of all charging sessions have mean connection times of 8.05 hour, and are therefore periodically recurring in charging frequency each day. In addition the larger seasonality factor - 0.424 - compared to the factor of the arrival -0.334 - can be induced from the distribution of frequencies in figure 15 in paragraph 7.2 that show larger distinct peaks at 08:15, thus imposing a stronger periodicity. This premise is in accordance with the larger auto-correlation factors in figure 18 (b) compared to (a).



Figure 16: Decomposition of daily observations in weekly patterns for departure time

7.2.3. Analysis of interconnection session characteristics

Consumer tendencies for charging are statistically analysed in terms of arrival time and interconnection times. In order to estimate apparent patterns or trends, frequency and probability plots are modelled and shown in appendix C together with a decomposition of the time series and auto-correlation. Periodicity in connection time are of great importance to the aggregator, because further inference in predictions are more accurate when BEV demand together with the flexibility potential is required to be known. The sessions used for analysis exhibit for the majority longer interconnection times than what theoretically is required. Figure 33 in appendix C exemplifies this trend by calculating the ratio of charging time (according to capacity charged) relative the total connection time for each session. The lower the ratio, the more idle time within sessions. As an example, 65% of the charging sessions have a ratio score below 50%. The ratio score of 50% means that connection time is twice the required charging time. This trend enables the possibility to utilise the fact that BEVs, while connected to the grid, can provide flexibility in scheduling their charging. The section below continues the analysis of the trends and periodicity in connection times for over a year.

CONNECTION TIME

Based on the sessions' connection time distributions a minority of the sessions (<7%) have connection times that exceed 24h, see summarised values in table 5. BEVs with longer connection times than 24h have multi-model distributions showing distinct peaks at subsequent 24 hour periods, sessions are sub-clustered according to connection time of 0-24, 24-48, 48-72h to analyse weekdays and seasonal influences. From figure 35 (a) in appendix C - representing more than 77% of sessions in the data set - it can be concluded that during weekdays, the connection time is usually 1.5h shorter than in weekend days. For the remaining clusters (48h and 72h) mean connection times differ 44 min and 37 min respectively for weekdays and weekend days, but offer significantly longer connection times between the clusters.

| | Sub alustors | Fraction of clusters | Mean | Mean | Mean |
|---------|--------------|----------------------|-----------------|-----------------------|----------------|
| | Sub-clusters | | connection time | idle time | charging time |
| | 0-24h | 77.45% | $8h\ 2\ min$ | $6h\ 11\ min$ | $1h\ 50\ min$ |
| Weekday | 24-48h | 3.406% | $36h\ 22\ min$ | 34 h $13~{\rm min}$ | $2h \ 9 \ min$ |
| | 48-72h | 1.176% | $61h\ 3\ min$ | 58h 59 min | 2h~6~min |
| | 0-24h | 18.00% | 9h 32 min | 7h 32 min | 2h |
| Weekend | 24-48h | 1.964% | $35h\ 38\ min$ | $33h\ 31\ min$ | $2h 7 \min$ |
| | 48-72h | 0.4049% | $60h\ 26\ min$ | $58\mathrm{h}$ 11 min | 2h 14 min |

Table 5: Summary of average and standard deviation of connection-, idle-, and charging time clusters

In general daily seasonal variations have significant effects on the averaged connection times during the day, while between weekdays and weekend days more variation in the distributions is visible. Although, the former decomposed time series show a relative low seasonality variance effect in the original series, the connection time series is emphasised by a daily seasonality cycle with a coefficient of 0.632. The trend component however, explains only 0.117 of the relative series variance, and shows no long term increase nor decrease. The large standard deviation is a sign of the peaks and valleys in the trend curve.

Table 6: Measure of each components influence of relative variance on total time series variance of connection time



Figure 17: Decomposition of daily observations in weekly patterns for connection time

IDLE TIME

The idle time construct is the time interval between completion of charging and departure time. It is essentially inefficient to exhibit long idle times if one takes infrastructure use into consideration. If for example, the future ratio of charging points per BEV is taken into account, a low to null idle time is desired to optimally use the charging point for BEVs in its proximity. However, most research, similar to this study, found approaches to make effective use of this idle time. In the sections below, the idle time - one of the four measures in demand response - is used to calculate potential flexibility What can be noticed from table 5 is that while average charging time for the clusters only differ at most 20 min, the difference between idle time for the subsequent clusters are on average approximately one complete day (24h).

PERIODICITY OF TIME SERIES TEMPORAL RELATED VARIABLES

Now that all temporal session variables are presented, the periodicity can be calculated. In figure 18 (a) & (b), auto-correlation values for arrival and departure are almost identical. The major peaks at ACF(1) = 0.6, ACF(2) = 0.5, ACF(3) = 0.4 apart correspond to a dependence between frequency of BEV arrivals each day, 2 days and 3 days respectively. A similar statement can be elicited for the departures, that show even a slight stronger auto-correlation. In the night, the values fall back within the significance bars, due to low charging activity. The oscillatory pattern can be explained by the fact that recurring arrival times each day are dispersed across time and show strong seasonality effects. ACF for departure time is even more correlated in terms of increased ACF values. In short, the arrival and departure time series are not random and show strong periodicity. Plots (c) & (d) appear also similar (what can be noticed from the distributions in figures 34 & 36) and show also high auto-correlation values for every adjacent day up to three days ahead. In accordance with the time series analysis of the connection time. When the lag-axis is proportioned to represent weeks, a similar phenomenon is observed in periodicity. Negative values can be a sign of anti-correlation of connectionand idle-time during the night which means the observations are scatted and dispersed without any pattern. In general if an observation has an above average value, is it likely that the next observation has a below average value.



Figure 18: Auto-correlation function of (a) arrival time, (b) departure time, (c) connection time, (d) idle time

7.2.4. Analysis of charging demand

For the calculation of charging demand at the points, the duration a BEV charges is affected by multiple variables that in the article of [41] are distinguished into environmental effects, charge points effects, and BEV characteristic effects. The authors have proposed an algorithm that incorporates these variables in a computation to determine actual charging time and demand of all charging sessions. This research adopts this charging algorithm for an accurate calculation of charging demand for the sessions in the data set. Furthermore, in this research a day is discretised into hourly intervals, as this sample time stamp is a general parameter used in the electricity sector. Each charging session demand profile is scaled into 1 hour time slots from [$t_{start}, t_{chargingcomplete}$] and added to sessions that fall within the same charging interval. These scaled sessions represent the actual charging point's output demand where BEVs start immediately with charging when they arrive. Afterwards, the total demand in every time slot is averaged for every hour of the day to inspect for weekday and seasonal variation, depicted in figure 20 (a) and (b) of appendix C.

On average, the mean capacity charged (consumption) in this data set is 10.17kWh and the measure of the variation in the capacity charged, expressed as the standard deviation equals 10.57kWh. The distribution is positively skewed for which approximately 75% of the charging sessions charge less than or equal to 10kWh. Reason for this can be that a large proportion of the data set contains sessions of PHEVs, that have small battery capacities which are driven to their lowest SOC, without imposing any range limitations for the consumer as opposed to BEVs. Aggregated daily charging demand throughout the year is plotted in boxplot 19 below. The smoothing curves is a regressing fitting equation that estimates the mean charging demand conditional on the number of measurements per month. The 25% and 75% quantiles increase during the winter months, indicating a larger spread in the capacities charged, while charging demand increases from August until January.



Figure 19: (a) Boxplot of daily charging demand for monthly effects

An explanation for the shape of the demand profile throughout the day is the direct correlation it has with the frequency of BEV arrivals and departures of figure 12 (a) and (b). In figure 20 (a) as suspected, the increase in recorded sessions relates to an increase in charging demand, which is mostly noticeable in the fall and winter. In the study from [85] the authors assessed the dependencies between temperature effects on BEV efficiency, range and charging demand in which they draw the conclusion that cold temperatures and deteriorated weather conditions increase the use of auxiliary equipment. As a consequence more battery capacity is used next to driving that subsequently increases charging demand. The analysis of the time series decomposition also shows this increasing trend. Further inspection of 20 shows a large peak demand around 19:00h that coincides with the peak of BEV arrivals in figure 13, while the departure frequency, figure 15, merely shows a peak that is roughly half of the arrival frequency. Finally, the peaks visible for different days of the week also follow the behaviour of arrival and departure times.



Figure 20: (a) Mean charging demand throughout the day for seasonal variation, (b) mean charging demand throughout the day for daily variation

Table 7: Measure of each components influence of relative variance on total time series variance of charging demand

| | seasonal | trend | $\operatorname{remainder}$ |
|-----------------|-----------|-----------|----------------------------|
| Charging Demand | 0.1567597 | 0.6777841 | 0.1507764 |

The time series of daily mean charging demand in the plot below is emphasised by the large relative variation of the trend (0.678) in the original series that increases during the winter period, for which a probable explanation is the temperature effect. When temperatures decrease battery efficiency degrades and use of auxiliary equipment increases, such as heating, subsequently a larger demand for energy. The variation attributed to seasonal- and remainder decomposition are much smaller compared to the trend, and thus imposes minor effects on the original series. The charging demand therefore shows only minor seasonality effects.



Figure 21: Decomposition of daily observations in weekly patterns for mean charging demand

7.2.5. Analysis of time- & load- flexibility

To properly elicit the difference among charging strategies' potential to aid costs and grid overload, this research distinguishes two measures of flexibility for BEV demand response, i.e. time- and load- flexibility. As mentioned in paragraph 4.1 the usability of flexibility for charging strategies through smart charging or V2G is expressed by three measures [62]:

- 1. Total amount of capacity that is connected to the grid that can be discharged within the time window constraints of the BEV owner,
- 2. The moment in time (during the day) to provide these services (offering flexibility),
- 3. The duration or period in which flexibility can be offered.

Once the potential flexibility is quantified into former three measures, it can be simulated by the demand response strategies. The assessment of these measures is composed of two parts, whereas the first part, described below, investigates the time- and load- flexibility potential in the BEV fleet. Whereas the second part can be found in paragraph 7.4, simulates the charging schedules under the smart charging and V2G strategies. First insight is provided in table 8, in which a summary is shown about characteristic values of averaged individual time- and load-flexibility distributions for each BEV.

| | Sub-clusters | Median | IQR | Median | IQR |
|---------|--------------|------------------|------------------|-----------------|----------------|
| | Sub-clusters | time-flex $[\%]$ | time-flex $[\%]$ | load-flex [kWh] | load-flex[kWh] |
| | 0-24h | 0.77 | 0.34 | 14.41 | 20.00 |
| Weekday | 24-48h | 0.95 | 0.039 | 69.90 | 23.33 |
| | 48-72h | 0.97 | 0.020 | 114.1 | 17.77 |
| Weekend | 0-24h | 0.75 | 0.39 | 14.50 | 24.29 |
| | 24-48h | 0.95 | 0.038 | 68.35 | 23.74 |
| | 48-72h | 0.97 | 0.02188628 | 113.6 | 23.54 |

Table 8: Summary central tendency of time- & load-flexibility for sub-cluster distributions and 50% IQR values

The simplified trajectory of baseline, uncontrolled charging is depicted in figure 22 along with the possibilities to provide demand response within the constraints of the battery limits and connection time.



Figure 22: Charging profiles for the baseline and either smart charging and V2G

TIME FLEXIBILITY

The time flexibility depends on the charging time and total connection time of each session. The mean flexibility is calculated for each individual BEV in the data set, plotted to depict daily and seasonal/monthly variations. The ratio is calculated with equation (29) that is expressed in terms of the percentage of available idle time given the complete connection duration; availability of time flexibility:

$$T_{Flex} = \frac{(t_{connection} - t_{charging})}{t_{connection}}$$
(29)

The interpretation of the percentages in the equation is as follows. When T_{Flex} approaches 1, it means charging time is near zero in relation to connection time, and time flexibility can be used completely over the connection period of the BEV. On average each BEV charging session in the dataset can be scheduled for over 75% of its connection time.

From the box plot in figure 23, it can be concluded that the time-flexibility distribution for each succeeding month has a negative skew and the interquartile 25% and 75%ranges decrease for time-flexibility. Further investigation behind the reason for this behaviour requires assessment on exogenous related factors. Factors relating to the degree of variation in time flexibility can be weather related, or infrastructure related, such as occupancy of charging points, charging point output, to name a few. The study of [85] examined impacts on BEV energy efficiency and demand for a case study in the U.S.A. on temperature variation. They found a relationship between temporal ambient temperature variation and the range of a BEV. Lower temperatures decreases battery efficiency and increases the use of energy demand for auxiliary equipment, resulting in an increased power demand. However, further analysis of variables that influence time-flexibility are out of scope of this study. Logically, if charging demand increases so does charging time and as a consequence the median of time-flexibility decreases slightly (see figure 37). An explanation of the smaller variations in time-flexibility than those of charging demand can be found in the fact that an increase in charging demand only imposes a relatively smaller part to the idle time part in total connection time.



Figure 23: Boxplot of time-flexibility fraction of charging sessions for during every season. [Note: bold black line = median; box = 25-75%; black whiskers inside 99% of the data = minimum and maximum, excluding outliers

Monthly differences are relatively small compared to the differences throughout the day, and from figure 23 a small increase in time flexibility can be observed during the summer months.Furthermore, on average, every day, each BEV has 61.7% of its total connection duration available to reschedule its charging demand. However, the box plot doesn't show when the time-flexibility metric occurs throughout periods of the day, while start-and end- times of the sessions are key in the determination of the benefits for provision of demand response. Therefore, the usability of time flexibility at various times of the day is plotted for seasonal and weekday variations, which is visible in figure 24.

The plot in figure 24 shows the mean of the time-flexibility and its availability at every hour of the day for aggregated charging sessions. Considering charging loads are introduced immediately after a charging sessions starts with uncontrolled charging, the distribution in figures 13 & 15 depict distinct connection times for a typical workday at non-residential charging locations. It suggests a larger fraction of consumers connect their vehicle when they arrive at work and disconnect when they commute back home. Availability of time flexibility (figure 24 (b)) is consequently rising from 13:00-18:00 and, apart from small dips around 19:00-20:00, remains high until 22:00. The odd behaviour of the negative slope from 22:00 - 05:00 can be explained by a decreasing numbers of arrivals and increasing number of departures, which is an indication of decreasing flexibility. Further inspection of seasonality impacts on flexibility backs the earlier mentioned assumptions about the fact that during the winter period charging demand is higher and hence flexibility is lower.



Figure 24: (a) Mean time flexibility per hour of charging sessions throughout the day for seasonal variation, (b) mean time flexibility per hour of charging sessions throughout the day for weekday variation

LOAD FLEXIBILITY

The measure for load flexibility is applicable for BEVs that provide demand response under both charging strategies. A distinction is made between provision of load-flexibility for over generation or over consumption in the grid, denoted as $P_{Flex,up}$ and $P_{Flex,down}$ respectively. These measures express the capacity that could be charged or discharged respectively within the connection time window for aggregated sessions. Equation 30 for $P_{Flex,up}$ is a measure for the total rate of energy demand at any time instant that can be decreased within the interconnection period of a charging session in response to low electricity prices (applicable for smart charging). For simplicity and to provide clarification in the amount of energy demand that can be discharged during the total connection time of all BEVs to reach the required consumption, the equation computes to lowest possible charging demand rate, with E_n the original energy consumption:

$$P_{Flex,up}^{n}(t) = \frac{E_n}{\Delta t_{connection time,n}}$$
(30)

 $E_{Flex,down}$ is the capacity that can be offered to the DSO by V2G discharging within the window of all BEV charging sessions for each hour of a day, and is therefore not expressed as power and aggregated. Simply defined, this flexibility can be distinguished into three parts: 1) the arrival- or initial capacity of all BEVs summed, 2) the total connected capacity of all BEVs summed from the moment they are fully charged, and 3) the discharging rate of all BEVs summed for every hour of a day. Distribution plots are made for these three measures that can be used to inspect the flexibility on average for every day of the week. These measures provide expected load-flexibility available on the distribution grid for aggregator DR purposes.



Figure 25: (a) Aggregated total battery capacity per hour of the day for weekday variation, (b) aggregated initial (arrival) battery capacity per hour of the day for weekday variation

The graphs in figure 25 clearly show a large deviation in total connected capacity throughout the day, explained by the characteristics of the arrival and departure frequencies, which were addressed in figures 13 & 15. At around 08:15h the departure rate is 21,000 that levels towards 7,500 at 10:00h. Whereas the arrival rate at 08:45h is only 11,000that levels towards a rate of also 7,500 between 10:00-16:00h before increasing to 17,000 at 18:15. The curve of the load-flexibility graph at weekdays therefore denotes a sharp decrease in total connected battery capacity of $2.3 \cdot 10^6 kWh$ to 900,000 around 07:00h and increases again around 17:00 to remain high untill 07:00h. A confirmation of the dependency of load-flexibility behaviour and the arrival-departure frequencies may be explained by the average connection time, i.e. 10h 23 min, that is approximately the length of the valley in the figure. Furthermore, on Fridays the load-flexibility curve decreases with a much lower rate compared to other weekdays. Either a lower departure rate of higher arrival rate after 08:00 explains the curve characteristic. Weekend days also show a deviant behaviour in load-flexibility, for which the sharp decrease of the curve from 06:00-08:00 is characterised by almost zero arrivals and a large peak of 5,000 departures. Afterwards arrivals slowly increase such as for the load-flexibility.

Furthermore, graph (b) depicts the initially connected capacity determined upon arrival of BEVs, and is readily available for the aggregator to provide demand response with. Initial connected capacity is calculated by using the available idle-time of each BEV and its specific charging demand, and thus represents the load-flexibility that can be discharged within that period. The shift of graph (a) is equal to the charging time due to the fact that total battery capacity is available only after the BEVs are fully charged. The daily trend is similar to that of the arrival- and departure peaks, which explains the sharp decline in load-flexibility that starts around 06:00 and increases again around 17:00. The decline in the morning is due to an almost double amount of BEVs that depart compared to BEVs that arrive.

Figure 26 is a derivative of the curves in figure 25, which exhibits the same characteristic curve but defined by the discharging rate. Figure 26 (a) and (b) show the load-flexibility discharging as calculated from the charging rate of the charging points originally in the data set (mean of 5.2kW), and with a discharging output of 10kW for future V2G charging points respectively. Due to the fact BEVs have the possibility to immediately discharge upon arrival makes them useful for flexibility services during peak demand periods throughout the day, i.e. from 06:00 to 18:30 load-flexibility discharging demand is increasing.



Figure 26: (a) Aggregated discharging rate for regular charging points per hour of the day for weekday variation, (b) aggregated discharging rate for V2G charging points per hour of the day for weekday variation

In addition, the mean load-flexibility is also computed for each individual vehicle to show insight in yearly behaviour, which is presented in the figure below.



Figure 27: Boxplot of load-flexibility for each individual charging session during every month. [Note: bold black line = median; box = 25-75%; black whiskers = minimum and maximum, excluding outliers;outlier dots = outside 99% of the data]

On average each BEV present in the data set exhibits the potential to discharge 21.03kWh during the week, and 24.68kWh in the weekend during its connection period. Most of the monthly distributions have a positive skew and a slight increase during the winter months. The reason behind this behaviour requires insight in factors that influence load-flexibility. Although charging demand increases during winter months, the load-flexibility shows minor averaged variation during the year, because a large proportion of the data sessions in the set (approximately 75%) charge 10kWh or less while showing recurring behaviour. During fall and winter charging sessions aggregated load-flexibility per hour of the day is higher compared to the summer/spring in figure below. An logical explanation follows from the higher amount of connected vehicles. While charging sessions on average exhibit high time-flexibility percentages, more connected vehicles at the same time instant exhibit coherently also more load-flexibility.

PERIODICITY OF TIME SERIES DEMAND RELATED VARIABLES

The periodicity of the demand related factors provides insight in the recurring behaviour of the impact that loads induce on the distribution grid. Auto-correlation of charging demand in figure 28 is throughout the entire time series strongly auto-correlated, meaning the auto-correlation at any specified lag value is not random. The overal positive lag indicates a high association up to 4 days apart. Especially for the next day (lag = 1) correlations are the largest. Time-flexibility and load-flexibility are derivatives from connection- and idle-time, and therefore, again show similar auto-correlations. While load-flexibility shows more randomness during night time, during day time the flexibility measures can be reasonably accurately estimated for the moment in time they occur.



Figure 28: Auto-correlation function of (a) charging demand, (b) time-flexibility, (c) load-flexibility

7.3. Charging behaviour implications for the distribution grid

In summary, the examined urban-level effects of BEV charging with empirical data elicits trends and temporal information of charging session characteristics, such as interconnections, energy demand, and potential flexibility. Paragraph 7.2 provides valuable and up-to-date insight for typical BEV charging demand in Amsterdam. It also quantifies the potential flexibility BEVs can offer at certain hours throughout the day. Whether that is for provision of $P_{Flex,up}(t)$ during low peak demand hours, or $P_{Flex,down}$ during high demand peak hours. It can be concluded that a significant part of the charging sessions show commuter driving behaviour in former mentioned characteristics, for which the interconnection times follow workday related times usually starting at 08:00 and ending at 18:00. Other behaviour can be denoted as park-to-charge, in which BEV drivers connect their BEV not excessively longer than necessary. As a result, time- and load-flexibility is available during standard Dutch work hours. Especially during peak charging demand hours around 07:00-09:00h and 17:00-19:00h time-flexibility for individual BEVs shows significant peaks of 63% and 74% respectively. On average every day, each BEV has 61.7% of its total connection duration available to reschedule 10.17kWh for the smart charging strategy. This equals the regular energy demand as exhibited in the data set but can be rescheduled over the BEVs complete connection time. In terms of load-flexibility on average each BEV present in the data set exhibits the potential to discharge 21.03kWh during the week, and 24.68kWh in the weekend within the individual connection periods. Because the load-flexibility represents the total- and initial- connected battery capacity every connected hour of each BEV, the relation with the time-flexibility has to be drawn carefully. The valleys of time-flexibility at 05:00h and 10:00h mean that the connected vehicles exhibit relatively low idle times but impose a connected battery capacity on the grid. These vehicles contribute therefore only a limited amount of capacity to discharge into the grid. In short, it is demonstrated that BEV charging sessions in Amsterdam on average exhibit sufficient time-and load-flexibility values that can be exploited for effective demand response in districts where charging hot spots occur (Amsterdam south/centre).

The inference of trends and periodicity in the arrival, departure, connection time, and energy demand of the charging sessions in the data set, allow to predict these variables for a few days ahead with reasonable accuracy. The available time- & load-flexibility shows a high auto-correlation coefficient for up to three days ahead for temporal related charging variables and up to four days ahead for energy demand related variables. Subsequently, the aggregator or the DSO are able to predict the periodic recurring behaviour of those former mentioned variables for a few days ahead. This allows for more effective daily operational grid tasks as charging demand can be optimised by taking prospective BEV demand into account.

The next section continues the analysis of charging behaviour impacts on the distribution grid by modelling a case study that simulates the proposed charging strategies. The mean, variance and standard deviation parameters of the sessions analysed in the current section are used to sample sessions that approximate these parameters of all the sessions, and are subsequently used as input for the simulation.

7.4. Case study - Simulation of electric vehicle demand response strategies

The experimental case study in this research proposes mathematical modelling of historical recorded charging sessions at a specific part of a low-voltage distribution grid in Amsterdam Nieuw-West to optimally schedule charging profiles for smart charging and V2G charging strategies. Objective functions as intended to minimise congestion rely on a prediction of BEV energy demand, which is performed in the previous paragraph. Although in reality actual charging characteristics are unknown at the time of scheduling, historic sessions may give an accurate estimation of the demand that is expected, and the capabilities for application of charging strategies

The charging sessions used as input in the simulation are sampled sessions from the complete data set, which are representative for the energy demand that is expected at the specific feeder location. Because the values of the parameters used as input in the optimisation model, i.e. arrival-departure time, charging demand, initial state-of-charge, battery capacity, are assumed to be dependent on each other and thus can be modelled as a multivariate distribution [55, 11]. First, the parameter values are discretised into hourly time intervals for 24 hour, in which the multivariate distribution expresses the covariance of each value's occurrence at a specific time stamp. The correct number of given vehicles for each day, depending on the scenario, are randomly sampled from the multivariate distribution according to the standard deviation, mean and co-variance between the variables of the original data for each session. The charging sessions are then stored as a vector containing: $P_{consumption}^n$, $t_{connection}^n$, E_{SOC}^n , $(t_{arrival}^n, t_{departure}^n, L^{n,c,d})$. The parameter values of the samples sessions are shown in table 9 and figure 29. This sampling method realises samples that approximate the whole data set for the standard deviation, mean, and co-variance measures. The reason behind this method is to sample session characteristics according to their probability occurrence as input in the model, in

order to make this proof-of-concept reproducible for a larger-scale impact study in the local distribution grid of Amsterdam.

| Sampled BEVs | Mean required consumption [kWh] | Total battery capacity [kWh] | Mean connection time | Mean arrival time [hh:mm] | Mean departure time [hh:mm] |
|--------------|---------------------------------------|------------------------------------|----------------------|------------------------------|--------------------------------|
| 5 | 12.7 | 118 | 8h 21min | 9:48 | 18:09 |
| 10 | 10.8 | 178 | 8h 59min | 12:12 | 21:11 |
| 15 | 12.6 | 282 | 8h~52min | 09:36 | 18:28 |
| 20 | 14.1 | 425 | 8h~47min | 12:21 | 21:07 |
| 25 | 13.2 | 482 | 9h 9min | 11:55 | 21:04 |
| 30 | 13.2 | 601 | 9h | 14:30 | 23:30 |
| 34 | 13.8 | 680 | 8h~51min | 13:57 | 22:47 |

Table 9: Parameter values of the sampled sessions, which are the input of the LP model



Figure 29: Charging demand of sampled sessions

Due to the fact this specific local grid is robust and developed to handle near future electrification increase, the simulations for different BEV penetration levels are performed for varying feeder capacity limits. In addition, the actual feeder base load is multiplied
by a factor of five to be more representative and realistic for other feeders located in Amsterdam. To provide more quantitative insight in the optimisation results for different scenarios, the BEV penetration values will range from the short-term scenario of five BEVs (10% market share), to the long-term scenario of 34 BEVs (75% market share) by steps of five. The linear program is subsequently performed to optimise charging demand of these BEVs for situations in which the feeder's maximum limit is varied at 50%,60%, 70%,80%, 90%, and 100% of its maximum base load, which is 144kW and depicted in figure 30. In this experimental setup of the distribution grid the maximum feeder base load, for all optimisation scenarios, is multiplied with five. This operation is performed to induce optimisation results that are in line with loads on other local grid assets in Amsterdam. These values are determined in collaboration with Liander to mimic actual feeder load values as can be apparent in other parts of the local distribution grid in Amsterdam. Furthermore, demand of uncontrolled charging profiles depends on the charging point equipment for which the rate of each specific point is used from the data set. The smart- & V2G charging points have a rated output of 10kW. The theoretical total maximum aggregated demand of 5 BEVs is 50kW and 34 BEVs is 340kW, and so on. Charging costs are modelled according to the RTP day-ahead tariff and expressed in price per energy (\in/kWh) . Furthermore, it is assumed that the base load in each receding interval of the day is known.



Figure 30: Feeder limit constraints according to base load of 144kW

To solve the LP a non-commercial linear program solver of R is used, i.e. LPsolve. The use of this particular solver is restricted by the privacy disclaimer concerning data duplication prohibition exerted by the data set provider. Results are depicted in a plot of total feeder power load through time for each of the charging modes. According to the total allowable feeder load, the simulated charging strategies show at which penetration rate of BEVs in the local grid, these boundaries will suffice. Benefits for either charging modes are listed for both the DSO (as total percentage of consumption shed) and consumer (as price per kWh).

7.4.1. Analysis of effects on grid congestion

In this section the linear program is simulated on the feeder load with the sampled electric vehicle charging sessions in Amsterdam Nieuw-West as an example. The loads that are optimised in the objective function of 16 depend on the prediction of both household- and BEV power demand, and the varied values of the feeder limit. Although, the power prediction of the household loads are beyond the scope of this research, the prediction of BEV consumption is obtained by performing a time-series local regression analysis and auto-correlation using historical data. Furthermore, the optimisation results of the charging sessions depend significantly on the maximum allowable feeder load. One should bear in mind that all feasible solutions of the LP will be similar to the uncontrolled charging scenario if no grid constraints will be violated, because the LP model always charges each BEV as fast as possible. Also, whenever the induced loads on the feeder exceed the maximum limit, the feeder will encounter a failure and immediately shuts down. Hence, in every simulated case congestion is apparent and the feeder limit is violated if the uncontrolled BEV charging demand is added on top of the base load. Also, the smart charging strategy is not applicable for a feeder limit other than 100%, thus this strategy will merely be used to verify the V2G charging strategy against that smart charging strategy in the simulation with a feeder limit of 100% the base load. The results from this comparison yield similar parameters for both the charging strategies. Furthermore, a simplistic partial validity experiment is set up to test whether the model can replicate reality. This test is emplyed by altering the parameter values such that the real-world values of the uncontrolled charging scenario are modelled. The results show that an optimal solution is found, which entails that the results are an exact replication of the reality, i.e. for uncontrolled charging.

The results of the optimisation runs are presented in the tables below. For each different case, the percentage of peak load shed as well as the average charging costs per kWh are presented, and together provide an overview of possible feeder congestion situations in Amsterdam. Additionally, two tables are provided that describe the possibilities to charge additional BEVs under the V2G optimisation cases and show the energy charged per BEV in comparison to uncontrolled charging. Only the graphs of the long-term scenario are depicted that represent the worst-case as a proof-of-concept visualisation.

PEAK DEMAND SHED

Tables 10, 11, 12, 13, 14, and 15 below present the results of the simulation runs for the V2G charging strategy. The columns in the table describe the maximum feeder load, the percentage of average- and maximum peak shed compared to the uncontrolled case, the required energy demand and the energy charged in the V2G strategy. In none of the 50% and 60% feeder limit simulations the V2G strategy was able to decrease the total energy demand below the required feeder constraints, and are therefore not applicable to provide sufficient flexibility to prevent congestion. These results, when applicable, are not presented in the tables under maximum peak shed, but highlighted red. In those simulations, the limit was unable to be sufficed due to either a discharging rate limited by the charging point output, too little battery capacity to discharge, or simply no BEV connected during that period. In addition, the simulations for 5,10,15, and 20 BEVs charging with a feeder limit of 70% show results that are insufficient to meet congestion requirements as well, those are highlighted in red. In all other simulated cases the optimal solution decreased total energy demand to remain within feeder limits, and can be seen in tables: 12, 13, 14, 15. The solutions for which the charged energy in the BEVs is decreased below 90% of the required energy to maintain within feeder boundaries are highlighted in yellow. The results highlighted in green show the solutions that charge more than 90% of the energy as required or with a similar consumption, while not violating the feeder constraint. Furthermore, the maximum peak shed, expressed in percentage and reduced energy demand (kW), is calculated from subtracting the maximum feeder load in the congestion situation (uncontrolled charging) from the V2G load and the base load. The resulting maximum peak shed is an indication for the potential of the V2G strategy at peak demand periods, while remaining within the imposed constraints of the feeder.

| BEVs | V2G peak shed | BEV energy | V2G charged |
|----------|---------------|--------------|--------------|
| charging | [%] | demand [kWh] | energy [kWh] |
| 5 | 9.97 | 63 | 41 |
| 10 | 8.25 | 108 | 63 |
| 15 | 14.2 | 189 | 81 |
| 20 | 15.2 | 282 | 82 |
| 25 | 18.2 | 332 | 142 |
| 30 | 20.3 | 398 | 39 |
| 34 | 24.8 | 449 | 33 |

Table 10: Optimisation results for 50% (72kW) feeder limit of the maximum base load

| BEVs | V2G peak shed | BEV energy | V2G charged |
|----------|---------------|--------------|--------------|
| charging | [%] | demand [kWh] | energy [kWh] |
| 5 | 9.97 | 63 | 54 |
| 10 | 8.25 | 108 | 85 |
| 15 | 14.2 | 189 | 97 |
| 20 | 15.2 | 282 | 175 |
| 25 | 18.2 | 332 | 113 |
| 30 | 20.3 | 398 | 171 |
| 34 | 22.4 | 449 | 97 |

Table 11: Optimisation results for 60% (87kW) feeder limit of the maximum base load

Table 12: Optimisation results for 70% (101kW) feeder limit of the maximum base load

| BEVs charging | Maximum feeder load | Mean V2G peak shed [%] | Max. V2G peak shed [%]/[kWh] | BEV energy demand [kWh] | V2G charged energy [kWh] |
|------------------|------------------------|------------------------------|------------------------------------|----------------------------|-----------------------------|
| 5 | 155 | 3.2 | - | 63 | 63 |
| 10 | 144 | 5 | - | 108 | 90 |
| 15 | 149 | 9.4 | - | 189 | 141 |
| 20 | 154 | 12.9 | - | 282 | 203 |
| 25 | 164 | 16.2 | 38.4/33 | 332 | 161 |
| 30 | 188 | 16.6 | 46.3/87 | 398 | 234 |
| 34 | 203 | 21.9 | 50.2/102 | 449 | 201 |

Table 13: Optimisation results for 80% (116kW) feeder limit of the maximum base load

| BEVs charging | Maximum feeder load | Mean V2G peak shed [%] | Max. V2G peak shed [%]/[kWh] | BEV energy demand [kWh] | V2G charged energy [kWh] |
|------------------|------------------------|------------------------------|------------------------------------|----------------------------|-----------------------------|
| 5 | 155 | 2.1 | - | 63 | 63 |
| 10 | 144 | 4.4 | 19.4/28 | 108 | 104 |
| 15 | 149 | 7.1 | 22.1/33 | 189 | 151 |
| 20 | 154 | 7.9 | 24.7/38 | 282 | 234 |
| 25 | 164 | 12.7 | 29.3/48 | 332 | 229 |
| 30 | 188 | 13.5 | 38.3/72 | 398 | 307 |
| 34 | 203 | 18.1 | 42.9/87 | 449 | 299 |

| BEVs charging | Maximum feeder load | Mean V2G peak shed [%] | Max. V2G peak shed [%]/[kWh] | BEV energy demand [kWh] | V2G charged energy [kWh] |
|------------------|------------------------|------------------------------|------------------------------------|----------------------------|-----------------------------|
| 5 | 155 | 1.77 | 16.1/25 | 63 | 63 |
| 10 | 144 | 3 | 10/14 | 108 | 108 |
| 15 | 149 | 5.9 | 12.8/19 | 189 | 184 |
| 20 | 154 | 7.3 | 15.6/24 | 282 | 265 |
| 25 | 164 | 10.5 | 20.7/34 | 332 | 265 |
| 30 | 188 | 10.4 | 30.8/58 | 398 | 344 |
| 34 | 203 | 8.9 | 36/73 | 449 | 365 |

Table 14: Optimisation results for 90% (130kW) feeder limit of the maximum base load

Table 15: Optimisation results for 100% (144kW) feeder limit of the maximum base load

| BEVs | Maximum | $\mathrm{Mean}\ \mathrm{V2G}$ | Max. V2G | BEV energy | V2G charged |
|----------|-------------|-------------------------------|-------------|--------------|--------------|
| charging | feeder load | peak shed | peak shed | demand [kWh] | energy [kWh] |
| | | [%] | [%]/[kWh] | L J | 001 |
| 5 | 155 | 1.7 | 7.1/11 / | 63 | 63 |
| 10 | 144 | 2.8 | 0/0 / | 108 | 108 |
| 15 | 149 | 5.2 | $3.4/5 \;/$ | 189 | 189 |
| 20 | 154 | 5.9 | 6.5/10 / | 282 | 271 |
| 25 | 164 | 6.7 | 12.2/20 / | 332 | 290 |
| 30 | 188 | 8.2 | 23.4/44 / | 398 | 364 |
| 34 | 203 | 4.9 | 29.1/59 / | 449 | 393 |

In continuation of the subject that addresses the difference in charged energy for uncontrolled charging and V2G charging, this research initially assumed electric vehicles depart with a fully charged battery, while in reality this is not necessarily required. The average distance travelled in Amsterdam equals 22km [51], which requires only a small proportion of the battery capacity. During daily average trips the energy consumption equals 4.4 kWh approximately, according to an inefficient vehicle mobility factor of 0.2 kWh/km, as used in the article of [25]. More so, the sessions analysed in this research show a mean average energy consumption for every session of only 10.17kWh. Based on the prior acquainted results in paragraph 7.2, the energy charged in some of simulated cases, i.e. with a feeder limit of 70%, the V2G charging strategy may impose range limitations as the resulting charged energy is for 25,30,34 BEVs respectively 51%,41%, 55% lower than the required energy demand in the uncontrolled case. This translates to a energy consumption of the values in table 16 for each BEV in the V2G scenario in relation to total energy demand in the uncontrolled case. Although a large concession is made on the energy charged, it may still be sufficient for the average driver in Amsterdam to reach its destination. While this may be a breach on the requirements of the prosumer,

this study assumes a satisfactory consumption in the V2G strategy that is between 90%and original required. Again according to the same analogy of the colours, green is 90%or higher, and yellow denotes 4.4kWh or higher.

| Table 10. Energy charged per DEV for both uncontrolled charging the V20 charging | | | | | |
|--|--------------|-----------------|-----------------|-----------------|-----------------|
| | Uncontrolled | V2G consumption | V2G consumption | V2G consumption | V2G consumption |
| DE VS | consumption | 70% limit | 80% limit | 90% limit | 100% limit |
| charging | [kWh/BEV] | [kWh/BEV] | [kWh/BEV] | [kWh/BEV] | [kWh/BEV] |
| 5 | 12.7 | - | - | 12.7 | 12.7 |
| 10 | 10.8 | - | 10.4 | 10.8 | 10.8 |
| 15 | 12.6 | - | 10.4 | 12.3 | 12.6 |
| 20 | 14.1 | - | 11.7 | 13.3 | 13.5 |
| 25 | 13.2 | 6.5 | 9.2 | 10.6 | 11.6 |
| 30 | 13.2 | 7.8 | 10.2 | 11.5 | 12.1 |
| 34 | 13.2 | 5.9 | 8.8 | 10.7 | 11.6 |

Table 16: Energy charged per BEV for both uncontrolled charging the V2G charging

By varying the feeder limit, different feeder load circumstances in Amsterdam are mimicked to induce results for a V2G optimisation through a range of different connected electric vehicles. The amount of vehicles that the DSO allows to be charged by application of the V2G optimisation strategy under the congestion situations are derived from the former described tables and shown in table 17 and are the most important results for the distribution grid operator. Again in none of the situations of the uncontrolled BEV charging, the feeder load remained within the imposed constraint boundaries. This means that in each of the proposed cases for uncontrolled charging, the feeder will encounter a failure and breaks down. To provide a fair comparison, only the results of the simulations that attained the imposed constraints, i.e. required energy demand and feeder limit, are shown in table 17. The values represent the amount of vehicles that can be charged with the V2G strategy under varying circumstances of the feeder limit. In conclusion, one can notice that when the feeder limit increases, naturally more BEVs can be connected compared to the uncontrolled case, however at the cost of energy charged (see table 16).

| BEVs | Uncontrolled Maximum | Additional V2G | V2G | V2G | V2G |
|----------|----------------------|----------------|-----------|-----------|------------|
| charging | feeder demand | 70% limit | 80% limit | 90% limit | 100% limit |
| 5 | 155 | - | - | 5 | 5 |
| 10 | 144 | - | 10 | 10 | 10 |
| 15 | 149 | - | - | 15 | 15 |
| 20 | 154 | - | - | 20 | 20 |
| 25 | 164 | - | - | - | - |
| 30 | 188 | - | - | - | 30 |
| 34 | 203 | - | - | - | - |

Table 17: Additional BEV connection potential in the V2G optimisation cases

CAPEX OPPORTUNITY COSTS

The Capital Expenditure opportunity costs of a distribution grid asset is measured in terms of the best alternative available. As been introduced in paragraph 5.1 the prospective value configuration for the DSO depends on outgoing cash flows that represent costs for procurement of delivered flexibility and grid asset depreciation. Actual data on long-term grid investments is not available for this research. However, the latter costs eventually lead to early grid reinforcement investments that need faster depreciation deduction on the balance sheet, which may be temporarily deferred when V2G charging is applied. As described in the model of figure 7, the value flow for the DSO is significantly dependent on the size of BEV enabled flexibility, and thus can be expressed in the amount of chargeable BEVs under the V2G optimisation, and can be found in table 17. The V2G optimisation for the short-term scenario allows charging of 5,10,15, an 20 BEVs for a feeder limit of 90% and 100% respectively. While these results are significantly dependent on the interconnection time variables of the electric vehicles and their synchronisation with peak base loads, the results from the table provide sufficient proof-of-concept for the possibilities to aid in congestion in Amsterdam for different future BEV penetration rates. In short, for some of the V2G simulation results, for which feeder base loads range from 80% to 100% of its limit, the DSO may still allow BEVs to be charged by using this strategy. For a feeder with a limit of 80% (116kW), a maximum of 10 BEVs can be charged against zero in the uncontrolled charging scenario, which translates to approximately a future scenario in 2021 according to [29]. At a feeder limit of 90% (130kW) the amount of BEVs allowed to charge piles up to 20 that tends to be around 2025. Lastly, with a feeder limit of 100% (144kW) and at a maximum of 30 BEVs, the feeder limited was not violated and congestion is cleared. 30 BEVs corresponds to approximately a scenario in 2028. The exemptions for 25 and 34 BEVs are not included in the total allowable BEVs that may be charged due to the fact that under the V2G strategy these BEVs charged to little energy (still 87% and 88% for 25 and 30 BEVs of the required demand). This may be due a misalignment of interconnection times with peak base loads.

CHARGING ASSOCIATED COSTS

The charging costs associated with the simulated sessions are scaled in order to make a fair comparison between the different conveyed strategies. Although the proposed V2G LP model does not handle a minimisation of prices, it checks whether the proposed RTP tariff in paragraph 6.3 produces lower prices after scheduling the charging sessions. The table below depicts the values for price per kWh charged/discharged and the change in percentage of the price/kWh what would have been charged in the uncontrolled case (ignoring congestion) and the price when adopting the V2G strategy. The same colour analogy is applied to this table for which the simulated cases that pass the feeder constraints and the required energy demand are coloured green, yellow if V2G charged energy is less than required energy demand, and red if feeder limits are violated. The cleared V2G optimisation cases in green show a little decrease in costs and in some cases (for 10,15 BEVs charging) even an increase in costs compared to the uncontrolled charging case. This may be due to the fact that the RTP tariff rate is not aligned with the variation in base load, but rather with the wholesale day-ahead prices. However, the optimisation results that display higher charged prices are only marginally larger than the prices in the uncontrolled case, and prices for all other results are significantly lower to the original case.

The costs of charging and discharging are especially low when the constrained limit on the feeder is low, the charging sessions fall within peak periods and are connected for a relatively long time. Most of the results of the V2G simulations. The V2G strategy outperforms in almost every simulation charging in the uncontrolled scenario, by both charging during periods of low prices and discharging during periods of high prices even though the LP model does not optimise on prices. By combining a load-price levelling model, the prices charged to the consumer may be decreased even more [44].

| BEVs | Uncontrolled | Costs [euro/kWh]/[%] | Costs [euro/kWh] | Costs $[euro/kWh]$ |
|----------|-------------------------|-------------------------|-------------------------|--------------------|
| charging | costs | 50% limit | 60% limit | 70% limit |
| | [euro/kWh] | | | |
| 5 | 0.344 | 0.246/-29 | 0.254/-26 | 0.282/-18 |
| 10 | 0.339 | 0.326/-3.8 | 0.280/-17 | 0.305/-10 |
| 15 | 0.312 | 0.241/-23 | 0.266/-15 | 0.283/-9.3 |
| 20 | 0.313 | 0.186/-41 | 0.257/-18 | 0.265/-15 |
| 25 | 0.341 | 0.254/-26 | 0.251/-26 | 0.291/-15 |
| 30 | 0.337 | -0.131/-139 | 0.226/-33 | 0.258/-23 |
| 34 | 0.332 | -0.192/-158 | 0.143/-57 | 0.251/-24 |
| BEVs | Costs $[euro/kWh]/[\%]$ | Costs $[euro/kWh]/[\%]$ | Costs $[euro/kWh]/[\%]$ | |
| charging | 80% limit | 90% limit | 100% limit | |
| 5 | 0.314/-8.7 | 0.322/-6.4 | 0.343/-0.29 | |
| 10 | 0.318/-6.2 | 0.335/-1.2 | 0.341/0.6 | |
| 15 | 0.296/-5.1 | 0.317/1.6 | 0.320/2.6 | |
| 20 | 0.289/-7.7 | 0.307/-1.9 | 0.301/-3.8 | |
| 25 | 0.313/-8.2 | 0.320/-6.2 | 0.322/-5.6 | |
| 30 | 0.291/-14 | 0.314/-6.8 | 0.327/-2.9 | |
| 34 | 0.289/-13 | 0.308/-7.2 | 0.314/-5.4 | |

Table 18: Optimisation cost results for varying feeder limits and amount of electric vehicles

LONG-TERM SCENARIO SIMULATION

In the long term simulation, corresponding to 34 electric vehicles charging under 70% feeder limit of the maximum base load, the optimisation allows to charge all vehicles as fast as possible while remaining close to the feeder limit. Figure 31 shows the most optimal solution for charging the 34 sampled BEVs, for which the colour red is the total feeder load in the uncontrolled case, blue the feeder's base load, green the total feeder load for the V2G optimised solution, and purple to total load that is decreased. The model is designed to charge BEVs between the feeder limits and is consequently in some cases unattainable to meet required consumption because of grid power and connection time limitations. However, what can be noticed from figure 31 is how well the model serves its intended purpose to not violate the feeder constraints, especially during peak base loads from 16:00-22:00. Unfortunately, the V2G optimisation yields a consumption that is 44% lower than the uncontrolled case in order ensure the feeder constraint of 101kW is not violated. Although, this consumption may be sufficient for the average driver in Amsterdam as explained earlier.



Figure 31: Simulation LP program of the V2G charging strategy for 34 BEVs and a feeder limit of 70% its maximum base load

In conclusion, the coordinated BEV charging for a V2G strategy is shown to cover multiple interests for the prosumer, DSO and aggregator. The DSO may use the unlocked flexibility in the cases shown in table 17 to decrease congestion causing loads and additionally allow more BEVs to be charged while still meeting the imposed feeder limits. For a feeder with a limit of 80% (116kW), a maximum of 10 BEVs can be charged against zero in the uncontrolled charging scenario, which translates to approximately a future scenario in 2021 according to [29]. At a feeder limit of 90% (130kW) the amount of BEVs allowed to charge piles up to 20 that tends to be around 2025. Lastly, with a feeder limit of 100% (144kW) and at a maximum of 30 BEVs, the feeder limited was not violated and congestion is cleared. 30 BEVs corresponds to approximately a scenario in 2028. The exemptions for 25 and 34 BEVs are not included in the total allowable BEVs that may be charged due to the fact that under the V2G strategy these BEVs charged to little energy (still 87% and 88% for 25 and 30 BEVs of the required demand). This may be due a misalignment of interconnection times with peak base loads. Hence, the potential for a V2G charging strategy is significantly dependent on the connection time and feeder limit, and will therefore show significantly increased peak shed percentages if the connection time increases. Also, whenever the induced loads on the feeder exceed the maximum limit, the feeder will encounter a failure and immediately shuts down. Subsequently, the smart charging strategy is not applicable for a feeder limit other than 100%, thus this strategy will merely be used to verify the V2G charging strategy against that smart charging strategy in the simulation with a feeder limit of 100% the base load. The results from this comparison yield similar parameters for both the charging strategies. As been said, the model's performance works especially well during peak demand periods throughout the day, although produces a solution that makes a significant concession for the amount of energy charged in the long-term scenario. Although, none of the optimisation solutions for which the feeder constraint is not violated charge less energy than necessary for the average trip distance in Amsterdam. Furthermore, the V2G strategy outperforms charged prices per kWh in almost every simulation compared to the uncontrolled scenario, by both charging during periods of low prices and discharging during periods of high prices even though the LP model does not optimise on prices.

8. CONCLUSIONS

CONCLUSIONS

In the near future BEV electricity demand increases significantly, and because BEV prosumers show behavioural tendencies that concentrate at a certain time and additionally exhibit charging session properties that also show periodic characteristics, their demand ought to be controlled to alleviate potential overloading in power system operations. Currently as a result of the EU Energy Efficiency Directive of 2012, an institutional base is presented for development of demand response in Europe [17]. Therefore, to assess potential future incumbent value configuration of BEV demand response, the most focal value flows for the DSO are elicited, that proceed on an already existing communication infrastructure, (USEF) market framework, and intermediary actor that has real-time access to all BEVs. This model provides insight in prospective value configuration for the DSO, for which procurement of delivered flexibility should weigh up to additional grid reinforcement costs.

To be able to provide an answer to the main research question in this study an analysis is performed towards integration of a coordinated direct load control, the trends and periodicity of charging demand and flexibility, and lastly an optimisation scheduling model:

What is the value of demand response management in a Vehicle-to-Grid network and does it provide increased benefits to smart charging for consumers and the distribution system operator in Amsterdam?

This research proposed a (multi-aggregator) centralised approach of directly controlling and optimising the consumers' energy demand by introducing a planning protocol that may be integrated in the existing USEF framework. To be able to control BEV energy demand and remain within distribution boundaries, it is imperative to know BEV charging behaviour. Their impact on the local grid is assessed in terms of potential flexibility throughout the year as to investigate seasonal and daily patterns. It is demonstrated that BEV charging sessions in Amsterdam on average exhibit sufficient time-and loadflexibility values that can be exploited for effective demand response in districts where charging hot spots occur (Amsterdam south/centre). Especially during peak charging demand hours around 07:00-09:00h and 17:00-19:00h time-flexibility for individual BEVs shows significant peaks of 63% and 74% respectively. On average every day, each BEV has 61.7% of its total connection duration available to reschedule 10.17kWh for the smart charging strategy. This equals the regular energy demand as exhibited in the data set but can be rescheduled over the BEVs complete connection time. In terms of load-flexibility on average each BEV present in the data set exhibits the potential to discharge 21.03kWh during the week, and 24.68kWh in the weekend within the individual connection periods. Furthermore, the aggregator or the DSO are able to predict the periodic recurring behaviour of charging demand for three to four days ahead. This allows for more effective daily operational grid tasks as charging demand can be optimised by taking prospective BEV demand into account.

Further investigation about the usable amount of flexibility is simulated with linear programming for charging strategies (smart charging and V2G charging) to aid in local feeder congestion. In this research a case study is performed on load data of a low-voltage feeder. According to the current and near future BEV penetration rate, multivariate sampled charging sessions' load is modelled on the feeder. In conclusion, the coordinated BEV charging for a V2G strategy is shown to cover multiple interests for the prosumer, DSO and aggregator. The DSO may use the unlocked flexibility in the cases shown in table 17 to decrease congestion causing loads and additionally allow more BEVs to be charged while still meeting the imposed feeder limits. For a feeder with a limit of 80%(116kW), a maximum of 10 BEVs can be charged against zero in the uncontrolled charging scenario, which translates to approximately a future scenario in 2021 according to [29]. At a feeder limit of 90% (130kW) the amount of BEVs allowed to charge piles up to 20 that tends to be around 2025. Lastly, with a feeder limit of 100% (144kW) and at a maximum of 30 BEVs, the feeder limited was not violated and congestion is cleared. 30 BEVs corresponds to approximately a scenario in 2028. The exemptions for 25 and 34 BEVs are not included in the total allowable BEVs that may be charged due to the fact that under the V2G strategy these BEVs charged to little energy (still 87% and 88% for 25 and 30 BEVs of the required demand). This may be due a misalignment of interconnection times with peak base loads. Hence, the potential for a V2G charging strategy is significantly dependent on the connection time and feeder limit, and will therefore show significantly increased peak shed percentages if the connection time increases.

As far as the comparison of the smart charging and the V2G charging strategies goes, in all simulated cases congestion is apparent and makes the use of smart charging impossible, the feeder would have failed under congestion without having the possibility to discharge energy. As been said, the model's performance works especially well during peak demand periods throughout the day. In addition, the V2G strategy outperforms charged prices per kWh in almost every simulation compared to the uncontrolled scenario, by both charging during periods of low prices and discharging during periods of high prices even though the LP model does not optimise on prices. Thus, the V2G strategy allows the DSO to postpone grid investments in a number of cases while simultaneously the consumer almost always receives remuneration for its delivered services. The V2G strategy therefore provides a significantly added value over a smart charging strategy.

9. FUTURE WORK AND RECOMMENDATIONS

- 1. More in depth statistical analysis methods, such as cluster density based algorithms, or multi-nomial logit regression could be assessed to derive dependency structures between variables of interest in defining future BEV charging/driving behaviour and to perform validations on the predicted demand. This will allow for a more accurate distinction of typical recurring behaviour, which is, subsequently useful for an aggregators forecast on total power demand in spatial- and temporal terms.
- 2. In the work of [63] an algorithm is defined that computes the amount of energy and time at which flexibility is used in their charging strategies. They calculate a shift profile in vector form that quantifies the amount of energy shifted to a particular time slot within the optimisation window. This algorithm was initially developed for residential demand response, and can therefore be generalised to any other scheduled optimisation of demand. While the potential to ameliorate congestion at the local grid is explored in this research, information regarding the exact amount and time at which this energy is scheduled provides relevant insight for the DSO's grid operations. Furthermore, using this algorithm the LP program can be validated by results. This means comparing the model's results with real world outcomes, i.e. calculated time- and load-flexibility.
- 3. The value of DR in a liberalised system should be analysed more on the subject of the split-incentives challenge. Who initiates demand response (consumer, retailer, aggregator, DSO) and how should the benefits be divided along the supply chain. Handling an optimisation problem of DR for holds strong requirements for an global system balance in which neither participating actors are discriminated and the whole system benefits. An assessment of relational dependencies between participators in DR should be incorporated in designing charging strategies.
- 4. The charging strategies framework as proposed in this research can be expanded to integrate residential, or even larger distributed renewable energy generation, with the objective to charge or store this electricity generation and fully capture green energy by starting control at the consumer-side.
- 5. Implementation of additional required CAPEX opportunity costs for local grid assets according to future BEV penetration is an important next step to be undertaken to investigate the costs and benefits of building a economical feasible V2G infrastructure.

Appendix

A. ENERGY MARKET TRANSITION

This section provides the reader with context knowledge on the transformation of the Dutch power system and energy markets in reaction to governmental policy arrangements. The global energy systems currently use mainly resources that are oil-, coal- and gas-based to produce energy. The demand of these resources is prospected to grow until 2040, which entails a large reliance on these resources today and in the future [73]. A large share of the worldwide-consumed energy depends on these carbon energy systems defined as a carbon lock-in. This means that many industries and technologies are designed for and reliant on fossil fuels as primary energy resource. They also share the largest percentage equivalent of global gross domestic product. Inevitably, carbon energy resources will remain interdependent on the global economy for many years. It is therefore difficult and expensive to change or displace the current global energy system by better, cleaner alternatives in the very near future [73, 67].

Hence the global energy system is currently subjected to a shift in the energy paradigm that opens up transformational technology- and economical systems for renewable energy production [79, 3, 36]. Subsequently renewable energy resources are thought to be gaining up on conventional resources but remain currently to electricity generation. New technology development in energy systems will impose changes in the economy, policies, investments and mind-set of consumers. Currently the main objective of the current Dutch Ministry is to enlarge the share of renewable energy in the energy mix and invest more in low-carbon energy systems [79]. The new 2017 coalition of the Dutch government describe their target on reaching reductions in greenhouse gas emissions of 49% by 2030 and with that increasing the bar of the 40% level that is elicited by the European Union. 1.5% of that reduction is allocated to measures that include electric vehicles [79].

HISTORY REFORMATION DUTCH ENERGY MARKET

The original energy grid in the Netherlands has a top-to-bottom hierarchical structure. Its network is built-up with a large number of centralised generating facilities and many vertically disintegrated distribution sectors to provide electricity from a medium- and local sized network, up to the end users. This design dates back to the 19th century imposing significant large infrastructural networks that contribute to the economy of the Netherlands. The energy grids were heavily regulated and owned by the government because it was thought involvement of national authorities would prevent market failures. Regulation is also necessary to protect customers from the natural monopolies that electricity production companies develop.

However, with incentives for market efficiency, the Dutch Ministry of Economic affairs elicited a new sector organisation for stable institutional arrangements in the 1980s. Trends towards liberalisation of the utility market, resulting in a growing need for disintegration between producers and distributors to impose economies of scale for cost savings and improved market efficiency. During 1985-1989 existing market structures were to disrupt and the electricity infrastructure came into a transition that opened up opportunities for free competition that was self-regulated under support of the government. A distinction can be noticed between legal vertical disintegration between the production and transportation sector, and the distribution sector. A reform effort to challenge the production monopoly unbundled the relationships between the producers and distributors, which caused a sharp decline in the amount of production- and distribution companies. The decline was due to merge of local distributors into large concentrated provincial distributors and large distributors geographically related to the four biggest cities in the Netherlands [3, 36].

The Electricity Act of 1989

The new Electricity act of 1989 facilitated new market opportunities for producers that allowed optimisation of production processes to cost criteria instead of political directives. Production became a legal form of private stock companies that used a self-regulation authority, controlled by the government to provide reliable, secure supply and improvement of economic performance. The reform process or liberalisation changed the regulatory role of public authorities by making voluntary agreements between producers and distributors that required minor legal enforcement power [36].

The infrastructure after the integration of the Act appeared not very efficient and cost effective, because distributors were allowed to buy electricity from any regional producer for a state regulated national pooling price (abolition of monopolies) However, under the absence of price differences, resulting from free electricity purchase, this horizontal market structure was not used. Under the Electricity Act, distributors were allowed to produce electricity in small-scale quantities to compete with large producers. Large industrial consumers were not to limited to obligatory use of distributors or national import and subsequently made collaborations with local distributor-producers to build (CHP: Combined Heat Power) decentralised capacity. Later grown in size to 20% of the national production, fuelled by large investments of distributors and partly by large industrial consumers. For environmental reasons, distributors gained societal and political support for their decentralised renewable production.

The Electricity Act of 1998

The electricity act of 1998, allowed not only free competition and unbundling of utility companies, but also free choice of supplier for the largest consumers, and later in phases for medium- and small consumers (phased out in 2004). Distribution came under state regulation to secure a reliable and obligatory utility to consumers. Decentralised trading of electricity was made possible due to the Electricity Act of 1998 on a wholesale market. No longer making use of feed-in tariff structures, the distributors expanded their position in the electricity market [3, 36]. From now on every actor could make use of a fair energy pricing system. Distributed generation however, produced overcapacity in the energy market because of its competitive production methods against the large producers.

B. WHOLESALE AND RETAIL ELECTRICITY MARKET

Development of open accessible wholesale markets for trading electricity regulated by legal entities is an important factor to maintain a market efficient competitive energy system emphasised by increasing distributed (renewable) generation. In the Netherlands costs of distributed generation are strongly correlated with electricity market rules, therefore regulations alleviate entrance barriers by developing efficient market design for pricing ancillary services, intermittent resource supply, and small consumer access [50, 79, 75, 17]. However, current bid characteristics for trading electricity are not always utilised in an efficient process, which is partly blamed on the rapid technological transition of distributed generation on consumer level. More on that subject can be read in 3.1.

There are multiple energy markets in the Netherlands that facilitate the trading of energy as described in the process above, such as the day-ahead or the unbalance market. On the former market, energy contracts (dispatch amounts) are made between buyers and sellers for every hour of the following day. On the unbalance market every 5 min time stamp energy is dispatched to account for last minute imbalance in supply and demand. These prices are higher than the prices on the day ahead market, owing to the short time period in which energy needs to be dispatched. The electricity wholesale price depends on the generators' offered capacity to retailers for the following day in hourly intervals, in which the price depends on both the volume and time-of-use. Under- or over- estimations in supply and demand are directly related to the electricity prices. Other methods to trade electricity include bilateral contracts for trading parts of the electricity demands, usually applicable to commercial actors. Below depicted are the day-ahead hourly prices for the previous year, that show the historic real-time electricity value on the wholesale market. Some quick observations show higher prices in the during winter and weekend days, for which demand is higher and has more variation.



Figure 32: APX spot Day-ahead electricity prices

C. Analysis of distributions of electric vehicle session Characteristics

RATIO ACTUAL CHARGING TIME TO CONNECTION TIME

A majority of the charging sessions exhibit significant longer connection time than actually required for to suffice charging demand for those sessions. This trend enables the possibility to utilise the fact that BEVs, while connected to the grid, can provide flexibility in scheduling their charging. To further explore how the charging sessions' charging time relative to connection time is distributed, the ratio of actual charging time to connection duration is computed and plotted with a frequency polygon in figure 33 above. The ratio is a measure to calculate the required charging time (according to capacity charged) relative the total connection time for each session. The lower the ratio, the more idle time within sessions. As an example, 65% of the charging sessions have a ratio score below 50%. The ratio score of 50% means that connection time is twice the required charging time. By further graphical inspection, the graph appears to exhibit multi-modal peaks, and is furthermore positively skewed. Skewness is a descriptor that expresses a value for the asymmetry of the distribution, and is derived by inspection of the mean (0.3826) than the median (0.3061). A greater mean than median points out a larger part of the distribution has a ratio score below the median, meaning a trend in connection times longer than the necessary charging times. This statement can be backed by the fact that only a small peak is visible at a ratio of 1, meaning only a small proportion of the BEVs charge their vehicle and depart when the BEV battery is fully charged. Further analysis of the variables connection time, idle time and charging time is given below, which reveal more insight in the temporal trends and characteristics of the distributions.



Figure 33: Frequency distribution of the ratio charging time/connection time



CONNECTION TIME

Figure 34: Probability-density plot of connection time duration by sub-clusters of week-end days and weekdays for respectively 24h (a) & (c), 2nd 24h cluster (b) & (d), 3rd 24h cluster (c) & (f)



Figure 35: Boxplot and regression smoothing line of connection time duration by subclusters of weekend days and weekdays for 24h (a), 2nd 24h cluster (b), 3rd 24h cluster (c)

IDLE TIME



Figure 36: Probability-density plot of idle time duration by sub-clusters of 24h connection duration



Figure 37: Boxplot and regression smoothing line of idle time duration by sub-clusters of weekend days and weekdays for 24h (a), 2nd 24h cluster (b), 3rd 24h cluster (c)

CHARGING DEMAND

Figure 38 is the probability-density function (PDF) of the relative- and cumulative occurrences for different capacities charged. The graph displays the density function of the univariate continuous charging demand variable (a) and is interpreted as a relative likelihood that this variable takes on any particular value on the given range. The cumulativedensity function (CDF) depicts the probability that the charging demand variable takes a value less then or equal to p_i . The PDF (31) and CDF (32) are expressed as:

$$P[1 \le P^i \le 100] = \int_1^{100} f(p_i) \quad dp_i \tag{31}$$

$$F^{i}(p_{i}) = P[P^{i} \le p_{i}] = \int_{1}^{100} f(p_{i}) \, dp_{i}$$
(32)

$$P(1 < P^i < 100), \quad \forall p_i \in P^i(p^i)$$

$$(33)$$



Figure 38: (a) Probability-density distribution of charging demand

C.1. TIME- & LOAD FLEXIBILITY

TIME FLEXIBILITY

Figure 39 displays the probability density distribution of charging sessions for the time flexibility. The plot on the right is a cumulative distribution of the time flexibility that represents the probability (or frequency of sessions), for which the load-flexibility variable takes a value less than or equal to the load-flexibility on the semi-closed interval (0, 1006]. Formally expressed as:

$$F_{T_{Flex}}(x) = P(T_{Flex} \le x) = \int_{-\inf}^{1} f(T_{Flex}) \quad dx \tag{34}$$

Where

$$P(0 < T_{Flex} < 1, \quad \forall x \in F_{T_{Flex}}(x), \tag{35}$$



Figure 39: Frequency polygon time-flexibility fraction of the charging session distribution

LOAD FLEXIBILITY

In the (probability-density) frequency distributions of the load-flexibility below, a positive skewness and a laptocurtic kurtosis characterise the shape indicating a long-tailed distribution with extreme outliers. To explain the dispersion these extremities cause, the Interquartile Range (IQR) characterises the spread in the data equal to the difference between the 75th and 25th percent quantiles of the median. Where the median is a measure of central tendency, and the IQR is the measure in spread of 50% of the data from the median, also plotted in figure 27. More formally addressed and computed with values of the data set (table 8), 50% of the sessions have a load-flexibility of: $IQR = Q_3 - Q_1$. The heavy kurtosis and skewness can be explained mainly by the fraction of the charging sessions that exhibit long (24h <) connection times. In other words, longer connection times are amongst other factors influenced by charging demand (initial battery capacity), however no session recorded charges longer than 22h which leaves every session connected longer than 22h with a increasing potential for load-flexibility.



Figure 40: Frequency polygon load-flexibility of the charging session distribution

D. FEEDER POWER LOAD PROFILES

Electricity profiles for load on low-voltage feeders in Amsterdam are used to retrieve hourly profiles for residential buildings. These profiles represent the electricity consumption and generation of electrical appliances, cooking, lighting and PV solar cells of residential buildings. This local distribution grid also operates two working V2G charging points. In the figures 41 & 42 below, the feeder load profiles without V2G charging, the feeder with V2G charging and the V2G charging load are measures and depicted.



Figure 41: Total power load on feeder k2195-k2193 at Amsterdam Nieuw-West with two V2G charging point



Figure 42: Total power load on feeder k
2192-k 2193 at Amsterdam Nieuw-West with two V2G charging point

E. DUTCH ELECTRIC VEHICLE MARKET SHARE & BATTERY CHARACTERISTICS

Information used in this study about battery characteristics of PHEVs and BEVs top five and top ten models with largest market shares respectively, are listed below in the tables.

| - | | | L / / |
|---------------------------------|-----------------|------------------------|--------------|
| $\mathbf{Brand}/\mathbf{Model}$ | Number/Fraction | Battery capacity (kWh) | Battery type |
| Mitsubishi Outlander | 24.992~/~25.5% | 12 | Li-ion |
| Volvo V60 | 15.554~/~15.9% | 11.2 | Li-ion |
| Volkswagen Golf | 10.907~/~11.1% | 8.7 | Li-ion |
| Volkswagen Passat | 7.994~/~8.2% | 9.9 | Li-ion |
| Audi A3 | 6.274~/~6.4% | 8.8 | Li-ion |
| Other | 32.298~/~33% | 10.7 | Li-ion |

Table 19: Top five most sold PHEVs in the Netherlands and market share [61, 43, 78, 77]

Table 20: Top ten most sold BEVs in the Netherlands and market share [61, 49, 57, 7, 28, 48, 58, 53, 45, 46]

| $\mathbf{Brand}/\mathbf{Model}$ | Number/Fraction | Battery capacity (kWh) | Battery type |
|---------------------------------|----------------------|------------------------|--------------|
| Tesla model S | $8.824 \ / \ 33.9\%$ | 75 | Li-ion |
| Nissan Leaf | 2.842~/~10.9% | 40 | Li-ion |
| Renault ZEO | 2.751~/~10.6% | 22 | Li-ion |
| BMW I3 | 2.196~/~8.4% | 33.2 | Li-ion |
| Volkswagen Golf | 2.152~/~8.3% | 35.8 | Li-ion |
| Tesla model X | 2.061~/~7.9% | 100 | Li-ion |
| Hyundai Ioniq | 1.600~/~6.2% | 28 | Li-ion |
| Nissan E-NV200 | $816 \ / \ 3.1\%$ | 40 | Li-ion |
| Renault Kangoo Z.E. | 785~/~3.0% | 33 | Li-ion |
| Opel Ampera | 595~/~2.3% | 60 | Li-ion |

References

- [1] Alliander. "De flexibiliteit van huishoudelijk stroomverbruik : Een nieuwe markt met kansen". In: (2016), pp. 1-50. URL: https://www.nederlandict.nl/wpcontent/uploads/2017/10/2016 - EnergieKoplopers_eindrapport_2016_ definitief.pdfhttps://www.energiekoplopers.nl/wp-content/uploads/ 2016/11/EnergieKoplopers_eindrapport.pdf.
- Mohammad Hassan Amirioun and Ahad Kazemi. "A new model based on optimal scheduling of combined energy exchange modes for aggregation of electric vehicles in a residential complex". In: *Energy* 69 (2014), pp. 186–198. ISSN: 03605442. DOI: 10.1016/j.energy.2014.02.063. URL: http://dx.doi.org/10.1016/j.energy.2014.02.063.
- M.J. Arentsen, R.W. Künneke, and H.C. Moll. "Chapter VI The Dutch Electricity Reform: Reorganisation by Negotiation". In: *European Electricity Systems in Transition*. Elsevier, 1997, pp. 167–196. ISBN: 9780080429946. DOI: 10.1016/B978-008042994-6/50006-7. URL: http://linkinghub.elsevier.com/retrieve/pii/ B9780080429946500067.
- [4] Mariz B. Arias and Sungwoo Bae. "Electric vehicle charging demand forecasting model based on big data technologies". In: *Applied Energy* 183 (2016), pp. 327–339. ISSN: 03062619. DOI: 10.1016/j.apenergy.2016.08.080. URL: http://dx.doi.org/10.1016/j.apenergy.2016.08.080.
- [5] Mariz B. Arias and Sungwoo Bae. "Prediction of electric vehicle charging-power demand in realistic urban traffic networks". In: *Applied Energy* 195 (2017), pp. 738–753. ISSN: 03062619. DOI: 10.1016/j.apenergy.2017.02.021. URL: http://dx. doi.org/10.1016/j.apenergy.2017.02.021.
- [6] Pradyumna C. Bhagwat et al. "The effectiveness of a strategic reserve in the presence of a high portfolio share of renewable energy sources". In: Utilities Policy 39 (2016), pp. 13–28. ISSN: 09571787. DOI: 10.1016/j.jup.2016.01.006.
- [7] BMW. BMW I3. 2018. URL: https://www.bmw.nl/nl/modellen/bmwi/i3/ ontdek/actieradius-opladen.html?&bmwcache&bmw=sea:93911831:aud-426277158377:kwd-60672320943:+bmw+i3+range&gclid=CjwKCAjw_47YBRBxEiwAYuKdw_ xKCeR-MdTet2uU_P6a4aDGZNF-crmAy8gFE9LQQ-mZTGiWOu_9jxoCY5oQAvD_BwE.
- [8] Tobias Brandt, Sebastian Wagner, and Dirk Neumann. "Evaluating a business model for vehicle-grid integration: Evidence from Germany". In: *Transportation Research Part D: Transport and Environment* 50 (2017), pp. 488–504. ISSN: 13619209.
 DOI: 10.1016/j.trd.2016.11.017. URL: http://dx.doi.org/10.1016/j.trd. 2016.11.017.
- CBS. Statistics Dutch registered vehicles. 2018. URL: http://statline.cbs.nl/ Statweb/publication/?DM=SLNL&PA=7374hvv&D1=0-5&D2=a&D3=1&HDR=G2,T& STB=G1&VW=T.

- [10] Robbert B. Cleveland et al. *STL: A seasonal-trend decomposition procedure based* on Loess. 1990.
- [11] NicolĀš Daina, Aruna Sivakumar, and John W. Polak. "Electric vehicle charging choices: Modelling and implications for smart charging services". In: *Transportation Research Part C: Emerging Technologies* 81 (2017), pp. 36–56. ISSN: 0968090X. DOI: 10.1016/j.trc.2017.05.006.
- [12] NicolĀš Daina, Aruna Sivakumar, and John W. Polak. "Modelling electric vehicles use: a survey on the methods". In: *Renewable and Sustainable Energy Reviews* 68. August 2015 (2017), pp. 447-460. ISSN: 18790690. DOI: 10.1016/j.rser.2016. 10.005. URL: http://dx.doi.org/10.1016/j.rser.2016.10.005.
- [13] Sara Deilami et al. "Paper PresentationâĂŤReal-Time Coordination of Plug-In Electric Vehicle Charging in Smart Grids to Minimize Power Losses and Improve Voltage Profile". In: *IEEE Transactions on Smart Grid* 2.3 (2011), pp. 456-467.
 ISSN: 19493053. DOI: 10.1109/TSG.2011.2159816. URL: http://ieeexplore. ieee.org/lpdocs/epic03/wrapper.htm?arnumber=5986769%OAhttp://ieeexplore. ieee.org/document/5986769/.
- [14] ECN. "Demand and supply of flexibility in the power system of the Netherlands, 2015-2050 Summary report of the FLEXNET project". In: November 2017 (2017), pp. 1-69. URL: https://www.ecn.nl/publicaties/PdfFetch.aspx?nr=ECN-E--17-053https://www.ecn.nl/publicaties/PdfFetch.aspx?nr=ECN-E--17-063.
- [15] ECN et al. "National Energy Outlook 2017- summary". In: (2017), pp. 1-3. URL: https://english.rvo.nl/sites/default/files/2017/11/NationalEnergyOutlook2017_ Summary.pdf.
- [16] Max Ehring and Jurjen Witteveen. "Breakthrough of electric vehicle threatens European car industry". In: July (2017), pp. 1–18. URL: https://www.ing.nl/ media/ING_EBZ_breakthrough-of-electric-vehicle-threatens-Europeancar-industry_tcm162-128687.pdf.
- [17] Cherrelle Eid et al. "Time-based pricing and electricity demand response: Existing barriers and next steps". In: Utilities Policy 40 (2016), pp. 15-25. ISSN: 09571787. DOI: 10.1016/j.jup.2016.04.001. URL: http://dx.doi.org/10.1016/j.jup. 2016.04.001.
- [18] Entsoe. EPEX energy market. URL: https://transparency.entsoe.eu/transmissiondomain / r2 / dayAheadPrices / show ? name = &defaultValue = true & viewType = GRAPH&areaType=BZN&atch=false&dateTime.dateTime=22.09.2018+00:00 | CET | DAY&biddingZone.values=CTY | 10YNL-----L!BZN | 10YNL------L&dateTime.timez.
- B. Priya Esther and K. Sathish Kumar. "A survey on residential Demand Side Management architecture, approaches, optimization models and methods". In: *Renewable and Sustainable Energy Reviews* 59 (2016), pp. 342–351. ISSN: 18790690.
 DOI: 10.1016/j.rser.2015.12.282. URL: http://dx.doi.org/10.1016/j.rser.2015.12.282.

- [20] A.V. Fiacco and G.P. McCormick. Nonlinear Programming: Sequential Unconstrained Minimization Techniques. John Wiley & Sons, New York, 1968.
- Murray Goulden et al. "Smart grids, smart users? the role of the user in demand side management". In: *Energy Research and Social Science* 2 (2014), pp. 21–29. ISSN: 22146296. DOI: 10.1016/j.erss.2014.04.008. URL: http://dx.doi.org/10.1016/j.erss.2014.04.008.
- [22] Salman Habib, Muhammad Kamran, and Umar Rashid. "Impact analysis of vehicle-to-grid technology and charging strategies of electric vehicles on distribution networks A review". In: Journal of Power Sources 277 (2015), pp. 205–214. ISSN: 03787753. DOI: 10.1016/j.jpowsour.2014.12.020. URL: http://dx.doi.org/10.1016/j.jpowsour.2014.12.020.
- [23] de Hans Heer and van den Willem Reek. *Flexibility Platforms*. Tech. rep. 2018.
- [24] Frederick S. Hillier and Gerald J. Lieberman. Introduction to operations research. 10th ed. Mcgraw-Hill Education - Europe, 1995. ISBN: 9780071267670.
- [25] T. W. Hoogvliet, G. B.M.A. Litjens, and W. G.J.H.M. van Sark. "Provision of regulating- and reserve power by electric vehicle owners in the Dutch market". In: *Applied Energy* 190 (2017), pp. 1008–1019. ISSN: 03062619. DOI: 10.1016/j.apenergy.2017.01.006. URL: http://dx.doi.org/10.1016/j.apenergy.2017.01.006.
- Junjie Hu et al. "Electric vehicle fleet management in smart grids: A review of services, optimization and control aspects". In: *Renewable and Sustainable Energy Reviews* 56 (2016), pp. 1207–1226. ISSN: 18790690. DOI: 10.1016/j.rser.2015. 12.014. URL: http://dx.doi.org/10.1016/j.rser.2015.12.014.
- [27] Rob Hyndman, Xiaozhe Wang, and Kate Smith. "Characteristic-based clustering for time series Data". In: *Data Mining and Knowledge Discovery* 13 (2005), pp. 457– 476. ISSN: 15220613. DOI: 10.1007/s10618-005-0039-x.
- [28] Hyundai. Hyundai IONIQ electric. 2018. URL: https://www.hyundai.nl/hyundaiioniq-ev.
- [29] IEA. "Global EV Outlook 2018". In: (2018). DOI: 10.1787/9789264302365-en. URL: https://www.oecd-ilibrary.org/energy/global-ev-outlook-2018_ 9789264302365-en.
- [30] Emil B. Iversen, Juan M. Morales, and Henrik Madsen. "Optimal charging of an electric vehicle using a Markov decision process". In: *Applied Energy* 123 (2014), pp. 1–12. ISSN: 03062619. DOI: 10.1016/j.apenergy.2014.02.003. URL: http://dx.doi.org/10.1016/j.apenergy.2014.02.003.
- [31] Emil Banning Iversen et al. "Inhomogeneous Markov Models for Describing Driving Patterns". In: *IEEE Transactions on Smart Grid* 8.2 (2017), pp. 581–588. ISSN: 19493053. DOI: 10.1109/TSG.2016.2520661.

- [32] Emre C. Kara et al. "Estimating the benefits of electric vehicle smart charging at non-residential locations: A data-driven approach". In: Applied Energy 155 (2015), pp. 515-525. ISSN: 03062619. DOI: 10.1016/j.apenergy.2015.05.072. URL: http://dx.doi.org/10.1016/j.apenergy.2015.05.072.
- [33] Ahsan Raza Khan et al. "Load forecasting, dynamic pricing and DSM in smart grid: A review". In: *Renewable and Sustainable Energy Reviews* 54 (2016), pp. 1311-1322.
 ISSN: 18790690. DOI: 10.1016/j.rser.2015.10.117. URL: http://dx.doi.org/ 10.1016/j.rser.2015.10.117.
- [34] Laurence D. Kirsch and Mathew J. Morey. "Pricing retail electricity in a distributed energy resources world". In: *Electricity Journal* 28.3 (2015), pp. 112-121. ISSN: 10406190. DOI: 10.1016/j.tej.2015.03.004. URL: http://dx.doi.org/10.1016/j.tej.2015.03.004.
- [35] Anco Koen, Maarten Afman, and Anouk Grinsven Van. "Uitbreiding publieke laadinfrastructuur tot 2020". In: (2020).
- [36] Rolf W. Künneke and Monique H. Voogt. "Modelling welfare effects of a liberalisation of the Dutch electricity market". In: *Energy* 22.9 (1997), pp. 897–910. ISSN: 03605442. DOI: 10.1016/S0360-5442(97)00024-8.
- [37] Liansheng Liu et al. "A review on electric vehicles interacting with renewable energy in smart grid". In: *Renewable and Sustainable Energy Reviews* 51 (2015). ISSN: 18790690. DOI: 10.1016/j.rser.2015.06.036.
- [38] J. A PeAğas Lopes, P. M Rocha Almeida, and F. J. Soares. "Using vehicle-to-grid to maximize the integration of intermittent renewable energy resources in islanded electric grids". In: 2009 International Conference on Clean Electrical Power, ICCEP 2009 (2009), pp. 290–295. DOI: 10.1109/ICCEP.2009.5212041.
- [39] A. Mahmood et al. "A new scheme for demand side management in future smart grid networks". In: *Procedia Computer Science* 32 (2014), pp. 477-484. ISSN: 18770509. DOI: 10.1016/j.procs.2014.05.450. URL: http://dx.doi.org/10.1016/j.procs.2014.05.450.
- [40] Petra Mesarić and Slavko Krajcar. "Home demand side management integrated with electric vehicles and renewable energy sources". In: *Energy and Buildings* 108 (2015), pp. 1–9. ISSN: 03787788. DOI: 10.1016/j.enbuild.2015.09.001.
- [41] J Mies, J Helmus, and Robert van den Hoed. "Estimating the charging profile of individual charge sessions of Electric Vehicles in the Netherlands". In: *Electric Vehicle Symposium and Exhibition 30* (2017), pp. 1–12.
- [42] Catherine Mitchell and Peter Connor. "Review of Current Electricity Policy and Regulation". In: *The University of Warwick* February (2002). ISSN: 1098-6596. DOI: 10.1017/CB09781107415324.004.
- [43] Mitsubishi-Motors. Outlander PHEV. 2018. URL: https://www.mitsubishimotors.nl/outlander-PHEV/#!specificaties.

- [44] Sang Keun Moon and Jin O. Kim. "Balanced charging strategies for electric vehicles on power systems". In: Applied Energy 189 (2017). ISSN: 03062619. DOI: 10.1016/ j.apenergy.2016.12.025.
- [45] Tesla Motors. Tesla model S. 2018. URL: https://www.tesla.com/models.
- [46] Tesla Motors. Tesla model X. 2018. URL: https://www.tesla.com/nl_NL/modelx.
- [47] Joy Chandra Mukherjee and Arobinda Gupta. "A Review of Charge Scheduling of Electric Vehicles in Smart Grid". In: *IEEE Systems Journal* 9.4 (2015), pp. 1541– 1553. ISSN: 19379234. DOI: 10.1109/JSYST.2014.2356559.
- [48] Nissan. Nissan E-NV200. 2018. URL: https://www.nissan.nl/voertuigen/ nieuw/e-nv200-evalia.html.
- [49] Nissan. Nissan Leaf. 2018. URL: https://www.nissan.nl/voertuigen/nieuw/ leaf/bereik-opladen.html?&cid=psmBu1FVciH_dc|D.
- [50] OECD. "Netherlands Regulatory Reform in the Electricity Industry". In: Report, Competition Law & Policy (1998), pp. 1-41. URL: http://www.oecd.org/ regreform/liberalisationandcompetitioninterventioninregulatedsectors/ 2497395.pdf.
- [51] OIS. Average distance travelled in Amsterdam per transport mode. Tech. rep. 2017, p. 2016.
- [52] OIS. Number PEVs in Amsterdam 2017. Tech. rep. 2017, pp. 2016–2017.
- [53] Opel. Opel Ampera. 2018. URL: http://www.opel.nl/onze-modellen/modellenoverzicht/ personenautos/ampera-e/index.html.
- [54] Alexander Osterwalder. "The Business Model Ontology A Proposition in a Design Science Approach". In: Business Doctor (2004), pp. 1–169. ISSN: 00336807. DOI: 10.1111/j.1467-9310.2010.00605.x. URL: http://citeseerx.ist.psu. edu/viewdoc/download?doi=10.1.1.134.8520&rep=rep1&type= pdf\nhttp://www.stanford.edu/group/mse278/cgi-bin/wordpress/wpcontent/uploads/2010/01/TheBusiness-Model-Ontology.pdf.
- [55] E. Pashajavid and M. A. Golkar. "Non-Gaussian multivariate modeling of plug-in electric vehicles load demand". In: *International Journal of Electrical Power and Energy Systems* 61 (2014), pp. 197–207. ISSN: 01420615. DOI: 10.1016/j.ijepes. 2014.03.021. URL: http://dx.doi.org/10.1016/j.ijepes.2014.03.021.
- [56] Jairo Quirós-tortós and Becky Lees. "A statistical analysis of EV charging behavior in the UK Accepted Paper A Statistical Analysis of EV Charging Behavior in the UK Accepted Paper". In: October (2015), pp. 445–449. DOI: 10.1109/ISGT-LA.2015.7381196.
- [57] Renault. Renault Kangoo Z.E. 2018. URL: https://www.renault.nl/modellen/ elektrische-autos/kangoo-ze/actieradius.html.
- [58] Renault. Renault Zoe. 2018. URL: https://www.renault.nl/modellen/elektrischeautos/zoe/uitvoeringen.html.

- [59] David B. Richardson. "Electric vehicles and the electric grid: A review of modeling approaches, Impacts, and renewable energy integration". In: *Renewable and Sustainable Energy Reviews* 19 (2013), pp. 247–254. ISSN: 13640321. DOI: 10.1016/j. rser.2012.11.042. URL: http://dx.doi.org/10.1016/j.rser.2012.11.042.
- [60] Peter Richardson, Damian Flynn, and Andrew Keane. "Local versus centralized charging strategies for electric vehicles in low voltage distribution systems". In: *IEEE Transactions on Smart Grid* 3.2 (2012), pp. 1020–1028. ISSN: 19493053. DOI: 10.1109/TSG.2012.2185523.
- [61] RVO, Netherlands Enterprise Agency. "Statistics Electric Vehicles in the Netherlands". In: May (2018), pp. 2-5. URL: https://www.rvo.nl/sites/default/ files/2017/10/Elektricvehicles-Special-Analysisover2016ENG.pdf.
- [62] N. Sadeghianpourhamami et al. "Modeling and analysis of residential flexibility: Timing of white good usage". In: *Applied Energy* 179 (2016), pp. 790-805. ISSN: 03062619. DOI: 10.1016/j.apenergy.2016.07.012. URL: http://dx.doi.org/ 10.1016/j.apenergy.2016.07.012.
- [63] N. Sadeghianpourhamami et al. "Quantitive analysis of electric vehicle flexibility: A data-driven approach". In: International Journal of Electrical Power and Energy Systems 95 (2018), pp. 451-462. ISSN: 01420615. DOI: 10.1016/j.ijepes.2017. 09.007. URL: http://dx.doi.org/10.1016/j.ijepes.2017.09.007.
- [64] Pierluigi Siano. "Demand response and smart grids A survey". In: Renewable and Sustainable Energy Reviews 30 (2014), pp. 461–478. ISSN: 13640321. DOI: 10.1016/ j.rser.2013.10.022. URL: http://dx.doi.org/10.1016/j.rser.2013.10.022.
- [65] Maarten Steinbuch. Schatting verkopen elektrisch vervoer Nederland 2025. 2016. URL: https://steinbuch.wordpress.com/2016/01/22/schatting-verkopenelektrisch-vervoer-nederland-tm-2025/.
- [66] Chua Liang Su and Daniel Kirschen. "Quantifying the effect of demand response on electricity markets". In: *IEEE Transactions on Power Systems* 24.3 (2009), pp. 1199–1207. ISSN: 08858950. DOI: 10.1109/TPWRS.2009.2023259. URL: http: //ieeexplore.ieee.org/lpdocs/epic03/wrapper.htm?arnumber=5156266.
- [67] Wencong Su et al. "A survey on the electrification of transportation in a Smart Grid Environment". In: *Ieee Transactions on Industrial Informatics* 8.1 (2012), pp. 1–10.
- [68] Olle Sundström and Carl Binding. "Flexible charging optimization for electric vehicles considering distribution grid constraints". In: *IEEE Transactions on Smart Grid* 3.1 (2012), pp. 26–37. ISSN: 19493053. DOI: 10.1109/TSG.2011.2168431.
- [69] Olle Sundström and Carl Binding. "Optimization Methods to Plan the Charging of Electric Vehicle Fleets". In: Proceedings of the International Conference on Control Communication and Power Engineering 1.2 (2010), 28âĂŞ29. DOI: 01.ijcom.
 2010.02.09. URL: http://searchdl.org/public/journals/2010/IJCOM/1/2/ 09.pdf\nhttp://www.zurich.ibm.com/pdf/csc/EDISON_ccpe_main.pdf.
- [70] Kang Miao Tan, Vigna K. Ramachandaramurthy, and Jia Ying Yong. "Integration of electric vehicles in smart grid: A review on vehicle to grid technologies and optimization techniques". In: *Renewable and Sustainable Energy Reviews* 53 (2016), pp. 720–732. ISSN: 18790690. DOI: 10.1016/j.rser.2015.09.012. URL: http://dx.doi.org/10.1016/j.rser.2015.09.012.
- [71] Kotub Uddin et al. "On the possibility of extending the lifetime of lithium-ion batteries through optimal V2G facilitated by an integrated vehicle and smart-grid system". In: *Energy* 133 (2017), pp. 710-722. ISSN: 03605442. DOI: 10.1016/j.energy.2017.04.116. URL: http://dx.doi.org/10.1016/j.energy.2017.04.116.
- [72] United Nations/Framework Convention on Climate Change. "Paris Agreement". In: 21st Conference of the Parties (2015), p. 3. ISSN: 1098-6596. DOI: FCCC/CP/2015/ L.9.
- U.S. Energy Information Administration. International Energy Outlook 2016. Vol. 0484(2016). May 2016. 2016, pp. 1-2. ISBN: 2025866135. DOI: www.eia.gov/forecasts/ieo/ pdf/0484(2016).pdf. URL: www.eia.gov/forecasts/ieo/pdf/0484(2016).pdf.
- [74] USEF Foundation and Universal Smart Energy Framework. USEF: The Framework explained. 2015, pp. 1-55. ISBN: 9789082462517. DOI: 10.1016/S1569 -9056(10)61174-8. URL: https://www.usef.energy/app/uploads/2016/ 12/USEF_TheFrameworkExplained-18nov15.pdf#popup__overlay1%0Ahttp: //www.globalsmartgridfederation.org/wp-content/uploads/2016/10/USEF_ TheFrameworkExplained-18nov15.pdf%0Ahttps://www.usef.energy/app/ uploads/2016/12/U.
- S. Vardakas John, Nizar Zorba, and Christos V. Verikoukis. "A Survey on Demand Response Programs in Smart Grids: PricingMethods and Optimization Algorithms". In: *IEEE Transactions on Industrial Informatics* 11.3 (2015), pp. 570–582. ISSN: 15513203. DOI: 10.1109/TII.2015.2414719. URL: http://ieeexplore.ieee.org/lpdocs/epic03/wrapper.htm?arnumber=7063260.
- [76] Remco A. Verzijlbergh et al. "Network impacts and cost savings of controlled EV charging". In: *IEEE Transactions on Smart Grid* 3.3 (2012), pp. 1203–1212. ISSN: 19493053. DOI: 10.1109/TSG.2012.2190307.
- [77] Volkswagen. Volkswagen Golf E. 2018. URL: https://www.volkswagen.nl/ modellen/golf/e-golf.
- [78] Volkswagen Golf GTE. 2018. URL: https://www.volkswagen.nl/voorraad/golfgte.
- [79] VVD, D66, and Christian-Union. Confidence in the Future: 2017-2021 Coalition Agreement. Tech. rep. October. 2017, pp. 41-47. URL: http://www.vodafone.com/ content/annualreport/annual_report16/downloads/vodafone-full-annualreport-2016.pdf.

- [80] Dai Wang et al. "Quantifying electric vehicle battery degradation from driving vs. vehicle-to-grid services". In: Journal of Power Sources 332 (2016), pp. 193-203. ISSN: 03787753. DOI: 10.1016/j.jpowsour.2016.09.116. URL: http://dx.doi.org/10.1016/j.jpowsour.2016.09.116.
- [81] Jidong Wang et al. "Optimal dispatching model of smart home energy management system". In: 2012 IEEE Innovative Smart Grid Technologies - Asia, ISGT Asia 2012 (2012), pp. 1–5. DOI: 10.1109/ISGT-Asia.2012.6303266.
- [82] Yue Wang and David Infield. "Markov Chain Monte Carlo simulation of electric vehicle use for network integration studies". In: *International Journal of Electrical Power and Energy Systems* 99.November 2017 (2018), pp. 85–94. ISSN: 01420615.
 DOI: 10.1016/j.ijepes.2018.01.008.
- [83] Werner van Westering and Hans Hellendoorn. Low Voltage Power Grid Stabilization using a Community Battery: Design principles, Control and Experimental Validation. 2018.
- [84] Rong Yu et al. "Balancing Power Demand Through EV Mobility in Vehicle-to-Grid Mobile Energy Networks". In: *IEEE Transactions on Industrial Informatics* 12.1 (2016), pp. 79–90. ISSN: 15513203. DOI: 10.1109/TII.2015.2494884.
- [85] Yuksel. "Effects of Regional Temperature on Electric Vehicle Efficiency, Range, and Emissions in the United States". eng. In: *Environmental Science Technology* 49.6 (2015), pp. 3974–3980.
- [86] Kan Zhou and Lin Cai. "Randomized PHEV charging under distribution grid constraints". In: *IEEE Transactions on Smart Grid* 5.2 (2014), pp. 879–887. ISSN: 19493053. DOI: 10.1109/TSG.2013.2293733.